

NEW MANGALORE PORT AUTHORITY

CIVIL ENGINEERING DEPARTMENT

Ref. No. NMPT/CIVIL/PPP PROJECT/2020-21/TS

CORRIGENDUM NO. 5 AND REPLY TO PRE-BID QUERIES

TO THE POTENTIAL BIDDERS

Sub: Request for Proposal (RFP) for Appointment of Transaction Advisor for Concessioning of Berth No. 13 for handling Liquid Bulk / LPG / POL products / Chemicals on Captive / PPP Mode – Reg.

Ref: 1) NIT No. CIVIL/CE(C)/EE(C)/05/2022-23, dated 29.04.2022 and eProcure Tender ID: 2022_NMPT_686079_1.

- 2) Corrigendum No. 1 dtd. 16.05.2022.
- 3) Corrigendum No. 2 dtd. 27.05.2022.
- 4) Corrigendum No. 3 dtd. 14.06.2022.

Consequent upon the queries raised by bidders for the subject work, New Mangalore Port Authority is pleased to clarify the queries of the Bidders. The Reply to Pre-Bid Queries and subsequent modification to RFP document is issued as Corrigendum No.5.

Further, the last date for submission of bids shall be 16.08.2022 upto 15:30 hrs. and the date of opening of the bids shall be on 17.08.2022 after 15:30 hrs.

The Corrigendum No.5 shall be duly signed and submitted along with the Cover-I of the bid.

Sd/-

Date: 25.07.2022

Chief Engineer (Civil)i/c

NEW MANGALORE PORT AUTHORITY

Date: 25.07.2022

Name of work: Request for Proposal (RFP) for Appointment of Transaction Advisor for Concessioning of Berth No. 13 for handling Liquid Bulk

/ LPG / POL products / Chemicals on Captive / PPP Mode – Reg.

CIVIL/CE(C)/EE(C)/05/2022-23 dated 29.04.2022 and eProcure Tender ID: 2022_NMPT_686079_1. NIT No.:

REPLY TO PRE BID QUERES

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA						
I	M/s. CRISIL Infrastructure Advisory, A division of CRISIL Risk & Infrastructure Solutions Limited, Mumbai									
1.	2.3.2 Technical Capacity:	years preceding the PDD, undertaken assignments	completed.	Tender condition prevails Please refer SI. No. 4 and 5 of Corrigendum No. 5.						

Page 1 of 25 **New Mangalore Port Authority**

S. No	Reference		Description	Bidders Query / Suggestion	Reply by NMPA
		has been exe And	e time period during which the work cuted and actual fees received.	Further the bidder may submit a self-undertaking for completion of the assignment along with notarized LOA/agreement copy with clear scope of work mentioned.	
			APPENDIX-I, ,Form-7 igible Assignments of the Applicant	Request the authority to kindly consider.	
			APPENDIX-I, Form-7A her relevant experience of the		
		And			
		Clause 2.12.2	2		
			order and completion certificate tof having successfully completed		
	2.	Page No. 18, 2.3.3. Financial Capacity:	2.3.3. Financial Capacity: The Applicant shall have received a minimum of Rs.1.00 crore (US \$ 0.13 million (US Dollar Zero Point One Three million) per annum as professional fees during each of the three financial years preceding the PDD. For the avoidance of doubt, professional fees hereunder refer to	Current financial criteria seem to be on lower side. The assignment requires preparation of Feasibility report, traffic assessment, analysis of PPP policy, captive policy etc. Hence a consultant with all the above skill sent shall be appointed by the Authority. Suggest the Authority to increase it to allow only experienced and capable consultants with significant consulting/Transaction Advisory experience.	Tender condition prevails.

New Mangalore Port Authority Page 2 of 25

S. No	Reference		Desc	ription			Bidders	Query / Sugge	stion	Reply by NMPA
			providing	•	he Applicant for or consultancy ts.					
3.	Page No. 18, Clause 2.3.4	Conditions	for Eligibili	ty of Key⊺	Personnel	the tende feel that with sign Transacti knowledg	reviewed the ase of the reducement and the Team leader of advisory of e etc. Hence we as given below	Please refer SI. No.7 of Corrigendum No. 5.		
		Key Personnel	Educati onal Qualific ation	Length of Profes sional Experi ence	Experience on Eligible Assignments	Key Person nel	Educational Qualification	Length of Professional Experience	Experience on Eligible Assignments	
		PPP Expert	Chart ered Acco untan t / MBA (Financ e) /ICWA / CFA	15 years	He should have led the financial team in Preparation of revenue Model, Bid Documents for not less	PPP Expert	Chartered Accountant / ICWA / CFA / MBA with B.E. or B. Tech (Civil/ Mech) / M Tech in	10 years	He should have led the financial team in Preparation of Feasibility report, TEFR, revenue Model, Bid Documents for not less than 5 (five) Public Private	

New Mangalore Port Authority Page 3 of 25

S. No	Reference		Desc	ription			Bidders	Query / Sugge	stion	Reply by NMPA
			or equival ent		than 5 (five) Public Private Partnership (PPP) project, out of which, at least 1 (one) should be a Port Sector PPP Project.		port related Studies		Partnership (PPP) projects, out of which, at least 1 (one) should be a Port Sector PPP Project.	
4.	Page No. 19, Clause 2.3.4	Conditions f	or Eligibili	y of Key	Personnel	the tende feel that with sign Transacti knowledg	er document and the Team leader nificant experience on advisory si	have understood position shall be been port feas tudies, Policy request the Au	ement from the TOR in the the expectation. We be given to a personal biblity related studies, matter expert, PPP of thority to kindly revise	
		Key Personnel	Educati onal Qualific ation	Length of Profes sional Experi ence	Experience on Eligible Assignments	Key Person nel	Educational Qualification	Length of Professional Experience	Experience on Eligible Assignments	

New Mangalore Port Authority
Page 4 of 25

S. No	Reference		Desc	ription			Bidders	Query / Sugges	stion	Reply by NMPA
5.	Page No. 19, Clause 2.3.4	Port Sector Expert Master / Bachelo r in Enginee ring or Equival ent Conditions for Eligibility of Key Personnel			Port Sector related studies / Ocean Engg./ Civil/ Mech/ Chemical) / Bachelor in Engineering (Civil/ Mechanical/ Chemical) / We have reviewed the assignment requirement from the TOR in the tender document and have worked as a sector expert in not less than 3 (three) Port Sector projects with minimum 2 Bulk Liquid handling facilities.				Please refer Sl. No.7 of	
						with sigr Transacti knowledg the criteri	nificant experience on advisory st le etc. Hence we a as given below:	ce in port feas audies, Policy request the Au	be given to a personal ibility related studies, matter expert, PPP ithority to kindly revise	
		Key Personnel	Educati onal Qualific ation	Length of Profes sional Experi ence	Experience on Eligible Assignments	Key Person nel	Educational Qualification	Length of Professional Experience	Experience on Eligible Assignments	

New Mangalore Port Authority Page 5 of 25

S. No	Reference	D	escription			Bidde	ers (Query / Sugge	stion		Reply by NMPA
		Law equiva ent	years el in or al	He should He should have worked as a Legal Consultant for issues relating to commercial contracts in not less than 3 (three) Port Sector projects.	Legal Consul tant	Law Contract related studies equivalent	/ in / or	5 years	project	to t	
6.	Page No. 34, 2.22	Substitution of Key I 2.22.2	Personnel C	ause 2.22.1 and	like to br personnel beyond th However and exper Further re	ing it to Aut resigning from e control of C the consultanti ienced person	horit m th onsu t has nnel horit	ty's notice that he services of the ultant, penalty s is to provide eq to the satisfact by to kindly limit	this clause. We in case of a ne Consultant, whould not be lever ally or better of the Authors the penalty am	ny Key vhich is vied. ualified rity.	Tender condition prevails.
7.	Page No. 38, 3. CRITERIA FOR EVALUATI ON	3.1.3 The scoring crit shall be as follows. B. Relevant Experien i. Length / status of A	ce of the Key	Personnel	Length / s	tatus of Assoc	ciatio	on with Firm for to evaluate key	the criteria rel any Key persor personnel onl in Assignment	nnel. y basis	Corrigendum No. 5.

New Mangalore Port Authority
Page 6 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
8.	Additional Clauses	We request the addition of the following clauses to the draft contract:	Anti-Bribery & Anti-Corruption: Each Party represents, warrants and undertakes that: (a) It has not and shall not offer, promise, give, encourage, solicit, receive or otherwise engage in acts of bribery or corruption in relation to this Agreement (including without limitation any facilitation payment), or to obtain or retain business or any advantage in business for any member of its group, and has and shall ensure to the fullest extent possible that its employees and agents and others under its direction or control and directly involved in providing Services under the Agreement do not do so. For the purposes of this clause it does not matter if the bribery or corruption is (i) direct or through a third party; (ii) of a public official or a private sector person; (iii) financial or in some other form; or (iv) relates to past, present, or future performance or non-performance of a function or activity whether in an official capacity or not, and it does not matter whether or not the person being bribed is to perform the function or activity to which the bribe relates, or is the person who is to benefit from the bribe. For the purposes of this clause, a "person" is any individual, partnership, company or any other legal entity, public or private. (b) Each Party shall, adhere to applicable anti-bribery and corruption laws. (c) Each Party shall, immediately upon becoming aware of them, give the other Party all details of any noncompliance with Clause (a) and Clause (b). (d) It is a condition of this Agreement that each Party fully complies with this Clause. If it does not do so, without prejudice to any other remedy available to a party, the non-breaching party shall have the right (but not the obligation) in its absolute discretion to terminate the whole of this Agreement, or that part of this Agreement to which the bribery or corruption relates. For the avoidance of doubt, any	of RFP.

New Mangalore Port Authority
Page 7 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
			breach of this Clause shall be deemed to be incapable of remedy.	
9.	Additional Clauses	We request the addition of the following clauses to the draft contract:	Non -Exclusivity: The Client acknowledges that Consultant or its associates may have other commercial transactions with the Client, other parties reviewed for the Client or referred in the agreement (if any) and the services provided under the agreement shall be on a non -exclusive basis.	Not agreed.
10.	Additional Clauses	We request the addition of the following clauses to the draft contract:	Economic and Trade Sanctions As of the date of this Agreement the Client warrants that, (a) neither Client nor any of its subsidiaries, or any director or corporate officer of any of the foregoing entities, is the subject of any economic or trade sanctions or restrictive measures issued by the United Nations, United States or European Union ("Sanctions"), (b) the Client is not 50% or more owned or controlled, directly or indirectly, individually or collectively, by one or more persons or entities that is or are the subject of Sanctions, and (c) to the best of Client's knowledge, no entity 50% or more owned or controlled by a direct or indirect parent of the Client, is the subject of Sanctions. For purposes of clause (c) in this section, "parent" is a person or entity owning or controlling, directly or indirectly, 50% or more of the Client. For so long as this Agreement is in effect, the Client will promptly notify CRISIL if any of these circumstances change, upon occurrence of which, CRISIL shall have the right to terminate the Agreement immediately in whole or in part for reasons of the Client's breach.	Not required.

New Mangalore Port Authority
Page 8 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
11.	Additional Clauses	We request the addition of the following clauses to the draft contract:	Client may notify the Consultant in writing within 10 calendar days of Consultant providing the Deliverables to the Client any substantial non - conformity of the Deliverables vis - à -vis the Requirements. Upon receiving Client's notification, the Consultant shall rectify the non -conformity verified by it and resubmit the Deliverables to the Client within 15 calendar days. The Deliverables shall be considered "accepted" upon such redelivery or the expiry of 10 calendar days as stated above, whichever is earlier.	Not agreed.
12.	Additional Clauses	We request the addition of the following clauses to the draft contract:	The total aggregate liability of Consultant, whether in contract, tort (including negligence) or otherwise, under or in connection with this agreement, shall in no circumstances exceed a sum equal to the fees paid or payable by the Client under this agreement. In no event will Consultant be liable for any consequential, incidental, indirect, punitive or special losses or damages (including loss of profits, data, anticipated savings, business or goodwill), regardless of whether such liability is based on breach of contract, tort, strict liability, breach of warranties, failure of essential purpose or otherwise.	Agreement Cl. 3.4, Liability of the Consultant,
13.	Additional Clauses	We request the addition of the following clauses to the draft contract:	The fees any amount Payable: - The fees and any amounts payable under this Agreement are exclusive of all applicable taxes (including GST), levies, duties etc. With regards to the applicability of Goods and Services Tax, the Client's address as mentioned for the purposes of GST will be considered as the consumption location for the Services provided by Consultant under this Agreement. The GST registration number ("GSTIN")	Please refer 5, Time and Payment Schedule, Sl. No.

New Mangalore Port Authority
Page 9 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA				
			provided by the Client will be used by Consultant for filing of the GST returns. With regards to the applicability of Goods and Services Tax, the Client's address as mentioned for the purposes of GST will be considered as the consumption location for the Services provided by Consultant under this Agreement. Where Consultant issues a credit note to the Client in relation to any invoice, the Client shall adjust and upload its Input Tax Credit on the GSTN on or before the end of the month in which the credit note is issued by Consultant to the Client. If the Client fails to do so, and this results in additional liability for Consultant, Client shall be liable to be reimburse Consultant for any liability incurred by Consultant (being the tax, interest and any penalties thereon). The current contract pricing are based on an assumption that GST will apply to the services provided by the Consultant to the Client and the consultant is able to claim credit of the GST charged by its partners, vendors, sub-consultants. In the event that such assumption is incorrect and Consultant is not able to claim GST credit for the services provided to it by vendors, partners or sub-consultants, the consultant reserves its rights to recover from the Client an amount equivalent to 18% or prevailing GST rate on such statement values to the Consultant.					
II.	M/s. RITES Limited, Gurugram, Haryana							
1.	Clause 2.3.2 Technical Capacity:	The Applicant shall have, over the past 7 (seven) years preceding the PDD, undertaken assignments as specified in Clause 3.1.4. The Consultant shall have to substantiate the above experience with a	completion of assignment, it is requested that proof of completion should also include Letter of Award or Signed Agreement with	Tender condition prevails.				

New Mangalore Port Authority Page 10 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
		work order and Completion certificate from the Client.		
2.	Clause 2.3.3 Financial Capacity:	The Applicant shall have received a minimum of Rs. 1.00 crore (US \$ 0.13 million)	Authority may kindly confirm if the requirement of Rs. 1 Crore is true. The financial capacity mentioned herein is very less and this will allow very small consultant with less experience.	Tender condition prevails.
3.	Terms of Reference, Para 3.1, Pg. 50	the Government of India / statutory authorities from	Government of India / statutory authorities from time to time and suggests any action to be taken by Port. It is inferred that the policies/guidelines related to PPP in Port sector will be scrutinise by Transaction Advisor. Kindly confirm.	Please refer Corrigendum No. 5, Sl. No. 10.
4.	Terms of Reference.	2. Preparation of Techno Economic Feasibility Report	The Authority may kindly clarify if TA is to prepare a fresh TEFR including hydrographic profile, bathometric surveys, etc., or the scope would be only review of Feasibility report already prepared by Authority. If review is to be undertaken, kindly specify what aspects are to be reviewed.	

New Mangalore Port Authority Page 11 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
				required for handling liquid bulk / LPG / POL products, land required for providing pipelines and backup area, power & water supply, fire fighting facilities, safety & environmental issues, Traffic Projections, Financial Projections / CAPEX and O&M expenses etc.
5.	Terms of Reference.	Captive/PPP	The Authority may also clarify which mode of implementation the Transaction Advisor should assist the authority. The Feasibility should be assessed keeping both modes in mind or purely on PPP as for Captive the analysis would be limited to captive cargoes of certain industries operating in the region.	assessed keeping both
6.	Reference -	In case the viability gap funding projected in the Appraisal Report exceeds the projection in the Inception Report by more than 10% of the capital costs, a deduction of 5% (five per cent) of the Agreement Value shall be made from the payment.	Inception Report shall not cover the projected VGF amount but would rather provide methodology for undertaking the assignment. The VGF (if any) would be worked out only in Appraisal Report. Considering this Authority should remove this condition.	

New Mangalore Port Authority Page 12 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA								
7.	Terms of Reference - Para 5.2:	Time period for KD 2 & 3	The time period for feasibility report (KD2) and EOI (KD3) is kept at 8 weeks. The minimum time required for KD2 (in case a fresh report is to be prepared) in minimum 12 weeks and thereafter minimum 2 weeks are required for EOI (EoI will contains the details of Project as finalised in TEFR). Considering this and we request that for KD2 & KD3 at least 14 weeks should be kept.	Tender condition prevails.								
III.	M/s. Resurgent India Limited, Gurgaon, Haryana											
1.	-	JV/ consortium	Whether JV / consortium is allowed.	JV is not permitted.								
2.	Clause 2.3.4 page no. 19	Legal Consultant. He should have worked as a Legal Consultant for issues relating to commercial contracts in not less than 3 (three) Port Sector projects.	We request you to please read as: Legal Consultant. He should have worked as a Legal Consultant for issues relating to commercial contracts in not less than 3 (three) Port Sector projects / core sector project.	Please refer Corrigendum No. 5, Sl. No. 7.								
3.	Clause 2.3.4 page no. 19	Associate Consultant: He should have worked as member of a Transaction Advisory team in preparation of Revenue Model and/or Bid Documents for at least 2 (two) Port Sector projects.	We request you to please read as : Associate Consultant : He should have worked as member of a Transaction Advisory team in preparation of Revenue Model and/or Bid Documents for at least 2 (two) Port Sector projects / core sector projects.	Please refer Corrigendum No. 5, Sl. No. 7.								

New Mangalore Port Authority Page 13 of 25

S. No	Reference	Descri	ption		Bidders Query /	Suggestion		Reply by NMPA		
4.	Clause 2.23 Page no. 35	Indemnity:	we request you to please read as : Indemnity:							
		The Consultant shall, sub the Agreement, indemnif amount not exceeding 3 (the Agreement, for any di is caused due to any defici	nt, ne) ge							
5.	Clause 3.1.4 page no. 41	Provided that all Eligible projects based on the s Agreement and awarded the shall be counted as one El	ame Model by the same	Concession public entity	authority and the said authority issu	of No. 5, Sl. No. 6.				
6.	Clause 5.2 page no. 55	Submission of Inception R	eport 2 Weeks	5 %	We request to modify as	We request to modify as				
		Report			Submission of Inception Report	8 Weeks	10 %			
		Submission of Feasibility Report	8 weeks	5 %	Submission of Feasibility Report and Revenue Model	8 weeks	10 %			
		and Revenue Model Submission of EOI	8 Weeks	5 %	Submission of EOI document for approval and invitation of EOI	8 Weeks	5 %			
		document for approval and invitation of EOI	OVVEERS	3 /6	Evaluation of EOI applications and concluding EOI process	20 Weeks	5 %			
		Evaluation of EOI	20 Weeks	5 %						

New Mangalore Port Authority Page 14 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
		applications and concluding EOI process		
7.	Clause 4.4 page no.86	Substitution of Key Personnel The Authority expects all the Key Personnel specified in the Proposal to be available during implementation of the Agreement. The Authority will not consider any substitution of Key Personnel except under compelling circumstances beyond the control of the Consultant and the concerned Key Personnel. Such substitution shall be limited to not more than two Key Personnel subject to equally or better qualified and experienced personnel being provided to the satisfaction of the Authority. Without prejudice to the foregoing, substitution of Key Personnel shall be permitted only upon reduction of remuneration weightage specified for the original Key Personnel in the scoring criteria as per clause 3.1.3 shall be deducted from the payments due to the Consultant.	authority to relax the criteria as the completion of the work comes	Tender condition prevails.
		The Consultant should specifically note that substitution of the Team Leader cum PPP Expert will not normally be considered during the implementation of the Agreement and may lead to disqualification of the Applicant or termination of		

New Mangalore Port Authority Page 15 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
		the Agreement. The PPP Expert must commit the time required for and be available for delivering the Consultancy in accordance with the terms specified herein.		
IV.	M/s. Capital	Fortunes Private Limited (CFL)		
1.	Page No 18, Clause 2.3.3; Financial Capacity	Financial Capacity: The Applicant shall have received a minimum of Rs.1.00 crore (US \$ 0.13 million (US Dollar Zero Point One Three million) per annum as professional fees during each of the three financial years preceding the PDD.	Request you to please minimize the amount of Rs 1 Cr per annum from professional fees in last 3 FYs to Rs 50 lakhs?	Tender condition prevails.
2.	Page No 18, Clause 2.3.4, Key Personnel – PPP Expert	Experience on Eligible Assignments: He should have led the financial team in Preparation of revenue Model, Bid Documents for not less than 5 (five) Public Private Partnership (PPP) projects, out of which, at least 1 (one) should be a Port Sector PPP project.	Request to please minimize the no of eligible PPP projects from 5 to 2 projects.	Please refer Corrigendum No. 5, Sl. No. 7.
3.	Page No 19, Clause 2.3.4, Key Personnel –	Experience on Eligible Assignments: He should have worked as a Legal Consultant for issues relating to commercial contracts in not less	Request to please consider that the legal expert needs to have experience for issues relating to commercial contracts in not less than 1 Port Sector project;	Please refer Corrigendum No. 5, Sl. No. 7.

New Mangalore Port Authority Page 16 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
	Legal Consultant	than 3 (three) Port Sector projects.	Or in not less than any 3 infrastructure projects	
4.	Page No 19, Clause 2.3.4, Key Personnel – Port Sector Expert	Experience on Eligible Assignments: He should have worked as a sector expert in not less than 3 (three) Port Sector projects preferably Bulk Liquid handling facilities.	Request to please consider that the expert needs to have experience in any Port related assignments (for e.g., Bulk Loading, Container Terminals, Multi Cargo, Coal Terminals etc) instead of only Bulk liquid facilities.	Please refer Corrigendum No. 5, Sl. No. 7.
5.	Page No. 41, Clause 3.1.4, Eligible Assignment s	Eligible Assignments: For the purposes of satisfying the Conditions of Eligibility and for evaluating the Proposals under this RFP, Consultancy / Transaction Advisory Services assignments granted by the government, regulatory commission, tribunal, statutory authority or public sector entity in respect of preparation of revenue model and/or preparation of transaction / bid documents and other similar assignments in relation to the bidding process or other transactions, as the case may be, for the following types of projects, for which the Bidder over the past 7 (seven) years preceding the PDD shall have	Request to please consider in modify - the experience of eligible assignments/projects over last 10 years instead last 7 years.	Please refer Corrigendum No. 5, Sl. No. 6.

New Mangalore Port Authority Page 17 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
		successfully completed atleast 5 (five) assignments shall be deemed as eligible assignments (the "Eligible Assignments")		
V.	M/s. Grant Ti	hornton Bharat LLP.		
1.	Pg. 59, Requiremen t of Team leader	The requirement is for the Team leader to be at site 15 working days in a month till completion of the services. The contract duration is 76 weeks.	It is difficult to get good talent with the stated requirement for him to be at site 15 days a month for almost 1.5 years. Our humble submission is to drop this requirement so that we can tap on to the talent from across the country. He will of course travel to NMPT as and when you will require his services, but binding the person to be at site for 15 working days a month is likely to dissuade experienced PPP experts.	Please refer SI. No. 12 of Corrigendum No. 5.
2.	Pg. 59, Port Sector Expert	The requirement is for the Port Sector Expert to be at site 7 working days in a month till completion of the services. The contract duration is 76 weeks.	Same as above. Our humble submission is to drop this requirement.	Please refer SI. No. 12 of Corrigendum No. 5.
3.	Pg. 60, Legal Consultant	The requirement is for the Legal consultant to be at site 7 working days in a month till completion of the services.	Same as above. Our humble submission is to drop this requirement.	Please refer SI. No. 12 of Corrigendum No. 5.

New Mangalore Port Authority Page 18 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
		The contract duration is 76 weeks.		
VI.	M/s. Darasha			
1.	Joint venture	2.1.1 In case an applicant firm possesses the requisite experience and capabilities required for undertaking the Consultancy, it may participate in the Selection Process either individually (the "Sole Firm") or as lead member of a consortium of firms (the "Lead Member") in response to this invitation.	Kindly clarify the number of members allowed to participate as consortium? (Lead+1 member) OR (Lead + 2 member) We request you to allow atleast 3 members in consortium.	JV is not permitted.
2.	2.3 Conditions of Eligibility of Applicants	2.3.2 Technical Capacity	We request authority to kindly allow consortium members to jointly meet the technical capacity under clause 2.3.2	Not agreed.
3.	3. Criteria For Evaluation	3.1 Evaluation of Technical Proposals: A Relevant Experience of the Applicant	We request authority to kindly allow consortium members to jointly meet the relevant experience technical under clause 3.1 (A)	Not agreed.
4.	3. Criteria For Evaluation	3.1.3 The scoring criteria to be used for evaluation shall be as follows: Relevant Experience of the Applicant	While the scoring criteria mentions as more than 10 project gets 15 marks; Whereas statement below the evaluation table says "While awarding marks for the number of Eligible Assignments, the	Please refer SI. No. 8 & 9 of Corrigendum No. 5.

New Mangalore Port Authority Page 19 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
		More than10 Projects- 15 marks 9-10 projects-12.5 marks 6-8 projects- 11 marks	Applicant or Key Personnel, as the case may be, that has undertaken the highest number of Eligible Assignments shall be entitled to the maximum score for the respective category and all other competing Applicants"	
		5 projects- 10.5 marks	This statement contradicts the Table above in which marks given for more than 10 projects are 15.	
			We request authority to kindly clarify the marking system and also we further request authority to not keep marking on relative basis but marks should be given in objective manner against number of eligible projects.	
5.	3.1.3 The scoring criteria	No score will be awarded to an Applicant/ Key Personnel for fulfilling the eligibility criteria of a minimum number of Eligible Assignments and only projects exceeding the eligibility criteria shall qualify for scoring.	We request authority to consider assignment mentioned in pre-qualification 2.3.2 and also consider while technical evaluation criteria 3.1.3	Please refer SI. No.7 & 8 of Corrigendum No. 5.
6.	7. CONSULT ANCY TEAM	Key Personnel: Legal Consultant He should have worked as a Legal Consultant for issues relating to commercial contracts in not less than 3 (three) Port Sector projects.	as: He should have worked as a Legal Consultant for issues	Please refer SI. No.7 & 8 of Corrigendum No. 5.

New Mangalore Port Authority Page 20 of 25

S. No	Reference		Descri	ption			Bidders Query / \$	Suggestion			Reply by NMPA
7.	7. CONSULT ANCY TEAM	should Transac Revenue	ersonnel: Asso have worked tion Advisory to e Model and/or two) Port Sector	as me eam in p Bid Doc	uments for at	We request authority to kindly relax the criteria for project as: He should have worked as member of a Transaction Advisory team in preparation of Revenue Model and/or Bid Documents for at least 1 (One) Port Sector/PPP projects.					Please refer SI. No.7 & 8 of Corrigendum No. 5.
8.	B Relevant Experience of the Key Personnel		GAL CONSUL ⁻ nent /Job iv. Exp assignment.		Adequacy in n carrying out	, , , , , , , , , , , , , , , , , , , ,					Please refer SI. No.7 & 8 of Corrigendum No. 5.
.8.	5. Time And Payment					We request you to consider the following timelines and payment terms.					Tender condition prevails.
	Schedule	Stage	Description	weeks	Payment %	Stage	Description	Timeline (weeks)	Payment %	·	
		KD1	Submission of Inception	4	5%	KD1 KD2	Submission of Inception Report Submission of	6	5% 15%		
		KD2	Report Submission of Feasibility Report and Revenue Mode	12	15%	KD3	Feasibility Report and Revenue Model Submission of EOI document for	20 15%			
		KD3	Submission of EOI document	12	5%		approval and invitation of EOI				

New Mangalore Port Authority Page 21 of 25

S. No	Reference		Descrip	tion			Bidders Query / St	uggestion		Reply by NMPA
			for approval and invitation of EOI			KD 4	Evaluation of EOI applications and concluding EOI	26	15%	
		KD 4	Evaluation of EOI applications and concluding	24	10%	KD 5	Submission of RFQ for approval And invitation of RFQ	32	10%	
		KD 5	Submission of RFQ for approval And invitation of	28	10%	KD 6	Submission of a sustainable Royalty/ Revenue Share model for PPP Project	36	10%	
		KD 6	RFQ Submission of a sustainable Royalty/	28	10%	KD 7	Evaluation of RFQ applications and Concluding of RFQ process	48	5%	
			Revenue Share model for PPP Project			KD 8	Submission of documents for Security clearance of	48	5%	
		KD 7	Evaluation of RFQ applications and Concluding of RFQ process	44	5%	KD 9	bidders Submission of RFP and Draft Concession Agreement for	52	3%	
		KD 8	Submission of documents for	44	5%		approval			
			Security clearance of			KD 10	Submission of appraisal report to	58	3%	

New Mangalore Port Authority Page 22 of 25

S. No	Reference		Descrip	otion			Bidders Query / Si	uggestion		Reply by NMPA
			bidders				PPPAC/SFC/MoS			
		KD 9	Submission of RFP and Draft Concession Agreement for	44	10%	KD 11	Issue of RFP including preparation of Draft Concession Agreement	64	3%	
		KD 10	approval Submission of appraisal report	50	5%	_ KD 12	Conclusion of Bid Process and issue of LoA	68	3%	
			to PPPAC/SFC/M oS			KD 13	Signing of Concession Agreement	72	3%	
		KD 11	Issue of RFP including preparation of Draft	54	5%	KD 14	Completion of Services till financial close	100	5%	
			Concession Agreement			TOTAL	A		100%	
		KD 12	Conclusion of Bid Process and issue of LoA	54	5%	also consider to go for dire	Authority to simplify the that if EOI stage is skingled to the payment of adjacent so	ipped and ent of that	Authority deci stage to be m	ides ade
		KD 13	Signing of Concession Agreement	62	5%	submissions,	approvals etc.)	- '	-	
		KD 14	Completion of Services till financial close	88	5%					

New Mangalore Port Authority Page 23 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
		TOTAL 100%		
10.		Proposal Due Date (PDD) Closing date of online etender for submission of Techno- Commercial Bid & Price Bid 30.05.2022 at 15:30 HRS	We request client to provide minimum Three Weeks extension in bid submission date from date of reply to pre bid queries/Issuing Corrigendum/Addendum with changes in terms and conditions to the tender document.	_
VII.	M/s. Gianeno	der & Associates, New Delhi		
1.	Clause 3.1.3	-	As per clause 3.1.3 mentioned No score will be awarded to an Applicant/ Key Personnel for fulfilling the eligibility criteria of a minimum number of Eligible Assignments and only projects exceeding the eligibility criteria shall qualify for scoring. As per clause 3.1.4 bidder have successfully completed atleast 5 (five) assignment shall be deemed as eligible assignments. And as per column no. 4 "criteria"A(i) of clause 3.1.3 mentioned 5 projects scoring 10.5 marks . Please clarify.	Please refer Corrigendum No. 4, Sl. No. 7, 8 & 9.

New Mangalore Port Authority Page 24 of 25

S. No	Reference	Description	Bidders Query / Suggestion	Reply by NMPA
2.	Clause 3.1.5,	Form No. 7 (A), 8 (A), 9(A) and 10(A)	Other relevant experience required in Form No. 7 (A), 8 (A), 9(A) and 10(A) in which reference has made to 3.1.5, whereas Clause 3.1.5 mentioned " Deleted "	Please refer Corrigendum No. 4, Sl. No. 4.
			Please clarify the same.	
3.	-	-	We attach our MSME certificate for claiming exemption of Earnest Money Deposit (EMD) and tender processing fee.	Will be considered.
			Please Clarify.	

New Mangalore Port Authority Page 25 of 25

NEW MANGALORE PORT AUTHORITY CORRIGENDUM- 5

Date: 25.07.2022

Name of work: Request for Proposal (RFP) for Appointment of Transaction Advisor for Concessioning of Berth No. 13 for handling Liquid Bulk

/ LPG / POL products / Chemicals on Captive / PPP Mode – Reg.

NIT No.: CIVIL/CE(C)/EE(C)/05/2022-23 dated 29.04.2022 and eProcure Tender ID: 2022_NMPT_686079_1.

New Mangalore Port Authority ("NMPA" or the "Authority") by way of this Corrigendum – 5 is making the following corrections to the Request for Proposal (RFP) and subsequent Corrigendum's issued for the subject Project.

S.No.	Clause No.	Original Clause	Modified Clause
1	Schedule – 2, Page 90 Cause 7.1, Performance Security	7.1 Performance Security Within ten days of the receipt of the award letter from Employer, the successful bidder/s shall furnish to the Employer a performance security in the form of a Bank Guarantee of a Nationalized/ Scheduled Bank, enforceable and encashable at Mangalore for an amount equivalent to 10% of the awarded cost/ fee including GST rounded off to nearest highest 1000 as per the draft proforma specified at Annexure-1 of the RFP document.	This clause shall be read as follows:- 7.1 Performance Security Within twenty one (21) days of the receipt of the award letter from Employer, the successful bidder/s shall furnish to the Employer a performance security in the form of a Bank Guarantee of a Nationalized/ Scheduled Bank, enforceable and encashable at Mangalore for an amount equivalent to 3% of the awarded cost/ fee including GST rounded off to nearest highest 1000 as per the draft proforma specified at Annexure-1 of the RFP document.
2	Page 73. Cl. 2.7.1 (a)	1.2.7.1 Definition	This clause shall be read as follows:-
		(a) For the purposes of this Agreement, "Force Majeure" means an event which is beyond the reasonable control of a Party, and which makes a Party's performance of its obligations hereunder impossible or so impractical as	1.2.7.1 Definition(a) For the purposes of this Agreement, "Force Majeure" means an event which is beyond the reasonable control of a Party, and which makes a Party's performance of its obligations hereunder

S.No.	Clause No.	Original Clause	Modified Clause
		reasonably to be considered impossible in the circumstances, and includes, but is not limited to, war, riots, civil disorder, earthquake, fire, explosion, storm, flood or other adverse weather conditions, strikes, lockouts or other industrial action (except where such strikes, lockouts or other industrial action are within the power of the Party invoking Force Majeure to prevent), confiscation or any other action by government agencies.	impossible or so impractical as reasonably to be considered impossible in the circumstances, and includes, but is not limited to, war, riots, civil disorder, earthquake, fire, lockdown or quarantine restrictions due to pandemic/ epidemic, explosion, storm, flood or other adverse weather conditions, strikes, lockouts or other industrial action (except where such strikes, lockouts or other industrial action are within the power of the Party invoking Force Majeure to prevent), confiscation or any other action by government agencies.
3	Page No. 1, Notice Inviting Tender, Section 1, Proposal Due Date (PDD)	Proposal Due Date (PDD) Closing date of online e-tender for submission of Techno- Commercial Bid & Price Bid: 29.06.2022 at 15:30 HRS.	Proposal Due Date (PDD) Closing date of online e-tender for submission of Techno-Commercial Bid & Price Bid shall be extended by three weeks from the date of uploading of Reply to pre-bid queries.
4.	i) Appendix – I Form – 7A (Page 114) ii) Form – 8A (Page 117) iii) Form – 9A	 Form – 7A Abstract of other relevant experience of the Applicant (Refer Clause 3.1.5) Form – 8A Abstract of other relevant experience of Key Personnel (Refer Clause 3.1.5) Form – 9A Other relevant assignments of Applicant (Refer Clause 3.1.5) Form – 10A Other relevant assignments of Key 	The following forms appearing in the respective page Nos. of the RFP document shall be deleted. Page 114, Form–7A: Abstract of other relevant experience of the Applicant Page 117, Form–8A: Abstract of other relevant experience of Key Personnel Page 119, Form–9A: Other relevant assignments of Applicant Page 121, Form–10A: Other relevant assignments of Key Personnel.

New Mangalore Port Authority Page 2 of 22

S.No.	Clause No.	Original Clause	Modified Clause
	(Page 119) and iv) Form – 10A (Page 121)	Personnel (Refer Clause 3.1.5)	
5.	2, Instruction to Applicants, SI. No. 2.3 Condition of Eligibility of Applicants , Page 18.	2.3.1 Technical Capacity: The Applicant shall have, over the past 7 (seven) years preceding the PDD, undertaken assignments as specified in Clause 3.1.4. The Consultant shall have to substantiate the above experience with a work order and Completion certificate from the Client of having successfully completed the works indicating the time period during which the work has been executed and actual fees received.	
6.	3, Criteria for Evaluation, 3.1.4, Eligible Assignments, Pg. 41 of RFP.	3.1.4 Eligible Assignments For the purposes of satisfying the Conditions of Eligibility and for evaluating the Proposals under this RFP, Consultancy / Transaction Advisory Services assignments granted by the government, regulatory commission, tribunal, statutory authority or public sector entity in respect of preparation of revenue model and/or preparation of transaction / bid documents and other similar assignments in relation to the bidding process or other transactions, as the case may be, for the following types of projects, for which the Bidder over the past 7 (seven) years preceding the PDD shall have	3.1.4 Eligible Assignments For the purposes of satisfying the Conditions of Eligibility and for evaluating the Proposals under this RFP, Consultancy / Transaction Advisory Services assignments granted by the government, regulatory commission, tribunal, statutory authority or public sector entity in respect of preparation of revenue model and/or preparation of transaction / bid documents and other similar assignments in relation to the bidding process or other transactions, as the case may be, for which the Bidder over the past 10 (ten) years preceding the PDD shall have successfully completed atleast 3 (three) assignments for the following types of projects, shall be deemed as eligible assignments (the "Eligible Assignments"):

New Mangalore Port Authority Page 3 of 22

S.No.	Clause No.	Original Clause	Modified Clause
		successfully completed atleast 5 (five) assignments shall be deemed as eligible assignments (the "Eligible Assignments"): (a) An infrastructure project undertaken through Public Private Partnership (PPP) or other forms of private participation and having an estimated capital cost (excluding land) of at least Rs. 100 crore (Rupees One hundred crore only) in case of a project in India, and [US \$ 50 million for infrastructure projects undertaken elsewhere] through PPP or other forms of private participation (the "PPP Projects"); OR (b) Privatization or disinvestment of government owned companies in India involving transfer of management control to a private sector entity where the financial transaction involves a payment of at least Rs.50 crore (Rupees Fifty crore) for such transfer (the "Disinvestment Projects"); OR (c) An infrastructure project having an estimated capital cost (excluding land) of at least Rs. 100 crore (Rupees One hundred) in case of a project in India, and US \$ 50 million projects elsewhere, provided that the payment of professional fee to the Applicant was at least Rs.5 lakh (Rupees five lakh)] in case of a project in India, and [US \$ 25,000 (US\$ twenty five thousand) for any infrastructure project elsewhere (the "Other Projects");	 (a) Infrastructure project undertaken through Public Private Partnership (PPP) or other forms of private participation and having an estimated capital cost (excluding land) of at least Rs. 100 crore (Rupees One hundred crore only) in case of a project in India, and [US \$ 50 million for infrastructure projects undertaken elsewhere] OR (b) Privatization or disinvestment of government owned companies in India involving transfer of management control to a private sector entity where the financial transaction involves a payment of at least Rs.50 crore (Rupees Fifty crore) for such transfer (the "Disinvestment Projects"). Note: The value of the project executed shall be brought to current costing level by enhancing the actual value of the work upon completion by using the following enhancement factors. For intermediate periods, the actual number of years will be calculated based on number of days and the enhancement / multiplying factor will be interpolated accordingly and the same will be considered for evaluation.

New Mangalore Port Authority Page 4 of 22

S.No.	Clause No.	Original Clause		N	Modified Clause	
		Provided that all Eligible Assignments relating to projects based		Year before	Enhancement factor / Multiplication factor	
		on the same Model Concession Agreement and awarded by the same public entity shall be counted as one Eligible		One year	1.07	
		Assignment.		Two years	1.14	
				Three years	1.21	
				Four years	1.28	
				Five years	1.35	
				Six years	1.42	
				Seven years	1.49	
				Eight years	1.56	
				Nine years	1.63	
7.	2. Instruction to Applicants, 2.3.4, Conditions of Eligibility	2.3.4 Conditions of Eligibility of Key Personnel (B) Each of the Key Personnel must fulfill the Conditions			y of Key Personnel nel must fulfill the Conditions of	f Eligibility
	of Key Personnel, Page No. 18 & 19.	of Eligibility specified below:	specified		- •	

Key Perso nnel	Educational Qualification	Length of Profession al Experience	Experience on Eligible Assignments
PPP Expert	Chartered Accountant	15 years	He should have led the financial team
-	/MBA		in preparation of

Key Personnel	Educational Qualification	Length of Professional Experience	Experience on Eligible Assignments
PPP	Chartered	15 years	He should have led
Expert-	Accountant		the financial team in
cum-Team	/ICWA / CFA		preparation of
Leader	/ MBA		revenue model, bid

S.No.	Clause No.		(Original C	Clause		Modif	ied Claus	se
			(Finance) / ICWA / CFA or equivalent		revenue Model, Bid Documents for not less than 5 (five) Public Private Partnership (PPP) project, out of which, at least 1 (one) should be a Port Sector PPP project.		(Finance) with Bachelors or Masters in Engineering		documents for atleast 3 (three) Public Private Partnership (PPP) projects, out of which, 1 (one) project should be a Port Sector PPP project.
		Port Sector Expert	Master / Bachelor in Engineering or Equivalent	10 years	He should have worked as a sector expert in not less than 3 (three) Port Sector projects preferably Bulk Liquid handling facilities.	Port Sector Expert	Masters or Bachelors degree in Engineering or Equivalent	10 years	He should have worked as a sector expert for 2 (two) Port Sector projects, out of which 1 project should be Bulk Liquid handling facilities.
		Legal Consu Itant	Master / Bachelor in Law or equivalent	5 years	He should have worked as a Legal Consultant for issues relating to commercial contracts in	Legal Consultant	Masters / Bachelors degree in Law or	5 years	He should have worked as a Legal Consultant for issues relating to commercial contracts in a

New Mangalore Port Authority Page 6 of 22

S.No.	Clause No.			Origina	Clause					M	odifie	d Claus	se .		
		Aggagi	Mostoro	E voor	Port S	ess than 3	ets.			Equivalent				or atleast	` ′
		Associ de la			Associate Consultant Bachelors degree in Engineering or Finance						worked of a Advisory ation of and/or for at				
8.	3, Criteria for Evaluation, 3.1, Evaluation of Technical Proposals, Page 37 to		he scoring cri llows.	teria to be	e used for	evaluation s	shall be as	3.1	I.3 The so	oring criteria to b	e used	d for eva	projects		s follows.
	40	Item Code		Max. Mark s	Criteria	Scoring	Remun eration weighta ge for Key Person nel in percent age (ref. Cl.		Item Code		Max. Marks	Cri	iteria	Scoring	Remune ration weighta ge for Key Personn el in percent age (ref. CI.

S.No.	Clause No.			Orig	inal Clause					r	Modified	d Clause		
		A	Relevant Experience of the Applicant	25			2.22.2)	A	1	Relevant Experience of the Applicant	25			2.22.2)
		i)		15	More than10 Projects 9-10	15 12.5			i)	Number of Eligible Assignments undertaken by the Applicant, over the	15	More than 5 Projects	15	
			undertaken by the		projects 6-8 projects	11				past 10 (Ten) years preceding to PDD				
			Applicant shall have, over the past 7		5 projects	10.5				proceduring to 1 BB		4-5 projects	12.75	
			(Seven) years preceding to									3 projects	10.5	
		ii)	Overall turnover of	10	More than 5 Crore	10			ii)	Overall turnover of the firm	10	More than 2 Crore	10	
			the firm		>2 Cr. upto 5 Crore	8.5						>1 Cr. upto 2	7	
					>1 Cr. upto 2 Crore	7						Crore		
			Proposed Methodolo gy and works plan	5	Evaluation based on quality of submission									

New Mangalore Port Authority Page 8 of 22

S.No.	Clause No.			Orig	inal Clause				ı	Modified	Clause		
		В	Relevant Experience of the Key Personnel	70					Proposed Methodology and works plan	5	Evaluation based on quality of submission		
		i	TEAM LEADER - PPP Expert i. Length /	25 5.0	Permanent	5.00	25%	В	Relevant Experience of the Key Personnel	70			
			status of Association with Firm	0.0	Expert with more than 15 years experience	0.00		i	TEAM LEADER - PPP Expert	25			25%
					or an Independen t Expert with more than 20years experience				i. Length / status of Association with Firm	5.0	Permanent Expert with more than 15 years experience with the Firm or an Independent	5.00	
					Permanent Expert with 10-15 years experience or an	4.25					Expert with more than 20years experience		
					Independen t Expert with 15 or more than 15years						Permanent Expert with more than 10 years experience	4.25	

New Mangalore Port Authority Page 9 of 22

S.No.	Clause No.	Orig	inal Clause	Modified Clause		
			experience Permanent 3.50 Expert with 5-10 years experience or an Independen t Expert	with the Firm or an Independent Expert with more than 15years experience		
		ii. 5.0 Qualification	t Expert with 10 or more than 10 years experience CA/ICWA/C 5.00 FA/ MBA(Fin) or equivalent	Permanent 3.50 Expert with 5 years experience with the firm or an Independent Expert with more than 10years experience		
			equivalent	ii. Qualification 5.0 CA/ICWA/ CFA or equivalent		
				MBA(Fin) 4.25 with Masters in Engineering		
		iii. Total 5.0 Work	20yrs	MBA(Fin) 3.50		
		Experience	16-20 yrs 4.25 15 yrs 3.50	with Bachelors in		

New Mangalore Port Authority Page 10 of 22

S.No.	Clause No.		Orig	inal Clause				N	odified	Clause		
		Adequacy in Assignmen t /Job iv. Experience	10.0	>10 Projects 9-10 projects 6-8 projects 5 projects	10.00 8.34 7.34 7.00			iii. Total Work Experience Adequacy in	5.0	20yrs 16-20 yrs 15 yrs More than 5	5.00 4.25 3.50 10.00	
		in carrying out Eligible assignment ii PORT SECTOR EXPERT		, ,		20%		Assignment /Job iv. Experience in carrying out Eligible assignment		projects (with atleast 1 port sector project) More than 3 projects upto	8.50	-
		i. Length / status of Association with Firm	2.50	Permanent Expert with more than 10 years	2.50					5 projects (with atleast 1 port sector project)		
				experience with the firm or an Independen t Expert						Minimum 3 projects (with atleast 1 port sector project)	7.00	
				with 10-15 years experience			ii	PORT SECTOR EXPERT i. Length / status of	20 5.0	Permanent	5.00	20%
				Permanent Expert with 5-10 years experience with the firm or an	2.125			Association with Firm	0.0	Expert with more than 10 years experience with the firm	0.00	

New Mangalore Port Authority Page 11 of 22

S.No.	Clause No.	Origi	inal Clause	Modified Clause	
			Independen t Expert with more than 10 years		or an Independent Expert with 15 years experience
			experience Permanent 1.75 Expert with less than 2-5 years experience with the firm or an Independen t Expert with more than 5 years		Permanent 4.25 Expert with more than 5years experience with the firm or an Independent Expert with more than 10 years experience Permanent 3.50
		ii. 2.50 Qualification	experience PhD / 2.50 Master		Expert with more than 2years experience
		iii. Total 5.00 Work Experience	more than 5.00 15 yrs 11-15 yrs 4.25 10 yrs 3.50		with the firm or an Independent Expert with
		Adequacy in 0 Assignmen 10.0	More 10.00 than10 Projects		more than 5 years experience

New Mangalore Port Authority Page 12 of 22

S.No.	Clause No.		Orig	inal Clause				ı	Modified	l Clause		
		t /Job		7-10	9.00			ii. Qualification	5.00	Master	5.00	
		Experience		projects 4-6 projects	8.00					Bachelor	3.50	
		in carrying out Eligible		3 projects	7.00			iii. Total Work Experience	5.00	more than 15 yrs	5.00	
		assignment								11-15 yrs	4.25	
		iii LEGAL	15.0			15%				10 yrs	3.50	
		CONSULT ANT	0					Adequacy in Assignment /Job	5.00	More than 4 Projects	5.00	
		i. Length /	2.50	Permanent	2.50			iv. Experience in		3-4 projects	4.25	
		status of Association		Expert with more than				carrying out Eligible assignment		2 projects	3.50	
		with Firm		5 years experience			iii	LEGAL CONSULTANT	15.00			15%
				with the firm or an Independen t Expert with more than 10 years experience Permanent Expert with 2-5 years experience	2.125			i. Length / status of Association with Firm	2.50	Permanent Expert with more than 10 years experience with the firm or an Independent Expert with more than 15 years experience	2.50	
				with the firm or an						Permanent	1.75	
				Independen t Expert						Expert with more than 5		

New Mangalore Port Authority Page 13 of 22

S.No.	Clause No.	Or	Original Clause			ı	Modified	Clause		
			with more than 5 years experience Permanent Expert with less than 2 years experience with the firm or an Independen t Expert with more than 2 years experience	1.75				years experience with the firm or an Independent Expert with more than 10 years experience		
		ii. Qualification	DO PhD / Master / Bachelor in law or any equivalent degree in law.	2.50		ii. Qualification	2.50	Master in	2.50	
		iii. Total 5.0 Work Experience		5.00 4.25 3.50		434343	2.00	Law or equivalent	2.00	
		Adequacy 5.0 in		5.00						

New Mangalore Port Authority Page 14 of 22

S.No.	Clause No.	Oriç	ginal Clause	Modified Clause
		Assignmen t /Job iv. Experience	4- 5 4.25 projects 3.50	Bachelor in Law or equivalent
		in carrying out Eligible assignment		iii. Total Work 5.00 more than 10 5.00 yrs
		iv ASSOCIA 10.0 TE 0	109	6-10 yrs 4.25 5 yrs 3.50
		CONSULT ANT	Dormor ont 250	Adequacy in S.00 More than 3 5.00 Assignment /Job Projects
		i. Length / 2.50		iv. Experience in 2- 3 projects 4.25
		Association with Firm	Expert with more than 5 years	carrying out Eligible 1 project 3.50 assignment
			experience with the firm or an	iv ASSOCIATE 10.00 10% CONSULTANT
			Infin of an Independen t Expert with more than 10 years experience Permanent 2.125 Expert with 2-5 years experience with the firm or an	i. Length / status of Association with Firm i. Length / status of Association with Firm i. Length / status of Association with Expert with Expert with more than 10 years experience with the firm or an Independent Expert with more than 15 years experience

New Mangalore Port Authority Page 15 of 22

S.No.	Clause No.	Oriç	ginal Clause	Modified Clause
			Independen t Expert with more than 5 years experience Permanent 1.75 Expert with less than 2 years experience with the	Permanent 1.75 Expert with more than 5 years experience with the firm or an Independent Expert with more than 10 years experience
			firm or an Independen t Expert	ii. Qualification 2.50 Master in 2.50 Engineering or Finance
			with more than 2 years experience	Bachelor in Engineering or Finance
		ii. 2.50 Qualification	0 PhD / 2.50 Master/	iii. Total Work 2.50 more than 10 2.50 Experience yrs
			CA/ACMA(I CMA)	6-10 yrs 2.00 5 yrs 1.75
			/CFA/MBA(Fin) or equivalent	Adequacy in 2.50 More than 3 2.50 Assignment /Job projects
		iii. Total 2.50 Work		iv. Experience in 2- 3 projects 2.00
		Experience	6-10 yrs 2.125 5 yrs 1.75	

New Mangalore Port Authority Page 16 of 22

S.No.	Clause No.	Or	iginal Clause			Modi	fied Clause		
		Adequacy in Assignmen t /Job iv. Experience in carrying out Eligible assignment	0 More than5 Projects 3- 5 projects 2 projects	2.50 2.125 1.75		carrying out Eligible assignment	1 project	1.75	
9.	3, Criteria for Evaluation, 3.1, Evaluation of Technical Proposals, pg. 37 to 40	3.1.3 Scoring Criteria Paragraphs appearing after While awarding marks for the Applicant or Key Persundertaken the highest nobe entitled to the maximuland all other competing Personnel, as the case assigning scores in respense Assignments, all Eligible Personnel shall be considered.	the number of onnel, as the cumber of Eligibum score for the ng Applicants may be, ct of the size a Assignments of	Eligible Assignments some category as a may be, that allowed by the category and respective and quality of Eligans	nts, on as nall ory (ey for ble	The paragraphs appearing after 3 clause shall be deleted.	1.3 Scoring Criter	ia as shov	vn in original

New Mangalore Port Authority Page 17 of 22

S.No.	Clause No.	Original Clause	Modified Clause
10.	Terms of Reference, 3, Scope of Services, Para 3.1, Pg. 50	3.1 The scope of services shall include:1. Scrutiny of various policies /guidelines issued by the Government of India / statutory authorities from time to time and suggests any action to be taken by Port.	 3.1 The scope of services shall include: 1. Scrutiny of the project with reference to PPP / Captive policies issued by the Government; submit Inception Report containing a Conceptual Business Plan including methodology and work plan for structuring the project shall be brought out by the Consultant.
11.	Schedule 1, Terms of Reference, Clause 4, Deliverables, Pg. 54-55 of RFP	C. Appraisal Report (Refer Clause 3.9) In case the viability gap funding projected in the Appraisal Report exceeds the projection in the Inception Report by more than 10% of the capital costs, a deduction of 5% (five per cent) of the Agreement Value shall be made from the payment due to the Consultant. Such deduction shall be deemed to be mutually agreed genuine pre-estimated compensation and damages suffered by the Authority on account of inaccurate projections leading to higher costs.	C. Appraisal Report (Refer Clause 3.9) In the event that a viable project does not seem possible, the Consultant shall not proceed with the Consultancy and the same shall be terminated. Upon such termination, the Consultant shall be entitled to (a) full payment for the Deliverables completed by it; and (b) part payment for the non-completed deliverables on pro-rata basis as assessed by the Authority and the Authority's decision in this regard shall be final.
		In the event that a viable project does not seem possible, the Consultant shall not proceed with the Consultancy and the same shall be terminated. Upon such termination, the Consultant shall be entitled to (a) full payment for the Deliverables completed by it; and (b) part payment for the non-completed deliverables on pro-rata basis as assessed	

New Mangalore Port Authority Page 18 of 22

S.No.	Clause No.		Original Clause		Modified Clause
		by the Authori	ty and the Authority's decision in this regard		
12	Schedule 1, Terms of Reference, Clause 7, Consultancy Team, Key Personnel: Page No.		Team Leader (the "PPP Expert")	Key Personnel: PPP Expert-cum-Team Lead	ler (the "PPP Expert")
	59 – 60 of RFP	Educational Qualifications Essential Experience	Chartered Accountant / MBA (Finance) /ICWA / CFA or equivalent 15 years. He should have led the financial team in	Educational Qualifications	Chartered Accountant /ICWA / CFA / MBA (Finance) with Bachelors or Masters in Engineering
			Preparation of revenue Model, Bid Documents for not less than 5 (five) Public Private Partnership (PPP) projects, out of which, at least 1 (one) should be a Port Sector PPP project.	Essential Experience	15 years. He should have led the financial team in preparation of revenue model, bid documents for atleast 3 (three) Public Private Partnership (PPP) projects, out of which, 1 (one) project should be a Port Sector PPP project.
		lob	He will load as ordinate and supervise the	Job Responsibilities	He will lead, co-ordinate and supervise the
		Job Responsibilities	He will lead, co-ordinate and supervise the Consultancy Team for delivering the Consultancy in a timely manner as envisaged in this RFP. He shall be responsible for preparation of Revenue Model and appropriate structure for the Project, prepare bid documents for selection of the Concessionaire in line with		Consultancy Team for delivering the Consultancy in a timely manner as envisaged in this RFP. He shall be responsible for preparation of Revenue Model and appropriate structure for the Project, prepare bid documents for selection of the Concessionaire in line with the principles of PPP / Captive

New Mangalore Port Authority Page 19 of 22

S.No.	Clause No.		Original Clause	Mo	odified Clause
			the principles of PPP / Captive Policy guidelines. He shall not delegate his responsibilities except with the prior written approval of the Authority	ар	sponsibilities except with the prior written oproval of the Authority s and when required
			Fifteen working days in a month till completion of the services.		
		Port Sector Expert Educational Qualifications	Master/ Bachelor in Engineering or equivalent	Port Sector Expert Educational Qualifications Essential Experience	Masters or Bachelors degree in Engineering or Equivalent 10 years.
		Essential Experience	10 years He should have worked as a sector expert in not less than 3 (three) Port Sector projects preferably Bulk Liquid handling facilities.	Job Responsibilities	He should have worked as a sector expert for 2 (two) Port Sector projects, out of which 1 project should be Bulk Liquid handling facilities. He will provide the sectoral perspective and shall be responsible for preparation of Feasibility Report for the project
		Job Responsibilities	He will provide the sectoral perspective and shall be responsible for preparation of Feasibility Report for the project including cost estimates and traffic projections and the Draft Concession Agreement. He shall assist the Team		including cost estimates and traffic projections and the Draft Concession Agreement. He shall assist the Team Leader in the preparation of the Revenue Model and preparation of bid documents for selection of the

New Mangalore Port Authority Page 20 of 22

S.No.	Clause No.		Original Clause	Mod	lified Clause
		Minimum time required on site	Leader in the preparation of the Revenue Model and preparation of bid documents for selection of the Concessionaire. Seven working days in a month till completion of the services.	Minimum time required on site	Concessionaire. As and when required.
		Legal Consultant Educational		Legal Consultant Educational Qualifications	Masters / Bachelors degree in
		Qualifications Essential Experience Job Responsibilities	Master / Bachelor in Law or equivalent 5 years He should have worked as a Legal Consultant for issues relating to commercial contracts in not less than 3 (three) Port Sector projects. He will provide the Legal Perspective	Essential Experience	Law or Equivalent 5 years He should have worked as a Legal Consultant for issues relating to commercial contracts in a Transaction Advisory team for atleast 1 (one) Port
		Minimum time	and will review the Model Bid Documents including Model Concession Agreement to affect project specific changes required from legal perspective. Seven working days in a month till	Job Responsibilities	Sector project. He will provide the Legal Perspective and will review the Model Bid Documents including Model Concession Agreement to affect project specific changes required from logal
		required on Site	completion of the services.	Minimum time required on site	changes required from legal perspective. As and when required.
				withintiant time required on site	As and when required.

New Mangalore Port Authority Page 21 of 22

S.No.	Clause No.		Original Clause	Modified Clause		
		Associate Consultant Educational Qualifications Essential Experience	Master / Bachelor degree in Engineering or Finance or equivalent	Associate Consultant Educational Qualifications Essential Experience Job Responsibilities	Masters / Bachelors degree in Engineering or Finance 5 years He should have worked as member of a Transaction Advisory team in preparation of Revenue Model and/or Bid Documents for at least 1 (one) Port Sector projects. He will assist the Team Leader / PPP Expert and Port Sector Expert by providing inputs for preparation of Revenue Model, project structure and preparation of bid documents.	
		Job Responsibilities				

New Mangalore Port Authority Page 22 of 22

NEW MANGALORE PORT AUTHORITY

Civil Engineering Department



TENDER NO. CIVIL/CE(C)/EE(C)/05/2022-23

eProcureTender ID: 2022_NMPT_686079_1

REQUEST FOR PROPOSAL (RFP)

FOR

Appointment of Transaction Advisor for Concessioning of Berth No. 13 for handling Liquid Bulk / LPG / POL products / Chemicals on Captive / PPP Mode

COMPLETED TENDERS ARE TO BE SUBMITTED ONLINE ON OR BEFORE 15:30 HRS ON 30.05.2022 TECHNICAL BIDS WILL BE OPENED AT 15:30 HRS ON 31.05.2022

CHIEF ENGINEER (CIVIL)i/c

New Mangalore Port Authority,

Panambur, Mangalore-575010

Phone/Fax: 0824-2407948 .Website: www.newmangaloreport.gov.in

APRIL 2022

For any further clarifications / queries on e-Tendering, Chief Engineer (Civil)i/c, NMPA, can be contacted at: Phone No. 0824-2407493 or; e-mail:chiefengineer@nmpt.gov.in

DISCLAIMER

The information contained in this Request for Proposal document ("RFP") or subsequently provided to Applicants, whether verbally or in documentary or any other form by or on behalf of the Authority or any of its employees or advisers, is provided to Applicants on the terms and conditions set out in this RFP and such other terms and conditions subject to which such information is provided.

This RFP is not an agreement or an offer by the Authority to the prospective Applicants or any other person. The purpose of this RFP is to provide interested parties with information that may be useful to them in the formulation of their Proposals pursuant to this RFP. This RFP includes statements, which reflect various assumptions and assessments arrived at by the Authority in relation to the Consultancy. Such assumptions, assessments and statements do not purport to contain all the information that each Applicant may require. This RFP may not be appropriate for all persons, and it is not possible for the Authority, its employees or advisers to consider the objectives, technical expertise and particular needs of each party who reads or uses this RFP. The assumptions, assessments, statements and information contained in this RFP, may not be complete, accurate, adequate or correct. Each Applicant should, therefore, conduct its own investigations and analysis and should check the accuracy, adequacy, correctness, reliability and completeness of the assumptions, assessments and information contained in this RFP and obtain independent advice from appropriate sources.

Information provided in this RFP to the Applicants may be on a wide range of matters, some of which may depend upon interpretation of law. The information given is not intended to be an exhaustive account of statutory requirements and should not be regarded as a complete or authoritative statement of law. The Authority accepts no responsibility for the accuracy or otherwise for any interpretation or opinion on the law expressed herein.

The Authority, its employees and advisers make no representation or warranty and shall have no liability to any person including any Applicant under any law, statute, rules or regulations or tort, principles of restitution or unjust enrichment or otherwise for any loss, damages, cost or expense which may arise from or be incurred or suffered on account of anything contained in this RFP or otherwise, including the accuracy, adequacy, correctness, reliability or completeness of the RFP and any assessment, assumption, statement or information contained therein or deemed to form part of this RFP or arising in any way in this Selection Process.

The Authority also accepts no liability of any nature whether resulting from negligence or otherwise however caused arising from reliance of any Applicant upon the statements contained in this RFP.

The Authority may in its absolute discretion, but without being under any obligation to do so, update, amend or supplement the information, assessment or assumption contained in this RFP.

The issue of this RFP does not imply that the Authority is bound to select an Applicant or to appoint the Selected Applicant, as the case may be, for the Consultancy and the Authority reserves the right to reject all or any of the Proposals without assigning any reasons whatsoever.

The Applicant shall bear all its costs associated with or relating to the preparation and submission of its Proposal including but not limited to preparation, copying, postage, delivery fees, expenses associated with any demonstrations or presentations which may be required by the Authority or any other costs incurred in connection with or relating to its Proposal. All such costs and expenses will remain with the Applicant and the Authority shall not be liable in any manner whatsoever for the same or for any other costs or other expenses incurred by an Applicant in preparation or submission of the Proposal, regardless of the conduct or outcome of the Selection Process.

GLOSSARY

Agreement As defined in Schedule -2

Agreement Value As defined in Clause 6.1.2 of Schedule -2

Applicable Laws

As defined in Schedule – 2

Applicant

As defined in Clause 2.1.1

Associate

As defined in Clause 2.3.3

Authorised Representative

As defined in Clause 2.14.3

Authority

As defined in Clause 1.1

Bid Documents

As defined in Clause 1.2

Conditions of Eligibility As defined in Clause 2.2

Conflict of Interest As defined in Clause 2.3

Consultancy As defined in Clause 1.2

Consultancy Team As defined in Clause 2.1.4

Consultant As defined in Clause 1.2

CV Curriculum Vitae

DBFOT Design, Build, Finance, Operate and Transfer

Deliverables As defined in Clause 4 of Schedule-1

Documents As defined in Clause 2.10

Effective Date As defined in Clause 2.1 of Schedule-2

Eligible Assignments As defined in Clause 3.1.4

Expatriate Personnel As defined in Clause 1.1.1(i) of Schedule-2

PPP Expert As specified in Clause 2.1.4

Financial Proposal As defined in Clause 2.16.1

Form of Agreement Form of Agreement as in Schedule -2

INR, Re, Rs. Indian Rupee(s)

Key Personnel As defined in Clause 2.1.4

LOA Letter of Award

Lead Member As defined in Clause 2.1.1

Member As defined in Clause 2.3.3 (a)

Official Website As defined in Clause 1.11.3

PPP Public Private Partnership

Personnel As defined in Clause 1.1.1(n) of Schedule-2

Project As defined in Clause 1.1.1

Team Leader/ Project Manager As defined in Clause 4.6 of Schedule-2

Professional Personnel As defined in Clause 2.15.6
Prohibited Practices As defined in Clause 4.1
Proposals As defined in Clause 1.2
Proposal Due Date or PDD As defined in Clause 1.5

RFP As defined in Disclaimer

Resident Personnel As defined in Clause 1.1.1(o) of Schedule-2

Scheduled Bank As specified in RBI Act, 1934

Selected Applicant As defined in Clause 1.6
Selection Process As defined in Clause 1.6

Services As defined in Clause 1.1.1(q) of Schedule -2

Sole Firm As defined in Clause 2.1.1

Statutory Auditor An Auditor appointed under Applicable Laws

Sub-Consultant As defined in Clause 1.1.1(r) of Schedule-2

Support Personnel As defined in Clause 2.15.6

Technical Proposal As defined in Clause 2.15.1

TOR As defined in Clause 1.2

US\$ United States Dollar

The words and expressions beginning with capital letters and defined in this document shall, unless repugnant to the context, have the meaning ascribed thereto herein.

Contents	Page No.
Section: I Notice Inviting E-Tender	1-2
Section: II Instructions to Applicants for Online Bid submission	3-5
Section: III	
1. Introduction	6-16
2. Instruction to Applicants	17-36
Criteria for Evaluation	37-42
Fraud and corrupt practices	43-44
5. Pre-Proposal Conference	45
6. Miscellaneous	46
Schedule - I	
Terms of Reference	47-62
Schedule - II	
Form of Agreement	63-93
Schedule - III	
Guidance Note on Conflict of Interest	94-95
Annexure-1 : Bank Guarantee for Performance Security	96-98
Annexure-2 : Proposed Deployment Schedule	99
Appendices	
Appendix-I: Technical Proposal	
Form 1: Letter of Proposal	101-104
Form 2: Particulars of the Applicant	105-106
Form 3: Statement of Legal Capacity	107
Form 4: Financial Capacity of Applicant	108
Form 5: Power of Attorney	109-110
Form 6: Particulars of Key Personnel	111
Form 7: Abstract of Eligible Assignments of the Applicant	112-113
Form 7A: Abstract of other relevant experience of the Applicant	114-115
Form 8: Abstract of Eligible Assignments of Key Personnel	116
Form 8A: Abstract of other relevant experience of the Key Personnel	117
Form 9: Eligible Assignments of Applicants	118
Form 9A: Other relevant assignments of the Applicant	119
Form 10: Eligible Assignments of Key Personnel	120
Form 10A: Other relevant assignments of the Key Personnel	121
Form 11: CV of Professional Personnel	122
Form 12: Integrity Pact	123-129
Appendix – II: Financial Proposal	
Form 1: Covering Letter	130
Form 2: Financial Proposal	131-132
Project Implementation Schedule	133
Layout of Berth No. 13	134

New Mangalore Port Authority SECTION: I

NOTICE INVITING E-TENDER

NIT NO. CIVIL/CE(C)/EE(C)/05/2022-23, dtd. 29.04.2022 [eProcureTender ID: 2022_NMPT_686079_1]

The Authority invites Proposals (the "Proposals") through E-Procurement System https://eprocure.gov.in/cppp/ of CPP from interested, experienced and reputed consultancy firm(s)/consortia to submit their offers containing Technical and Financial Bids (i.e. Technical Proposal and Financial Proposal) for the work of Appointment of Transaction Advisor for Concessioning of Berth No. 13 for handling Liquid Bulk / LPG / POL products / Chemicals on Captive / PPP

Mode at New Mangalore Port. Appointment of Transaction Advisor Name of the Work Concessioning of Berth No. 13 for handling Liquid Bulk / LPG / POL products / Chemicals on Captive / PPP Mode at New Mangalore Port. Rs. 59,000.00 (Rupees Fifty Nine Thousand only) Bid Security / or exemption as per clause No.22 of Section II. **Earnest Money Deposit** Cost of Tender One Rs.1,120.00 (Rupees Thousand Hundred Twenty only) or exemption as per clause No:22 of Section II. Date of tender document available 29/04/2022 at 16.00 Hrs. to parties to download Last date for receiving Pre 16.05.2022 at 15:30 HRS bid queries/clarifications Date: 16/05/2022: Time: 15:30 Hrs Pre-Proposal Conference Mode: Virtual Meeting (Microsoft Teams) Contact details for Pre-bid Conference: Chief Engineer (Civil)i/c, New Mangalore Port Authority, Panambur, Mangalore - 575 010 Phone: 0824 2407948 / 0824 2407413 Email: chiefengineer@nmpt.gov.in Starting date of e-Tender for 23.05.2022 at 11:00 HRS submission of on line Techno-Commercial Bid and price Bid Proposal Due Date (PDD) 30.05.2022 at 15:30 HRS Closing date of online etender for submission of Techno-Commercial Bid Price Bid

RFP for Appointment of Transaction Advisor for Concessioning of Berth13 for handling Liquid Bulk on Captive / PPP Mode

Part-I : Date of Opening of	31.05.2022 at 15:30 HRS		
Technical Proposal			
Part-II -Financial Proposal:	Shall be communicated separately.		
Validity of Tender	180 days from the date of tender opening.		
Commencement of providing	30 Days from the date of issue of Letter of		
Transaction advisory service	Acceptance (LOA)		

Amendments / further information etc. pertaining to the tender if any will be issued only through www.newmangaloreport.gov.in and CPP Portal www.eprocure.gov.in/eprocure/app. Prospective bidders are instructed to view these websites from time to time for any such notification.

Tender documents may be viewed or downloaded by interested and eligible Tenderers from the Central Public Procurement Portal (CPPP) of Government of India, website: www.eprocure.gov.in/eprocure/app. on the above mentioned dates after payment of Tender fee of Rs.1,120/-.

Note:

Micro and Small Enterprises (MSE) registered with District Industries Centre (DIC) or Khadi and Village Industries commission or Khadi & Industries Board (KVIB) or Coir Board or National Small Industries Corporation (NSIC) or Directorate of Handicrafts and Handlooms or Udyog Aadhar memorandum or any other body specified by Ministry of MSME shall be exempted of EMD / Bid Security and tender fee on producing self attested supporting certificates along with Technical Bid.

NMPA Bank Details

Name of the Payee: The FA&CAO, NMPA, Panambur, Mangalore

Name of the Bank: State Bank of India, Panambur,

Mangalore - 575 010.

Bank A/C No. 10205649448 IFSC Code: SBIN0002249 MICR Code: 575002011

Sd/-

Chief Engineer (Civil)i/c New Mangalore Port Authority Panambur, Mangalore

SECTION: II

INSTRUCTIONS TO THE BIDDERS FOR E-SUBMISSION OF THE BIDS ONLINE THROUGH E-PROCUREMENT PORTAL

- 1. Bidder should do Online Enrolment in the Portal using the option Click Here to Enroll available in the Home Page. Then the Digital Signature enrollment has to be done with the e-token, after logging into the portal.
- 2. Bidder then logs into the portal giving user id / password chosen during enrollment.
- 3. The e-token that is registered should be used by the bidder and should not be misused by others.
- 4. DSC once mapped to an account cannot be remapped to any other account. It can only be inactivated.
- 5. The Bidders can update well in advance, the documents such as certificates, purchase order details etc., under My Documents option and these can be selected as per tender requirements and then attached along with bid documents during bid submission. This will ensure lesser upload of bid documents.
- 6. After downloading / getting the tender schedules, the Bidder should go through them carefully and then submit the documents as per the tender document; otherwise, the bid will be rejected.
- 7. The BOQ template must not be modified/ replaced by the bidder and the same should be uploaded after filling the relevant columns, else the bidder is liable to be rejected for that tender. Bidders are allowed to enter the Bidder Name and Values only.
- 8. If there are any clarifications, this may be obtained online through the e-Procurement Portal, or through the contact details given in the tender document. Bidder should take into account of the corrigendum published before submitting the online bids on the portal or http://eprocure.gov.in/eprocure/app www.newmangalore-port.com or Bidder, in advance, should prepare the bid documents to be submitted as indicated in the tender schedule and they should be in PDF formats.
- 9. Bidder should arrange for the EMD / Bid Security and tender fee as specified in the tender. The original should be posted/couriered/given in person to the Tender Inviting Authority, within the bid submission date and time for the tender.
- 10. The bidder should read the terms and conditions and accepts the same to proceed further to submit the bids.

- 11. The bidder has to submit the tender document(s) online well in advance before the prescribed time to avoid any delay or problem during the bid submission process.
- 12. There is no limit on the size of the file uploaded at the server end. However, the upload is decided on the Memory available at the Client System as well as the Network bandwidth available at the client side at that point of time. In order to reduce the file size, bidders are suggested to scan the documents in 75-100 DPI so that the clarity is maintained and the size of file gets reduced. This will help in quick uploading even at very low bandwidth speeds.
- 13. It is important to note that, the bidder has to click on the Freeze Bid Button, to ensure that, he/she completes the Bid Submission Process. Bids, which are not frozen, are considered as Incomplete/Invalid bids and are not considered for evaluation purposes.
- 14. The Tender Inviting Authority (TIA) will not be held responsible for any sort of delay or the difficulties faced during the submission of bids online by the bidders due to local issues.
- 15. The bidder may submit the bid documents online mode only, through this portal. Offline documents will not be handled through this system. Only hard copy of the Technical Bid and soft copy in CD(word format) should reach immediately after PDD to the Chief Engineer (Civil), NMPA
- 16. At the time of freezing the bid, the e-Procurement system will give a successful bid updating message after uploading all the bid documents submitted and then a bid summary will be shown with the bid no., date & time of submission of the bid with all other relevant details. The documents submitted by the bidders will be digitally signed using the e-token of the bidder and then submitted.
- 17. After the bid submission, the bid summary has to be printed and kept as an acknowledgement as a token of the submission of the bid. The bid summary will act as a proof of bid submission for a tender floated and will also act as an entry point to participate in the bid opening event.
- 18. Successful bid submission from the system means, the bids as uploaded by the bidder is received and stored in the system. System does not certify for its correctness.
- 19. The bidder should see that the bid documents submitted should be free from virus and if the documents could not be opened, due to virus, during tender opening, the bid is liable to be rejected.
- 20. The time that is displayed from the server clock at the top of the tender Portal, will be valid for all actions of requesting bid submission, bid opening

- etc., in the e-Procurement portal. The Time followed in this portal is as per Indian Standard Time (IST) which is GMT+5:30. The bidders should adhere to this time during bid submission.
- 21. The bidders are requested to submit the bids through online e-Procurement system to the Tender Inviting Authority (TIA) well before the bid submission end date and time (as per Server System Clock).
- 22. Tender form Fee and Bid Security shall be submitted with the Part I-Technical BID. BID submitted without fees, as mentioned above will not be considered for evaluation and shall be rejected summarily. Micro and Small Enterprises (MSE) registered with District Industries Centre (DIC) or Khadi and Village Industries commission or Khadi & Industries Board (KVIB) or Coir Board or National Small Industries Corporation (NSIC) or Directorate of Handicrafts and Handlooms or Udyog Aadhar memorandum or any other body specified by Ministry of MSME shall be exempted of EMD / Bid Security and tender fee on producing self attested supporting certificates along with Technical Bid.
- 23. The bidder/Tenderer/contractor shall file the applicable returns with Tax departments in time and submit the same as documentary proof.
- 24. The GST applicable shall be shown as separate line items in the Tax invoices to avail input credit to Port.
- 25. In the event of forfeiting the LD/SD, GST is applicable and while imposing penalty, applicable GST shall be collected.

SECTION: III

1. INTRODUCTION

1.1 Background

- 1.1.1 The New Mangalore Port Authority (the "Authority") intends to hand over existing Berth No. 13 for handling POL products / Chemicals as per PPP / Captive Policy Guidelines (the "Project").
- 1.1.2 With a view to invite bids for the Project, the Authority requires preparation of a feasibility report for project appraisal, developing a revenue model based on the feasibility and has decided to conduct a financial viability of the Project. If found technically feasible and financially viable, the Project may be awarded on Captive User / PPP basis to a private entity (the "Concessionaire") selected through a competitive bidding process. The Project would be implemented in accordance with the terms and conditions stated in the concession agreement to be entered into between the Authority and the Concessionaire (the "Concession Agreement").
- 1.1.3 In pursuance of the above, the Authority has decided to carry out the process for selection of a Consultant cum Transaction Adviser (the "Consultant"). The Transaction Adviser shall, in accordance with the Terms of Reference specified at Schedule-1 (the "TOR"), prepare a Feasibility Report, appraise the Project, develop a revenue model and project structure, prepare the Bid Documents (RFQ, RFP including Draft Concession Agreement) based on Model Bid Documents including Model Concession Agreement to be provided by Authority, and assist the Authority in the bidding process, execution of Concession Agreement with selected Concessionaire and achievement of Financial close.

2 Request for Proposal

The Authority invites tenders from reputed Consulting firms (the "Proposals") for selection of Transaction Adviser (the "Consultant") who shall conduct a feasibility study for handing over of Berth No. 13 in line with the principles of PPP / Captive Policy guidelines, develop a revenue model and appropriate structure for the Project (the "Revenue Model"), prepare bid documents for selection of the Concessionaire (the "Bid Documents"), assist the Authority in the bidding process, execution of Concession Agreement with selected Concessionaire and achievement of Financial close (collectively the "Consultancy").

2.1 Brief of New Mangalore Port.

2.1.1 The Major Port of New Mangalore (NMP), a lagoon type harbour located on the west coast of India (12⁰55'N; 74⁰48';E) is a gateway port for State of

Karnataka. Amongst the total traffic handled at the port, the bulk liquid crude and POL products forms a major part (about 2/3rd). The Liquid Cargo handling capacity at the port currently stands at 21 MTPA of crude and POL.

2.1.2 The Port, a lagoon type harbor is entered through a dredged channel, about 7.5 km long. The entrance is protected by two breakwaters extending to a length of 770m on either side of the channel. The dredged channel has a bottom width of 245m, with a dredged depth of 15.40m below CD. The harbor basin and the Oil Dock Arm have dredged depths of 15.1 m below CD, suitable for handling 65/ 85,000 DWT crude tankers.

2.1.3 Marine & Land Environment:

The Port is located on the alluvial plain, about 10km north of the mouth of the Gurupur and the Netravathi rivers. The old port of Mangalore, located on the confluence of these two rivers, is a roadstead port. This port is operational only during the fair weather season viz., 15th September to 15th May.

2.1.4 Climate

The climate of Mangalore has seasonal variation on account of the alternating SW and NE monsoon. The different seasons are characterized as under:

- The cool season (December to March), during which winds blow from NE and the weather is dry and little cold.
- The hot season (April to May). During this season, winds are light and variable with sea breezes. Tropical cyclonic storms (cyclones) may cross the Arabian Sea during this season.
- During SW monsoon or rainy season (June to September), the wind over the sea is between SW and W, but mainly W to NW along the coast.
- The interim period (October and November) is marked by light winds with land and sea breezes. Occasional tropical cyclones occur in the Arabian Sea in this period.

2.1.5 Temperature

In Mangalore, the air temperature is high throughout the year. The mean temperature in the hottest month, which generally occurs just before the onset of the SW monsoon, varies between 33° C to 37° C and lowest temperature recorded in a year has been 16.7° C.

2.1.6 Humidity

Humidity is fairly high here at all times and seasons. There is a large variation as well from damp conditions in early mornings in the wet season to the drier conditions in the afternoon in the dry season. Daily variation is least during the SW monsoon. The maximum relative humidity observed during monsoon period is 92% (August). The minimum relative humidity during non monsoon period is 60% (December and January).

2.1.7 Visibility

Sometimes mist develops on the coast after a calm and clear night, but it soon disperses after sunrise. In Mangalore, during the SW monsoon (from June to September) thick haze develops. The maximum number of foggy days in a year is only 3.

2.1.8 Wind

The winds in the monsoon months (June, July and August) are predominantly from SW to W, with a maximum intensity of 20 to 61 kmph. The winds in the remaining months of the year are predominantly from NW with a maximum intensity of 20 to 61 kmph.

2.1.9 Rainfall

The average annual rainfall is 3467mm. The rainfall is concentrated in te SW monsoon (June, July, August and September), during which period the precipitation is as much as 84% of the total annual rainfall. The rainfall is maximum in July (1102.7mm); February is the most dry month (with rainfall being about 2 mm).

The maximum rainfall recorded in a day (24 hour period) was 27 cm. The average number of rainy days in a year is 123.

2.1.10 Cyclones

While the average frequency of cyclonic storms in the Arabian Sea is about one per year, there have been years when two or three such storms have occurred. There have also been years without any such storms. The maximum wind speed so far recorded in a cyclonic storm generally does not exceed 62 kmph (16.9m/s.), except once during 1965 when the maximum speed recorded was 97 kmph (26.9m/s.).

2.1.11 Wave Climate

The predominant direction of waves in the vicinity of New Mangalore Port during monsoon months (June, July and August) is West to South West, whereas the predominant direction during the fair months is North – West and North. High waves are experienced only during the monsoon month on site.

2.1.12 Tides

An automatic recording tide gauge has been installed by the Survey of India, National Hydrographic Office, Dehradun in the vicinity of the Oil Jetty area of the Port, inside the lagoon.

2.1.13 Current

The current along the coast during the SW monsoon (from February to September) is generally towards the South (from 160° to 200°) with a strength of 0.22 to 0.80 knots. During the NE monsoon (from November to January), the current is generally towards the North (from 0° to 40° and 320° to 360° bearing) with a velocity of 0.22 to 0.60 knots.

In the port entrance channel protected by breakwater, the current direction lags 6° to 8° behind the coastal current. The current in the lagoon area further lags behind the approach channel on a average by 10° to 15°. The magnitude of the current outside the lagoon area during the monsoon as experienced by pilots is about 1 to 1.5 knots.

2.1.14 Littoral Drift

The study carried out by CWPRS, Pune during 2002 has concluded as under:-

- ➤ Significant littoral drift occurs between 1 and 2m depth contours, which are at a distance of about 200 to 400m from shoreline.
- Seasonal drift distribution has indicated that during North-East monsoon, littoral drift is towards north, whereas during South West monsoon and non-monsoon period, the drift is towards South.
- The northwards drift is comparatively less than the southward drift.
- ➤ The average littoral drift in the region is of the order of 0.58 lakh cubic meter towards south during southwest monsoon and non-monsoon period and 0.08 lakhs cubic meter towards north in North East monsoon. The average net littoral drift is 0.5 lakh cubic meter per year towards South.
- ➤ The mathematical model studies for shoreline stability have indicated that the effect of breakwater is felt upto a distance of about 2.5 km on either side of the breakwaters. The examination of remote sensing imageries confirms these observations.

2.1.15 Soil Characteristics

The details soil borings carried out in the Oil Dock Arm (near Berth No.13)

indicate that hard rock is at about -23.5 m to -25 m. Top layers consisted of coarse sand / fine sand.

2.2 Existing Port Facilities:

NMP is spread over 820 Ha of land area and water area of about 120 Ha.

2.2.1 Berths and Jetties

The port has total of 16 berths. In addition, MRPL has constructed one SPM within the Port's limits at -22 m contour at a distance of about 17 Km from the shoreline and laid underwater crude oil pipelines. The details of Berths are presented in the following table.

Berth Particulars

Name of Berth	Type of Berth	Draught (In Mtrs)	Quay Length (Mtrs.)	DWT (In MT)
B No.1	General Cargo	7.0	125	4000
B No.2	General Cargo	10.50	198	30000
B No.3	General Cargo	10.30	198	30000
B No.4	General Cargo/ Liquid Ammonia / Phosphoric Acid	9.50	198	30000
B No.5	General Cargo / Bulk Cement/ Edible Oil	9.50	198	30000
B No.6	General Cargo/ Bulk cargo	9.50	198	30000
B No.7	General Cargo/ Bulk cargo	9.50	198	30000
B No.14	Coal/ Bulk cargo Mechanized for handling containers on PPP mode (JSW MCTPL)	14.00	350	90000
B No.15	Coal (UPCL)	14 .00	350	90000
B No.16	Coal (M/s. CMCTPL)	14 .00	350	90000
B No. 9	LPG	10.50	330	45000
B No.10	Crude / POL product	14.00	320	85000
B No.11	Crude / POL product	14.00	320	85000
B No.12	POL product/ LPG/ Chemicals	12.50	320	50000
B No.13	POL product	14.00	350	85000
B No. 8	Coal/Iron Ore (KIOCL)	12.50	300	60000
SPM	Crude oil	21.00	-	300000

2.2.2 The Total Traffic handled by the Port during past 7 (seven) years is presented in the following table:-

SI. No.	Year	Total Traffic handled in NMPA (in MTPA)
1	2014-15	36.57
2	2015-16	35.58
3	2016-17	39.95
4	2017-18	42.05
5	2018-19	42.51
6	2019-20	39.14
7	2020-21	36.50

2.2.3 The Total Traffic handled by the Port vis-a-vis traffic handled at Oil Berths (Berth No. 9 to 13) and traffic handled at Berth No. 13 is presented in the following Table:-

SI.	Year	Total Traffic	Traffic handled at Oil	Traffic handled at
No.		handled in	Berths (B.9 to B.13)	Berth No. 13
		NMPA	(in MTPA)	(in MTPA)
		(in MTPA)	, ,	,
1	2014-15	36.57	13.73	0.33
2	2015-16	35.58	13.93	1.34
3	2016-17	39.95	16.09	1.93
4	2017-18	42.05	16.98	2.07
5	2018-19	42.51	16.84	2.34
6	2019-20	39.14	15.53	2.27
7	2020-21	36.50	13.71	2.28

2.2.4 The commodity-wise traffic handled at Berth No. 13 is as follows:-

COMMODITY NAME	2014-15	2015-16	2016-17	2017-18	2018-19	2019-20	2020-21
BENZENE	55,614	1,67,487	1,87,860	1,93,722	2,25,413	1,55,563	1,25,043
LPG (HPCL)		2,01,406	6,78,597	7,30,784	9,30,516	11,55,185	13,29,043
LPG (TOTAL GAZ)		3,57,456	2,33,902	2,00,772	1,09,159	2,47,584	1,59,976
PARAXYLENE	2,64,441	5,84,858	6,76,026	6,91,683	7,85,745	6,43,498	4,27,510
PARAFFINIC RAFFINATE						22,349	86,002
REFORMATE							1,52,876
POL		31,771	1,51,666	2,46,402	2,91,381	41,171	
POL MRPL				4,649			

RFP for Appointment of Transaction Advisor for Concessioning of Berth13 for handling Liquid Bulk on Captive / PPP Mode

TOTAL	3,34,520	13,43,938	19,28,051	20,68,012	23,42,214	22,65,350	22,80,450
STY. MONOMER		961					
POL OMPL	14,465						

2.2.5 The salient features of berth No. 13 are as follows:-

- Design Ship 1,50,000DWT
- Length 350 Meters
- Open Piled Structures
- Approach Trestle (102 x 16.4 Meters)
- Service Platform (50 x 20 Meters)
- Breasting Dolphin 4Nos (13.96 x 13.4 Meters)
- Mooring Dolphin 6Nos (16 x 12.5 Meters)
- Walkways connecting BDs & MDs, 2.30 Meters wide
- Cell Fenders 1700H on BDs 4 Nos.
- Cell Fenders 800H on flexible dolphins 2 Nos.
- Cast Steel Bollards 5 Tonne capacity 6 Nos.
- Quick Release Mooring Hooks
 - o 300 Tonne capacity 6 Nos.
 - 150 Tonne capacity 4 Nos.
- Gas Detectors 4 Nos.
- Flame Detectors 10 Nos.
- Tower Monitors & Ground Monitors for Fire Fighting with fire water pipelines.
- Operation Both Discharge and Loading operation

2.2.6 Existing Fire fighting system at Berth No 13 consists of the following

SI. No.	Description	Equipments available		
1	Fire Water System	 Two main pumps of 15000 LPM each Two booster pumps (One stand by) to boost the monitor line to a pressure of 12 kg /cm2 15 double headed hydrants and 9 single headed are provided along with hose boxes covering jetty, control tower and 		

KFFI	Appointment of Transaction Advisor	r for Concessioning of Berth13 for handling Liquid Bulk on Captive / PPP Mode
		 booster pump house Two Foam/water monitors of 6000 LPM capacity (90/100mtrs throw) Two Foam/water monitors of 3000 LPM capacity (55/60 mtrs throw) 37 jumbo curtain nozzles (each of 362 LPM capacity) Two fixed water monitors of 2700 LPM each
2	Foam System	 Foam tank of 35000 ltrs capacity for storing foam solution One diesel engine foam pump and one electric driven foam pump for feeding the monitors
3	DCP Protection System	 10 kg DCP – 12 nos. 75 Kg. wheel mounted DCP - 4 Nos. 6.8 kg CO2 extinguisher on wheel – 4 Nos. 5 kg DCP – 4 Nos.
4	Portable & Mobile Fire Fighting Equipment	 75 Kg. wheel mounted DCP - 4 Nos. 6.8 kg CO2 extinguisher on wheel – 4 Nos.
5	Fire/smoke/Gas Detection and Alarm System	 Gas Detectors (4 no's) and flame detectors (10 no's) are provided in the service platform for detection of gas vapours and flames. On sensing signals are sent to the respective panels at the control room.
6	Fire Alarm/Communication System	 Flame proof MCPs (10 no's) are provided on the jetty area. On activation sirens placed in the booster pump house and hooter in the control building will be activated indicating to evacuate the jetty. At the same time hooter available in the pump house II will alert the operators. Flame proof speakers (10 no's) are installed in the jetty. With a Public Address system necessary instructions can be given. Talk back system (11 no's) is installed at control room and jetty for 2 way communications.

2.2.7 Present Proposal

The facilities provided in the Oil Dock Arm and consists of two berths for crude and POL (B.No. 10 & 11) operated by MRPL, two berths for POL products and chemicals (B. No. 12 & 13) and one berth for Liquid Bulk, POL and LPG (B. No. 9).

Considering the immediate demand and future prospects, the Port intends to hand over existing Berth No. 13 to a PPP Operator for handling POL products / Chemicals on PPP mode based on Captive Policy Guidelines.

Berth No. 13 was constructed during the years 2013 at a capital cost of Rs.59.28 Crore (Cost of Berth Rs. 53.58 Cr. & Cost of Dredging Rs. 5.70 Cr.). The commercial operation of Berth No. 13 commenced from 05.09.2014.

The Fire Fighting Facilities of Berth No. 13 was augmented for handling VLGCs by construction of Fire Pump House at a total cost of Rs. 35 Crore and the facility was commissioned on 31.12.2016.

2.2.8 Recent Developments

The PPP Projects for Asset Monetization at Major Ports envisions increase in private participation through PPP mode in the Indian Maritime Sector by strengthening institutional frameworks, model agreements and legislations.

The extract of relevant points of the concept note on PPP projects for Asset Monetization prepared by the Indian Ports Association is as follows:-

In order to improve governance and functioning of Major ports, so that the overall ecosystem for private sector participation in port sector improves, some major steps have been taken recently:

- Major Port Authorities Act 2021: MPA Act 2021, enacted by the
 parliament in February 2021, will provide more autonomy to ports
 and help them take decisions faster with respect to infrastructure
 development and changing market conditions. It is expected to
 usher in a new era for administration of Major Ports in India in which
 the Major Ports will contribute significantly to the economic growth
 and provide world-class port infrastructure by adopting Landlord
 Model of development.
 - The Major Ports shall gain autonomy on many key matters including tariffs, development of port assets, master planning of infrastructure within port limits and powers to make regulations for operations of the Major Ports.
 - ii. Also, the role of Tariff Authority for Major Ports (TAMP) for fixation of the tariff has been done away in the new MPT Act as a tariff regulator and ports would be free to set their tariff based

on market forces. The impact of the changes in Tariff fixation will be enormous as it will create inter port competition and also once the ports are empowered for tariff fixation, they will be in better position to take on markets with near by non major ports handling similar cargo.

- iii. The Act also provides for formulation of an Adjudicatory Board that will adjudicate disputes among Major Ports, Public Private Partnership concessionaires and captive users. MPA Act 2021 will have enormous impact on overall maritime trade as focus would be on meeting the market demand and providing high level of services to the customers, leading to adoption of high-tech in overall port operations.
- Captive Policy: Recently, MoPSW has revised the existing Captive Policy allowing existing captive users to add new berth/ jetty/ terminal/ single point mooring (SPM) for their enhanced capacity requirement; and changes in existing Captive Policy for renewal/ extension of concession period to non-government entities taking into consideration their ongoing business requirements, allotment of waterfront to Government bodies on nomination basis and examining the various aspects like valuation of assets, period of concession, tariff, first right of refusal, etc. The proposal is under review with the Cabinet Committee on Economic Affairs (CCEA) for approval.
- Model Concession Agreement (MCA 2021): A new MCA 2021 has been notified after revisiting the existing Model Concession Agreement 2018, considering the change in policy environment due to enactment of Major Ports Authoroties Act 2021 and exploring provision of further flexibility to respond to dynamic market and regulatory conditions, many provisions have been provided or modified to cater to changing environment and develop a conducive atmosphere around PPP to boost investment.
- Tariff Policy (2021): New Tariff Guidelines, 2021, for framing of Scale of Rates for PPP projects under bidding stage and all future concession agreements to be entered by Major Ports as per MPA Act, 2021 has been notified on 21st Dec.2021. PPP Concesionaires for projects after commencement of the MPA Act are free to fix the tariff based on market conditions and on such other conditions as may be notified by the Government. MoPSW has also issued a Policy to the Major Ports Authorities on 3rd November, 2021 as a policy direction under Section 53 of the MPA Act 2021 for fixation of rates for assets and services offered by the Major Ports. New Tariff Policy for Major Ports empowers the board of the ports and provides

flexibility to fix tariff for assets and services operated by Major Ports themselves as per the market condition.

2. INSTRUCTIONS TO APPLICANTS

A. GENERAL

2.1 Scope of Proposal

- 2.1.1 Detailed description of the objectives, scope of services, Deliverables and other requirements relating to this Consultancy are specified in this RFP. In case an applicant firm possesses the requisite experience and capabilities required for undertaking the Consultancy, it may participate in the Selection Process either individually (the "Sole Firm") or as lead member of a consortium of firms (the "Lead Member") in response to this invitation. The term applicant (the "Applicant") means the Sole Firm or the Lead Member, as the case may be. The manner in which the Proposal is required to be submitted, evaluated and accepted is explained in this RFP.
- 2.1.2 Applicants are advised that the selection of Consultant shall be on the basis of an evaluation by the Authority through the Selection Process specified in this RFP. Applicants shall be deemed to have understood and agreed that no explanation or justification for any aspect of the Selection Process will be given and that the Authority's decisions are without any right of appeal whatsoever.
- 2.1.3 The Applicant shall submit its Proposal in the form and manner specified in this Section of the RFP. The Technical proposal shall be submitted in the form at Appendix-I and the Financial Proposal shall be submitted in the form at Appendix-II. Upon selection, the Applicant shall be required to enter into an agreement with the Authority in the form specified at Schedule-

2.2 Key Personnel

2.2.1 The Consultant's team (the "Consultancy Team") shall consist of the following key personnel (the "Key Personnel") who shall discharge their respective responsibilities as specified below:

Key Personnel	Responsibilities
PPP Expert-cum- Team Leader (the "PPP Expert")	He will lead, co-ordinate and supervise the Consultancy Team for delivering the Consultancy in a timely manner as envisaged in this RFP. He shall be responsible for preparation of Revenue Model and appropriate structure for the Project, prepare bid documents for selection of the Concessionaire in line with the principles of PPP / Captive Policy guidelines. He shall not delegate his responsibilities except with the prior written approval of the Authority
Port Sector Expert	He will provide the sectoral perspective and shall be responsible for preparation of Feasibility Report for the project including cost estimates and traffic projections and the Draft Concession Agreement. He shall assist the Team Leader in the preparation of the Revenue Model and

	preparation of bid documents for selection of the
	Concessionaire.
Legal Consultant	He will provide the Legal Perspective and will
	review the Model Bid Documents including
	Model Concession Agreement to affect project
	specific changes required from legal perspective
Associate Consultant	He will assist the Team Leader / PPP Expert and
	Port Sector Expert by providing inputs for
	preparation of Revenue Model, project structure
	and preparation of bid documents.

2.3 Conditions of Eligibility of Applicants

- 2.3.1 Applicants must read carefully the minimum conditions of eligibility (the "Conditions of Eligibility") provided herein. Proposals of only those Applicants who satisfy the Conditions of Eligibility will be considered for evaluation. To be eligible for evaluation of its Proposal, the Applicant shall fulfill the following:
- 2.3.2 <u>Technical Capacity:</u> The Applicant shall have, over the past 7 (seven) years preceding the PDD, undertaken assignments as specified in Clause 3.1.4. The Consultant shall have to substantiate the above experience with a work order and Completion certificate from the Client of having successfully completed the works indicating the time period during which the work has been executed and actual fees received.
- 2.3.3 Financial Capacity: The Applicant shall have received a minimum of Rs.1.00 crore (US \$ 0.13 million (US Dollar Zero Point One Three million) per annum as professional fees during each of the three financial years preceding the PDD. For the avoidance of doubt, professional fees hereunder refer to fees received by the Applicant for providing advisory or consultancy services to its clients.

2.3.4 Conditions of Eligibility of Key Personnel

Each of the Key Personnel must fulfill the Conditions of Eligibility specified below:

- (A) The Applicant shall offer and make available all Key Personnel meeting the requirements specified in Sub-clause (B) below.
- (B) Each of the Key Personnel must fulfill the Conditions of Eligibility specified below:

Key	Educational	Length of	Experience on Eligible
Personnel	Qualification	Professional	Assignments
		Experience	
PPP Expert	Chartered	15 years	He should have led the financial
	Accountant /		team in Preparation of revenue
	MBA (Finance)		Model, Bid Documents for not
	/ICWA / CFA or		less than 5 (five) Public Private
	equivalent		Partnership (PPP) projects, out

Т		T	C 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
			of which, at least 1 (one) should
			be a Port Sector PPP project.
Port Sector	Master/	10 years	He should have worked as a
Expert	Bachelor in		sector expert in not less than 3
	Engineering or		(three) Port Sector projects
	equivalent		preferably Bulk Liquid handling
			facilities.
Legal	Master /	5 years	He should have worked as a
Consultant	Bachelor in		Legal Consultant for issues
	Law or		relating to commercial contracts
	equivalent		in not less than 3 (three) Port
			Sector projects.
Associate	Master /	5 years	He should have worked as
Consultant	Bachelor		member of a Transaction
	degree in		Advisory team in preparation of
	Engineering or		Revenue Model and/or Bid
	Finance or		Documents for at least 2 (two)
	equivalent		Port Sector projects.

- 2.3.1 The Applicants who satisfies the above criteria shall be considered for further evaluation in accordance with the method of selection specified in Clause 3.1.3 of this document.
- 2.3.2 The Applicant should submit a Power of Attorney as per the format at Form-4 of Appendix-I; provided, however, that such Power of Attorney would not be required if the Application is signed by a partner or Director (on the Board of Directors) of the Applicant, in case the Applicant is a partnership firm or limited liability partnership.
- 2.3.3 Any entity which has been barred by the Central Government, any State Government, a statutory authority or a public sector undertaking, as the case may be, from participating in any project, and the bar subsists as on the date of the Proposal, would not be eligible to submit a Proposal either by itself or through its Associate.
- 2.3.4 An Applicant or its Associate should have, during the last three years, neither failed to perform on any agreement, as evidenced by imposition of a penalty by an arbitral or judicial authority or a judicial pronouncement or arbitration award against the Applicant or its Associate, nor been expelled from any project or agreement nor have had any agreement terminated for breach by such Applicant or its Associate.
- 2.3.5 While submitting a Proposal, the Applicant should attach clearly marked and referenced continuation sheets in the event that the space provided in the specified forms in the Appendices is insufficient. Alternatively,

Applicants may format the specified forms making due provision for incorporation of the requested information.

2.4 Conflict of Interest

- 2.4.1 An Applicant shall not have a conflict of interest that may affect the Selection Process or the Consultancy (the "Conflict of Interest"). Any Applicant found to have a Conflict of Interest shall be disqualified. In the event of disqualification, the Authority shall forfeit and appropriate the Performance Security, if available, as mutually agreed genuine preestimated compensation and damages payable to the Authority for, inter alia, the time, cost and effort of the Authority including consideration of such Applicant's Proposal, without prejudice to any other right or remedy that may be available to the Authority hereunder or otherwise.
- 2.4.2 The Authority requires that the Consultant provides professional, objective, and impartial advice and at all times hold the Authority's interests paramount, avoid conflicts with other assignments or its own interests, and act without any consideration for future work. The Consultant shall not accept or engage in any assignment that would be in conflict with its prior or current obligations to other clients, or that may place it in a position of not being able to carry out the assignment in the best interests of the Authority.
- 2.4.3 Some guiding principles for identifying and addressing Conflicts of Interest have been illustrated in the Guidance Note at Schedule-3. Without limiting the generality of the above, an Applicant shall be deemed to have a Conflict of Interest affecting the Selection Process, if:
 - the Applicant, its consortium member (the "Member") or Associates (a) (or any constituent thereof) and any other Applicant, its consortium member or Associate (or any constituent thereof) have common controlling shareholders or other ownership interest; provided that this disqualification shall not apply in cases where the direct or indirect shareholding or ownership interest of an Applicant, its Member or Associate (or any shareholder thereof having a shareholding of more than 5 per cent of the paid up and subscribed share capital of such Applicant, Member or Associate, as the case may be) in the other Applicant, its consortium member or Associate is less than 5% (five per cent) of the subscribed and paid up equity share capital thereof. For the purposes of this Clause 2.3.3(a), indirect shareholding held through one or more intermediate persons shall be computed as follows: (aa) where any intermediary is controlled by a person through management control or otherwise, the entire shareholding held by such controlled intermediary in any other person (the "Subject Person") shall be taken into account for computing the shareholding of such controlling person in the Subject Person; and (bb) subject always to Sub-clause (aa) above, where a

person does not exercise control over an intermediary, which has shareholding in the Subject Person, the computation of indirect shareholding of such person in the Subject Person shall be undertaken on a proportionate basis; provided, however, that no such shareholding shall be reckoned under this Sub-clause (bb) if the shareholding of such person in the intermediary is less than 26% (twenty six per cent) of the subscribed and paid up equity shareholding of such intermediary; or

(b) a constituent of such Applicant is also a constituent of another Applicant;

or

- such Applicant or its Associate receives or has received any direct or indirect subsidy or grant from any other Applicant or its Associate; or
- (d) such Applicant has the same legal representative for purposes of this Application as any other Applicant; or
- (e) such Applicant has a relationship with another Applicant, directly or through common third parties, that puts them in a position to have access to each others' information about, or to influence the Application of either or each of the other Applicant; or
- (f) there is a conflict among this and other consulting assignments of the Applicant (including its personnel and other members, if any) and any subsidiaries or entities controlled by such Applicant or having common controlling shareholders. The duties of the Consultant will depend on the circumstances of each case. While providing consultancy services to the Authority for this particular assignment, the Consultant shall not take up any assignment that by its nature will result in conflict with the present assignment; or
- (g) a firm which has been engaged by the Authority to provide goods or works or services for a project, and its Associates, will be disqualified from providing consulting services for the same project save and except as provided in Clause 2.3.4; conversely, a firm hired to provide consulting services for the preparation or implementation of a project, and its Members or Associates, will be disqualified from subsequently providing goods or works or services related to the same project; or
- (h) the Applicant, its Member or Associate (or any constituent thereof), and the bidder or Concessionaire, if any, for the Project, its contractor(s) or sub-contractor(s) (or any constituent thereof) have common controlling shareholders or other ownership interest; provided that this disqualification shall not apply in cases where the direct or indirect shareholding or ownership interest of an Applicant.

its Member or Associate (or any shareholder thereof having a shareholding of more than 5% (five per cent) of the paid up and subscribed share capital of such Applicant, Member or Associate, as the case may be,) in the bidder or Concessionaire, if any, or its contractor(s) or sub-contractor(s) is less than 5% (five per cent) of the paid up and subscribed share capital of such Concessionaire or its contractor(s) or sub-contractor(s); provided further that this disqualification shall not apply to ownership by a bank, insurance company, pension fund or a Public Financial Institution referred to in section 4A of the Companies Act 1956. For the purposes of this Sub-clause (h), indirect shareholding shall be computed in accordance with the provisions of Sub-clause (a) above.

For purposes of this RFP, Associate means, in relation to the Applicant, a person who controls, is controlled by, or is under the common control with such Applicant, or is deemed or published as an "Associate Office"; or has a formal arrangement such as tie up for client referral or technology sharing, joint venture with the Applicant (the "Associate"); provided, however, that if the Applicant has any formal arrangement such as consortium membership in a consortium of advisers/ consultants for a particular assignment/project, not being this project, with any other person, then such other person shall not be treated to be an Associate of the Applicant solely due to the reason of forming such consortium. As used in this definition, the expression "control" means, with respect to a person which is a company or corporation, the ownership, directly or indirectly, of more than 50% (fifty per cent) of the voting shares of such person, and with respect to a person which is not a company or corporation, the power to direct the management and policies of such person by operation of law or by contract.

2.4.4 An Applicant eventually appointed to provide Consultancy for this Project, its Associates, affiliates and the PPP Expert, shall be disqualified from subsequently providing goods or works or services related to the construction and operation of the same Project and any breach of this obligation shall be construed as Conflict of Interest; provided that the restriction herein shall not apply after a period of 3 (three) years from the completion of this assignment or to any consulting assignments granted by banks/ lenders at any time; provided further that this restriction shall not apply to consultancy/ advisory services performed for the Authority in continuation of this Consultancy or to any subsequent consultancy/ advisory services performed for the Authority in accordance with the rules of the Authority. For the avoidance of doubt, an entity affiliated with the Consultant shall include a partner in the Consultant's firm or a person who holds more than 5% (five per cent) of the subscribed and paid up share capital of the Consultant, as the case may be, and any Associate thereof.

For the avoidance of doubt, in the event that: (a) the PPP Expert of an Applicant/ Consultant was a partner or an employee of another firm, which attracts the provisions relating to Conflict of Interest hereunder; and (b) was directly or indirectly associated with any assignment that causes a Conflict of Interest hereunder, then such PPP Expert shall be deemed to suffer from Conflict of Interest for the purpose hereof.

2.4.5 In the event that the Consultant, its Associates or affiliates are auditors or financial advisers to any of the bidders for the Project, they shall make a disclosure to the Authority as soon as any potential conflict comes to their notice but in no case later than 7 (seven) days from the opening of the RFQ applications for the Project and any breach of this obligation of disclosure shall be construed as Conflict of Interest. The Authority shall, upon being notified by the Consultant under this Clause 2.3.5, decide whether it wishes to terminate this Consultancy or otherwise, and convey its decision to the Consultant within a period not exceeding 15 (fifteen) days.

2.5 Number of Proposals

No Applicant or its Associate shall submit more than one Application for the Consultancy. An Applicant applying individually or as an Associate shall not be entitled to submit another application either individually or as a member of any consortium, as the case may be.

2.6 Cost of Proposal

The Applicants shall be responsible for all of the costs associated with the preparation of their Proposals and their participation in the Selection Process including subsequent negotiation, visits to the Authority, Project site etc. The Authority will not be responsible or in any way liable for such costs, regardless of the conduct or outcome of the Selection Process.

2.6.1 Visit to the Authority and verification of information

Applicants are encouraged to submit their respective Proposals after visiting the office of the Authority and ascertaining for themselves the availability of documents and other data with the Authority, Applicable Laws and regulations or any other matter considered relevant by them.

2.6.2 Acknowledgement by Applicant

It shall be deemed that by submitting the Proposal, the Applicant has:

- (a) made a complete and careful examination of the RFP;
- (b) received all relevant information requested from the Authority;
- (c) accepted the risk of inadequacy, error or mistake in the information provided in the RFP or furnished by or on behalf of the Authority or

relating to any of the matters referred to in Clause 2.6 above;

- (d) satisfied itself about all matters, things and information, including matters referred to in Clause 2.6 herein above, necessary and required for submitting an informed Application and performance of all of its obligations there under;
- (e) acknowledged that it does not have a Conflict of Interest; and
- (f) agreed to be bound by the undertaking provided by it under and in terms hereof.

The Authority shall not be liable for any omission, mistake or error in respect of any of the above or on account of any matter or thing arising out of or concerning or relating to RFP or the Selection Process, including any error or mistake therein or in any information or data given by the Authority.

2.7 Right to reject any or all Proposals

- 2.7.1 Notwithstanding anything contained in this RFP, the Authority reserves the right to accept or reject any Proposal and to annul the Selection Process and reject all Proposals, at any time without any liability or any obligation for such acceptance, rejection or annulment, and without assigning any reasons thereof.
- 2.7.2 Without prejudice to the generality of Clause 2.8.1, the Authority reserves the right to reject any Proposal if:
- (a) at any time, a material misrepresentation is made or discovered, or
- (b) the Applicant does not provide, within the time specified by the Authority, the supplemental information sought by the Authority for evaluation of the Proposal.

Misrepresentation/ improper response by the Applicant may lead to the disqualification of the Applicant. If the Applicant is the Lead Member of a consortium, then the entire consortium may be disqualified / rejected. If such disqualification / rejection occurs after the Proposals have been opened and the highest ranking Applicant gets disqualified / rejected, then the Authority reserves the right to consider the next best Applicant, or take any other measure as may be deemed fit in the sole discretion of the Authority, including annulment of the Selection Process.

2.8 Integrity Pact

Integrity Pact (IP) shall cover this RFP throughout its various phases, and IP would be deemed as a part of the contract. The Applicant should sign and submit a 'Integrity Pact' to be executed between the Applicant and the Authority as per the format at Form-9, and shall be submitted along with the Bid Enclosure in the manner provided in this RFP. IP would be

implemented through the following Independent External Monitor ("IEM") for this Bid:

Sri. Prem Chand Pankaj,

Ex-CMD, NEEPCO, M 402, Pioneer Park,

Sector-61, Golf Course Extn. Road,

Gurgaon, Haryana -Mob: 9717433886

Email: prempankaj@gmail.com

DOCUMENTS

2.9 Contents of the RFP

2.9.1 This RFP comprises the Disclaimer set forth hereinabove, the contents as listed below and will additionally include any Addendum / Amendment issued in accordance with Clause 2.11:

Request for Proposal

Section I Notice Inviting Online Tender

Section II Instructions to Applicants for Online Bid Submissions

Section III

- 1. Introduction
- 2. Instructions to Applicants
- 3. Criteria for Evaluation
- 4. Fraud and corrupt practices
- 5. Pre-Proposal Conference
- 6. Miscellaneous

Schedules

- 1 Terms of Reference
- 2 Form of Agreement
- 3 Guidance Note on Conflict of Interest

Appendices

Appendix - I

Technical Proposal

Form 1: Letter of Proposal

Form 2: Particulars of the applicant

Form 3: Statement of Legal Capacity

Form 4: Financial Capacity of Applicant

Form 5: Power of Attorney

Form 6: Particulars of Key Personnel

Form 7: Abstract of Eligible Assignments of Key Personnel

Form 7A: Abstract of other relevant experience of the Applicant

Form 8: Abstract of Eligible Assignments of Key Personnel

Form 8A: Abstract of other relevant experience of the Key Personnel

Form 9: Eligible Assignments of Applicants

Form 9A: Other relevant assignments of the Applicant

Form 10: Eligible Assignments of Key Personnel

Form 10A: Other relevant assignments of the Key Personnel

Form 11: CV of Professional Personnel

Form 12: Integrity Pact

Appendix - II

Form 1: Covering Letter

Form 2: Reference Format of Financial Proposal

Project Implementation Schedule

Layout of Berth No. 13

2.10 Clarifications

2.10.1 Applicants requiring any clarification on the RFP may send their queries to the Authority in writing before the date mentioned in the Schedule of Selection Process I-Notice Inviting Online Tender. The envelopes shall clearly bear the following identification:

"QUERIES/REQUEST FOR ADDITIONAL INFORMATION CONCERNING RFP FOR APPOINTMENT OF TRANSACTION ADVISER FOR CONCESSIONING OF BERTH NO.13 FOR HANDLING LIQUID BULK / LPG / POL PRODUCTS / CHEMICALS ON CAPTIVE / PPP MODE "

The Authority shall endeavour to respond to the queries within the period specified therein but no later than 7 (seven) days prior to the PDD. The Authority will post the reply to all such queries on the E-Tender portal and Official Website and copies thereof will also be circulated by e-mail to all Applicants who have purchased the RFP document without identifying the source of queries.

2.10.2The Authority reserves the right not to respond to any queries or provide any clarifications, in its sole discretion, and nothing in this Clause 2.10 shall be construed as obliging the Authority to respond to any question or to provide any clarification.

2.11.3 Amendment of RFP

At any time prior to the deadline for submission of Proposal, the Authority may, for any reason, whether at its own initiative or in response to clarifications requested by an Applicant, modify the RFP document by the issuance of Addendum/ Amendment and posting it on the E-Tender portal and Official Website.prior to the deadline for submission of Proposal, the Authority may, for any reason, whether at its own initiative or in response to clarifications requested by an Applicant, modify the RFP document by the issuance of Addendum/ Amendment and posting it on the Official Website and by conveying the same to the prospective Applicants (who have purchased the RFP document) by e-mail. All such amendments will be notified in writing through e-mail to all Applicants who have purchased the RFP document. The amendments will also be posted on the Official Website along with the revised RFP containing the amendments and will

be binding on all Applicants.

In order to afford the Applicants a reasonable time for taking an amendment into account, or for any other reason, the Authority may, in its sole discretion, extend the PDD.

C. PREPARATION AND SUBMISSION OF PROPOSAL

2.10 Language

The Proposal with all accompanying documents (the "Documents") and all communications in relation to or concerning the Selection Process shall be in English language and strictly on the forms provided in this RFP. No supporting document or printed literature shall be submitted with the Proposal unless specifically asked for and in case any of these Documents is in another language, it must be accompanied by an accurate translation of all the relevant passages in English, in which case, for all purposes of interpretation of the Proposal, the translation in English shall prevail.

2.11 Format and signing of Proposal

- 2.11.1 The Applicant shall provide all the information sought under this RFP. The Authority would evaluate only those Proposals that are received in the specified forms and complete in all respects.
- 2.11.2 The Applicant shall prepare the Proposal together with the enclosures required to be submitted along therewith pursuant to this RFP in ORIGINAL.
- 2.11.3 The Proposal shall be typed or written in indelible ink and signed by the authorised signatory of the Applicant who shall initial each page, in blue ink. In case of printed and published Documents, only the cover shall be initialed. All the alterations, omissions, additions, or any other amendments made to the Proposal shall be initialled by the person(s) signing the Proposal. The Proposals must be properly signed by the authorised representative (the "Authorised Representative") as detailed below:
 - i. by the proprietor, in case of a proprietary firm; or
 - ii. by a partner, in case of a partnership firm and/or a limited liability partnership; or
 - iii. by a duly authorised person holding the Power of Attorney, in case of a Limited Company or a corporation; or
 - iv. by the authorised representative of the Lead Member, in case of consortium.

A copy of the Power of Attorney certified by a notary public in the form specified in Appendix-I (Form-4) shall accompany the Proposal (if required).

2.11.4 Applicants should note the PDD, as specified in the Notice Inviting Tender, for submission of Proposals. Except as specifically provided in this RFP, no supplementary material will be entertained by the Authority, and that

evaluation will be carried out only on the basis of Documents received by the closing time of PDD as specified in Clause 2.17.1. Applicants will ordinarily not be asked to provide additional material information or documents subsequent to the date of submission, and unsolicited material if submitted will be summarily rejected.

The Applicants would provide all information as per this Proposal Document as well as subsequent clarifications/explanations furnished by NMPA. New Mangalore Port Authority reserves the right to reject any non responsive proposal.

2.12 Technical Proposal

- 2.12.1 Applicants shall submit the technical proposal in the formats at Appendix-I (the "Technical Proposal").
- 2.12.2 While submitting the Technical Proposal, the Applicant shall, in particular, ensure that:
 - (a) all forms are submitted in the prescribed formats and signed by the prescribed signatories;
 - (b) power of attorney, if applicable, is executed as per Applicable Laws;
 - (c) CVs of all Key Personnel have been included;
 - (d) Key Personnel have been proposed only if they meet the Conditions of Eligibility laid down at Clause 2.2.2 (D) of the RFP;
 - (e) no alternative proposal for any Key Personnel is being made and only one CV for each position has been furnished;
 - (f) the CVs have been recently signed and dated, i.e. after the date of issue of this Request for Proposal document in blue ink by the respective Personnel and countersigned by the Applicant. Photocopy or unsigned/countersigned CVs shall be rejected;
 - (g) the CVs shall contain an undertaking from the respective Key Personnel about his/her availability for the duration specified in the RFP;
 - (h) Key Personnel proposed have good working knowledge of English language;
 - (i) Key Personnel would be available for the period indicated in the TOR;
 - (j) no Key Personnel should have attained the age of 70 years at the time of submitting the proposal; and
 - (k) the proposal is responsive in terms of Clause 2.21.3.
 - (I) the proposal shall include EMD and Tender document fee in the form of DD.
 - (m)The work order and completion certificate from the client of having successfully completed eligible works.

- 2.12.3 Failure to comply with the requirements spelt out in this Clause 2.14 shall make the Proposal liable to be rejected.
- 2.12.4 If an individual Key Personnel makes a false averment regarding his qualification, experience or other particulars, he shall be liable to be debarred for any future assignment of the Authority for a period of 3 (three) years. The award of this Consultancy to the Applicant may also be liable to cancellation in such an event.
- 2.12.5 The Technical Proposal shall not include any financial information relating to the Financial Proposal.
- 2.12.6 The proposed team shall include experts and specialists (the "Professional Personnel") in their respective areas of expertise and managerial/support staff (the "Support Personnel") such that the Consultant should be able to complete the Consultancy within the specified time schedule. The Key Personnel specified in Clause 2.1.4 shall be included in the proposed team of Professional Personnel. The team shall comprise other competent and experienced professional personnel in the relevant areas of expertise (where applicable) as required for successful completion of this Consultancy. The CV of each such professional personnel, if any, should also be submitted in the format at Form-11 of Appendix-I.
- 2.12.7 An Applicant may, from time to time, if it considers necessary, propose suitable Sub-Consultants in specific areas of expertise (where applicable). Credentials of such Sub-Consultants should be submitted by the Applicant in Form-12 of Appendix-I. A Sub-Consultant, however, shall not be a substitute for any Key Personnel.
- 2.12.8 The Authority reserves the right to verify all statements, information and documents, submitted by the Applicant in response to the RFP. Any such verification or the lack of such verification by the Authority to undertake such verification shall not relieve the Applicant of its obligations or liabilities hereunder nor will it affect any rights of the Authority there under.
- 2.12.9 In case it is found during the evaluation or at any time before signing of the Agreement or after its execution and during the period of thereof, that one or more of the eligibility conditions have not been met by the Applicant or the Applicant has made material misrepresentation or has given any materially incorrect or false information, the Applicant shall be disqualified forthwith if not yet appointed as the Consultant either by issue of the LOA or entering into of the Agreement, and if the Selected Applicant has already been issued the LOA or has entered into the Agreement, as the case may be, the same shall, notwithstanding anything to the contrary contained therein or in this RFP, be liable to be terminated, by a communication in writing by the Authority without the Authority being liable in any manner whatsoever to the Selected Applicant or Consultant, as the case may be.

In such an event, the Authority shall forfeit and appropriate the Performance Security, if available, as mutually agreed pre-estimated compensation and damages payable to the Authority for, inter alia, time, cost and effort of the Authority, without prejudice to any other right or remedy that may be available to the Authority.

2.13 Financial Proposal

- 2.13.1 Applicants shall submit the financial proposal in the formats at Appendix-II (the "Financial Proposal") clearly indicating the total cost of the Consultancy (Item F of Form-2 of Appendix II) in both figures and words, in Indian Rupees, and signed by the Applicant's Authorised Representative. In the event of any difference between figures and words, the amount indicated in words shall prevail. In the event of a difference between the arithmetic total and the total shown in the Financial Proposal, the lower of the two shall prevail.
- 2.13.2 While submitting the Financial Proposal, the Applicant shall ensure the following:
 - (i) All the costs associated with the assignment shall be included in the Financial Proposal. These shall normally cover remuneration for all the Personnel (Expatriate and Resident, in the field, office etc), accommodation, air fare, equipment, printing of documents, etc. The total amount indicated in the Financial Proposal shall be without any condition attached or subject to any assumption, and shall be final and binding. In case any assumption or condition is indicated in the Financial Proposal, it shall be considered non-responsive and liable to be rejected.
 - (ii) The Financial Proposal shall take into account all expenses and tax liabilities. For the avoidance of doubt, it is clarified that all taxes shall be deemed to be included in the costs shown under different items of the Financial Proposal. Further, all payments shall be subject to deduction of taxes at source as per Applicable Laws.
 - (iii) Costs (including break down of costs) shall be expressed in INR.

2.14 Submission of Proposal

Bidders shall submit their bid through E-tenders https://eprocure.gov.in/cppp/ of CPP.

2.15 Proposal Due Date

- 2.15.1 Proposal must be uploaded in the website https://eprocure.gov.in/cppp/ not later than the time and date indicated in the online bid reference.
- 2.15.2 The Authority may, in its sole discretion, extend the PDD by issuing an Addendum in accordance with Clause 2.11 uniformly for all Applicants.
- 2.15.3 Since the bid document is downloaded from web site and uploaded through e- Tender, the bidder shall give an undertaking that no changes have been made in document.
- 2.15.4 For both Technical and Financial evaluation purpose, e-tender only will be considered.

2.16 Modification/ substitution/ withdrawal of Proposals

- 2.16.1 The Applicant may modify, substitute, or withdraw its Proposal after submission, provided that written notice of the modification, substitution, or withdrawal is received by the Authority prior to PDD. No Proposal shall be modified, substituted, or withdrawn by the Applicant on or after the PDD.
- 2.16.2 The modification, substitution, or withdrawal notice shall be prepared, sealed, marked, and delivered in accordance with Clause 2.16, with the envelopes being additionally marked "MODIFICATION", "SUBSTITUTION" or "WITHDRAWAL", as appropriate.
- 2.16.3 Any alteration / modification in the Proposal or additional information or material supplied subsequent to the PDD, unless the same has been expressly sought for by the Authority, shall be disregarded.

2.17 Bid Security

The Applicant shall furnish as part of its Proposal, a bid security of Rs.59,000.00 (Rupees Fifty nine thousand only) through NEFT in favour of the Financial Advisor & Chief Accounts Officer (FA&CAO), NMPA. The "Bid Security", is returnable not later than 30 days from PDD except in case of the two highest ranked Applicants as required in Clause 2.24.1. In the event that the first ranked Applicant commences the assignment as required in Clause 2.29, the second ranked Applicant, who has been kept in reserve, shall be returned its Bid Security forthwith, but in no case not later than 120 (one hundred and twenty) days from PDD. The Selected Applicant's Bid Security shall be returned, upon the Applicant signing the Agreement and completing the Deliverables assigned to it for the first 2 (two) months of the Consultancy in accordance with the provisions thereof.

The benefit of Exemption of EMD to all Micro and small Enterprises (MSE) will be allowed. The bidder shall upload with their offer, proof of their being MSE registered with District Industries Centre (DIC) or Khadhi and village industries commission or Khadhi and Industries board (KVIV) or Coir board or National Small Scale Industries Corporation (NSIC) or Directorate of handicrafts and handlooms or Udhyog Aadhar Memorandum or any other body specified by Ministry of MSME, will be considered.

- 2.17.1 Any Bid not accompanied by the Bid Security shall be rejected by the Authority as nonresponsive.
- 2.17.2The Authority shall not be liable to pay any interest on the Bid Security and the same shall be interest free.
- 2.17.3 The Applicant, by submitting its Application pursuant to this RFP, shall be deemed to have acknowledged that without prejudice to the Authority's any other right or remedy hereunder or in law or otherwise, the Bid Security shall be forfeited and appropriated by the Authority as the mutually agreed preestimated compensation and damage payable to the Authority for, inter alia,

time, cost and effort of the Authority in regard to the RFP including the consideration and evaluation of the Proposal under the following conditions:

- (a) If an Applicant submits a non-responsive Proposal;
- (b) If an Applicant engages in any of the Prohibited Practices specified in Clause 4 of this RFP;
- (c) If an Applicant withdraws its Proposal during the period of its validity as specified in this RFP and as extended by the Applicant from time to time:
- (d) In the case of the Selected Applicant, if the Applicant fails to reconfirm its commitments during negotiation as required vide Clause 2.24.1;
- (e) In the case of a Selected Applicant, if the Applicant fails to sign the Agreement or commence the assignment as specified in Clause 2.28 and 2.29 respectively; or
- (f) If the Applicant is found to have a Conflict of Interest as specified in Clause 2.3.

D. EVALUATION PROCESS

2.18 Evaluation of Proposals

- 2.18.1 Prior to evaluation of Proposals, the Authority will determine whether each Proposal is responsive to the requirements of the RFP. A Proposal shall be considered responsive only if:
 - a. It is received by the Proposal Due Date (PDD) and Time.
 - b. It is Digitally Signed.
 - c. It contains the information and documents as required in the RFP Document.
 - d. It contains information in formats specified in the RFP Document.
 - e. It mentions the validity period as set out in the document.
 - f. It provides the information in reasonable detail. The Port Trust reserves the right to determine whether the information has been provided in reasonable detail.
 - g. There are no significant inconsistencies between the proposal and the supporting documents.
 - h. The Technical qualification conforms to as specified in the qualification criteria.
 - i. It contains sealed and signed copy of RFP document, amendments and addendum (if any) issued by New Mangalore Port are submitted with initials on all pages.
 - j. the EMD and Tender document fee have been paid.

- 2.18.2 The Authority reserves the right to reject any Proposal which is non-responsive and no request for alteration, modification, substitution or withdrawal shall be entertained by the Authority in respect of such Proposals.
- 2.18.3 The Authority shall subsequently examine and evaluate Proposals in accordance with the Selection Process specified at Section 3 of this RFP.
- 2.18.4 After the technical evaluation, the Authority shall prepare a list of prequalified Applicants in terms of Clause 3.1.1 for opening of their Financial Proposals. A date, time and venue will be notified to all Applicants for announcing the result of evaluation and opening of Financial Proposals. The opening of Financial Proposals will be through E-Tender. The Authority will not entertain any query or clarification from Applicants who fail to qualify at any stage of the Selection Process. The financial evaluation and final ranking of the Proposals shall be carried out in terms of Clauses 3.2 and 3.3
- 2.18.5 Applicants are advised that Selection shall be entirely at the discretion of the Authority. Applicants shall be deemed to have understood and agreed that the Authority shall not be required to provide any explanation or justification in respect of any aspect of the Selection Process or Selection.
- 2.18.6 Any information contained in the Proposal shall not in any way be construed as binding on the Authority, its agents, successors or assigns, but shall be binding against the Applicant if the Consultancy is subsequently awarded to it.

2.19 Confidentiality

Information relating to the examination, clarification, evaluation, and recommendation for the selection of Applicants shall not be disclosed to any person who is not officially concerned with the process or is not a retained professional adviser advising the Authority in relation to matters arising out of, or concerning the Selection Process. The Authority shall treat all information, submitted as part of the Proposal, in confidence and shall require all those who have access to such material to treat the same in confidence. The Authority may not divulge any such information unless it is directed to do so by any statutory entity that has the power under law to require its disclosure or to enforce or assert any right or privilege of the statutory entity and/or the Authority or as may be required by law or in connection with any legal process.

2.20 Clarifications

2.20.1 To facilitate evaluation of Proposals, the Authority may, at its sole discretion, seek clarifications from any Applicant regarding its Proposal. Such clarification(s) shall be provided within the time specified by the Authority for this purpose. Any request for clarification(s) and all clarification(s) in response thereto shall be in writing.

2.20.2 If an Applicant does not provide clarifications sought under Clause 2.23.1 above within the specified time, its Proposal shall be liable to be rejected. In case the Proposal is not rejected, the Authority may proceed to evaluate the Proposal by construing the particulars requiring clarification to the best of its understanding, and the Applicant shall be barred from subsequently questioning such interpretation of the Authority.

E. APPOINTMENT OF CONSULTANT

2.21 Negotiations

- 2.21.1 The Selected Applicant may, if necessary, be invited for negotiations. The negotiations shall generally not be for reducing the price of the Proposal, but will be for re-confirming the obligations of the Consultant under this RFP. Issues such as deployment of Key Personnel, understanding of the RFP, methodology and quality of the work plan shall be discussed during negotiations. A Key Personnel who did not score 70% (seventy per cent) marks as required under Clause 3.1.2 shall be replaced by the Applicant with a better candidate to the satisfaction of the Authority. In case the Selected Applicant fails to reconfirm its commitment, the Authority reserves the right to designate the next ranked Applicant as the Selected Applicant and invite it for negotiations.
- 2.21.2 The Authority will examine the CVs of all other Professional Personnel and those not found suitable shall be replaced by the Applicant to the satisfaction of the Authority.
- 2.21.3 The Authority will examine the credentials of all Sub-Consultants proposed for this Consulting services and those not found suitable shall be replaced by the Applicant to the satisfaction of the Authority.

2.22 Substitution of Key Personnel

- 2.22.1 The Authority will not normally consider any request of the Selected Applicant for substitution of Key Personnel during negotiations as the ranking of the Applicant is based on the evaluation of Key Personnel and any change therein may upset the ranking. Substitution, will, however be permitted if the Key Personnel is not available for reasons of any incapacity or due to health, subject to equally or better qualified and experienced personnel being provided to the satisfaction of the Authority.
- 2.22.2 The Authority expects the Key Personnel to be available during implementation of the Agreement. The Authority will not consider substitution of the Key Personnel except for reasons of any incapacity or due to health. Such substitution shall ordinarily be limited to one Key Personnel subject to equally or better qualified and experienced personnel being provided to the satisfaction of the Authority. As a

condition to such substitution, a sum equal to 10% (ten per cent) of the remuneration weightage specified for the original Key Personnel in the scoring criteria as per clause 3.1.3 shall be deducted from the payments due to the Consultant. In the case of a second substitution hereunder, such deduction shall be 20% (twenty per cent) of the remuneration specified for the original Key Personnel.

The deduction shall be made as follows:-

For first substitution	Percentage of Remuneration Weightage for the original Key Personnel from the Contract Agreement Amount	Х	10%
For second substitution	Percentage of Remuneration Weightage for the original Key = Personnel from the Revised Agreement Amount arrived after first substitution	X	20%

- 2.22.3 Any further substitution may lead to disqualification of the Applicant or termination of the Agreement.
- 2.22.4 Substitution of the Team Leader will not normally considered and may lead to disqualification of the Applicant or termination of the Agreement.

2.23 Indemnity

The Consultant shall, subject to the provisions of the Agreement, indemnify the Authority, for an amount not exceeding 3 (three) times the value of the Agreement, for any direct loss or damage that is caused due to any deficiency in Services.

2.24 Award of Consultancy

After selection, a Letter of Award (the "LOA") shall be issued, in duplicate, by the Authority to the Selected Applicant and the Selected Applicant shall, within 7 (seven) days of the receipt of the LOA, sign and return the duplicate copy of the LOA in acknowledgement thereof. In the event the duplicate copy of the LOA duly signed by the Selected Applicant is not received by the stipulated date, the Authority may, unless it consents to extension of time for submission thereof, cancel the LOA and the next highest ranking Applicant may be considered.

2.25 Execution of Agreement

After acknowledgement of the LOA as aforesaid by the Selected Applicant, it shall execute the Agreement within 7 (seven) days from the date of issue of LoA. The Selected Applicant shall not be entitled to seek any deviation in the Agreement.

The successful Bidder shall prepare and submit agreement format in line with the draft enclosed to the proposal document. On Approval of the Authority the agreement shall be signed by both the Bidder and Authority, after payment of stamp duties as may be necessary by the successful Bidder. The successful bidder shall also bear the cost for making of 10 sets of Agreement copies.

2.26 Commencement of Assignment

The Consultant shall commence the Consultancy within seven days of the date of the Agreement, or such other date as may be mutually agreed. If the Consultant fails to either sign the Agreement as specified in Clause 2.28 or commence the assignment as specified herein, the Authority may invite the second ranked Applicant for negotiations. In such an event, the LOA or the Agreement, as the case may be, may be cancelled /terminated.

2.27 Proprietary data

Subject to the provisions of Clause 2.22, all documents and other information provided by the Authority or submitted by an Applicant to the Authority shall remain or become the property of the Authority. Applicants and the Consultant, as the case may be, are to treat all information as strictly confidential. The Authority will not return any Proposal or any information related thereto. All information collected, analysed, processed or in whatever manner provided by the Consultant to the Authority in relation to the Consultancy shall be the property of the Authority.

3. CRITERIA FOR EVALUATION

3.1 Evaluation of Technical Proposals

- 3.1.1 In the first stage, the Technical Proposal will be evaluated on the basis of Applicant's experience, its understanding of TOR, proposed methodology and Work Plan, and the experience of Key Personnel. Only those Applicants whose Technical Proposals score 70 points or more out of 100 shall be ranked as per score achieved by them, from highest to the lowest technical score (ST).
- 3.1.2 Each Key Personnel must score a minimum of 70% marks except as provided herein. A Proposal shall be rejected if the PPP Expert scores less than 70% (seventy per cent) marks or any two of the remaining Key Personnel score less than 70% (seventy per cent) marks.
- 3.1.3 The scoring criteria to be used for evaluation shall be as follows.

Item Code	Parameter	Max. Marks	Criteria	Scoring	Remuneration weightage for Key Personnel in percentage (ref. Cl. 2.22.2)
A	Relevant Experience of the Applicant	25			
i)	Number of Eligible Assignments undertaken by the Applicant shall have, over the past 7 (Seven) years preceding to PDD	15	More than10 Projects 9-10 projects 6-8 projects 5 projects	15 12.5 11 10.5	
ii)	Overall turnover of the firm	10	More than 5 Crore >2 Cr. upto 5 Crore >1 Cr. upto 2 Crore	10 8.5 7	
	Proposed Methodology and works plan	5	Evaluation based on quality of submission		
В	Relevant Experience of	70			

	the Key Personnel		TOI Concessioning of bertill 3 for manufing		
i	TEAM LEADER	25			25%
	-PPP Expert				
	i. Length / status	5.0	Permanent Expert with	5.00	
	of Association		more than 15 years		
	with Firm		experience or an		
			Independent Expert with		
			more than 20years		
			experience		
			Permanent Expert with	4.25	
			10-15 years experience		
			or an Independent		
			Expert with 15 or more		
			than 15years experience	0.50	
			Permanent Expert with	3.50	
			5-10 years experience or an Independent		
			Expert with 10 or more		
			than 10years experience		
	ii. Qualification	5.0	CA/ICWA/CFA/MBA(Fin)	5.00	
			or equivalent		
	iii. Total Work	5.0	More than 20yrs	5.00	
	Experience		16-20 yrs	4.25	
			15 yrs	3.50	
	Adequacy in	10.0	>10 Projects	10.00	
	Assignment		9-10 projects	8.34	
	/Job		6-8 projects	7.34	
	iv. Experience in		5 projects	7.00	
	carrying out				
	Eligible				
	assignment				000/
ii	PORT SECTOR	20			20%
	i. Length / status	2.50	Permanent Expert with	2.50	
	of Association	2.00	more than 10 years	2.50	
	with Firm		experience with the firm		
	***************************************		or an Independent		
			Expert with 10-15 years		
			experience		
			Permanent Expert with	2.125	
			5-10 years experience		
			with the firm or an		
			Independent Expert with		
			more than 10 years		

	Transa	action Adviso	over a rich an	g Elquiu Duik Oii	captive / 111 Mode
			experience	4 75	
			Permanent Expert with	1.75	
			less than 2-5 years		
			experience with the firm		
			or an Independent		
			Expert with more than 5		
	0 1.6	0.50	years experience	0.50	
	ii. Qualification	2.50	PhD / Master	2.50	
	iii. Total Work	5.00	more than 15 yrs	5.00	
	Experience		11-15 yrs	4.25	
			10 yrs	3.50	
	Adequacy in	10.00	More than10 Projects	10.00	
	Assignment		7-10 projects	9.00	
	/Job		4-6 projects	8.00	
	iv. Experience in		3 projects	7.00	
	carrying out				
	Eligible				
	assignment				
iii	LEGAL	15.00			15%
	CONSULTANT				
	i. Length / status	2.50	Permanent Expert with	2.50	
	of Association		more than 5 years		
	with Firm		experience with the firm		
			or an Independent		
			Expert with more than		
			10 years experience		
			Permanent Expert with	2.125	
			2-5 years experience		
			with the firm or an		
			Independent Expert with		
			more than 5 years		
			experience		
			Permanent Expert with	1.75	
			less than 2 years		
			experience with the firm		
			or an Independent		
			Expert with more than 2		
			years experience		
	ii. Qualification	2.50	PhD / Master / Bachelor	2.50	
			in law or any equivalent		
	::: T-4 ! \A/ !	5.00	degree in law.	F 00	
	iii. Total Work	5.00	more than 10 yrs	5.00	
	Experience		6-10 yrs 5 yrs	4.25 3.50	
			L - 100	ったり	

	Adequacy in	5.00	More than 5 Projects	5.00	
	Assignment		4- 5 projects	4.25	
	/Job		3 projects	3.50	
	iv. Experience in		, ,		
	carrying out				
	Eligible				
	assignment				
iv	ASSOCIATE CONSULTANT	10.00			10%
	i. Length / status	2.50	Permanent Expert with	2.50	
	of Association		more than 5 years		
	with Firm		experience with the firm		
			or an Independent		
			Expert with more than		
			10 years experience		
			Permanent Expert with	2.125	
			2-5 years experience		
			with the firm or an		
			Independent Expert with		
			more than 5 years		
			experience		
			Permanent Expert with	1.75	
			less than 2 years		
			experience with the firm		
			or an Independent		
			Expert with more than 2		
			years experience		
	ii. Qualification	2.50	PhD / Master/	2.50	
			CA/ACMA(ICMA)		
			/CFA/MBA(Fin) or		
			equivalent		
	iii. Total Work	2.50	more than 10 yrs	2.50	
	Experience		6-10 yrs	2.125	
	-		5 yrs	1.75	
	Adequacy in	2.50	More than5 Projects	2.50	
	Assignment		. 3- 5 projects	2.125	
	/Job		2 projects	1.75	
	iv. Experience in				
	carrying out				
	Eligible				
	assignment				

While awarding marks for the number of Eligible Assignments, the Applicant or Key Personnel, as the case may be, that has undertaken the highest number of Eligible Assignments shall be entitled to the maximum score for the respective category and all other competing Applicants or

respective Key Personnel, as the case may be, shall be entitled to a proportionate score. No score will be awarded to an Applicant/ Key Personnel for fulfilling the eligibility criteria of a minimum number of Eligible Assignments and only projects exceeding the eligibility criteria shall qualify for scoring. For the avoidance of doubt and by way of illustration, if the minimum number of Eligible Assignments for meeting the eligibility criteria is 3 (three), then an equivalent number will be ignored for each Applicant/ Key Personnel and only the balance remaining will be considered for awarding scores relating to the number of Eligible Assignments on a proportionate basis. However, for assigning scores in respect of the size and quality of Eligible Assignments, all Eligible Assignments of the Applicant/ Key Personnel shall be considered.

3.1.4 Eligible Assignments

OR

For the purposes of satisfying the Conditions of Eligibility and for evaluating the Proposals under this RFP, Consultancy / Transaction Advisory Services assignments granted by the government, regulatory commission, tribunal, statutory authority or public sector entity in respect of preparation of revenue model and/or preparation of transaction / bid documents and other similar assignments in relation to the bidding process or other transactions, as the case may be, for the following types of projects, for which the Bidder over the past 7 (seven) years preceding the PDD shall have successfully completed atleast 5 (five) assignments shall be deemed as eligible assignments (the "Eligible Assignments"):

- (a) An infrastructure project undertaken through Public Private Partnership (PPP) or other forms of private participation and having an estimated capital cost (excluding land) of at least Rs. 100 crore (Rupees One hundred crore only) in case of a project in India, and [US \$ 50 million for infrastructure projects undertaken elsewhere] through PPP or other forms of private participation (the "PPP Projects");
- (b) Privatization or disinvestment of government owned companies in India involving transfer of management control to a private sector entity where the financial transaction involves a payment of at least Rs.50 crore (Rupees Fifty crore) for such transfer (the "Disinvestment Projects");
 OR
- (c) An infrastructure project having an estimated capital cost (excluding land) of at least Rs. 100 crore (Rupees One hundred) in case of a project in India, and US \$ 50 million projects elsewhere, provided that the payment of professional fee to the Applicant was at least Rs.5 lakh (Rupees five lakh)] in case of a project in India, and [US \$ 25,000 (US\$ twenty five thousand) for any infrastructure project elsewhere (the "Other Projects");

Provided that all Eligible Assignments relating to projects based on

the same Model Concession Agreement and awarded by the same public entity shall be counted as one Eligible Assignment.

3.1.5 Other relevant assignments

Deleted

3.2 **Short-listing of Applicants**

Of the Applicants ranked as aforesaid, not more than 7 (seven) shall be pre-qualified and short-listed for financial evaluation in the second stage. However, if the number of such prequalified Applicants is less than 2 (two), the Authority may, in its sole discretion, pre-qualify the Applicant(s) whose Technical Score is less than the score specified in Clauses 3.1.1 and 3.1.2; provided that in such an event, the total number of pre-qualified and short-listed Applicants shall not exceed 2 (two).

3.3 Evaluation of Financial Proposal

- 3.3.1 In the second stage, the financial evaluation will be carried out as per this Clause 3.3. Each Financial Proposal will be assigned a financial score (S_F) as specified in Clause 3.3.3.
- 3.3.2 For financial evaluation, the total cost (excluding GST) indicated in the Financial Proposal will be considered.
- 3.3.3 The Authority will determine whether the Financial Proposals are complete, unqualified and unconditional. The cost indicated in the Financial Proposal shall be deemed as final and reflecting the total cost of services. Omissions, if any, in costing any item shall not entitle the Consultant to be compensated and the liability to fulfil its obligations as per the TOR within the total quoted price shall be that of the Consultant. The lowest Financial Proposal (FM) will be given a financial score (SF) of 100 points. The financial scores of other proposals will be computed as follows:

$$S_F = 100 \text{ x } F_M/F$$

(F = amount of Financial Proposal)

3.4 Combined and Final Evaluation

3.4.1 Proposals will finally be ranked according to their combined technical (ST) and financial (SF) scores as follows:

$$S = S_T \times T_w + S_F \times F_w$$

Where, T_w and F_w are weights assigned to Technical Proposal and Financial Proposal that shall be 0.7 and 0.3 respectively.

3.4.2 The Selected Applicant shall be the Applicant having the highest combined score. The second highest Applicant shall be kept in reserve and may be invited in its discretion for negotiations in case the first-ranked Applicant withdraws, or fails to comply with the requirements specified in Clauses 2.24, 2.28 and 2.29, as the case may be.

4. FRAUD AND CORRUPT PRACTICES

- 4.1 The Applicants and their respective officers, employees, agents and advisers shall observe the highest standard of ethics during the Selection Process. Notwithstanding anything to the contrary contained in this RFP, the Authority shall reject a Proposal without being liable in any manner whatsoever to the Applicant, if it determines that the Applicant has, directly or indirectly or through an agent, engaged in corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice (collectively the "Prohibited Practices") in the Selection Process. In such an event, the Authority shall, without prejudice to its any other rights or remedies, forfeit and appropriate the Performance Security, if available, as mutually agreed genuine pre-estimated compensation and damages payable to the Authority for, inter alia, time, cost and effort of the Authority, in regard to the RFP, including consideration and evaluation of such Applicant's Proposal.
- 4.2 Without prejudice to the rights of the Authority under Clause 4.1 hereinabove and the rights and remedies which the Authority may have under the LOA or the Agreement, if an Applicant or Consultant, as the case may be, is found by the Authority to have directly or indirectly or through an agent, engaged or indulged in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice during the Selection Process, or after the issue of the LOA or the execution of the Agreement, such Applicant or Consultant shall not be eligible to participate in any tender or RFP issued by the Authority during a period of 2 (two) years from the date such Applicant or Consultant, as the case may be, is found by the Authority to have directly or through an agent, engaged or indulged in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice, as the case may be.
- 4.3 For the purposes of this Clause, the following terms shall have the meaning hereinafter respectively assigned to them:
 - (a) "corrupt practice" means (i) the offering, giving, receiving, or soliciting, directly or indirectly, of anything of value to influence the action of any person connected with the Selection Process (for avoidance of doubt, offering of employment to or employing or engaging in any manner whatsoever, directly or indirectly, any official of the Authority who is or has been associated in any manner, directly or indirectly with the Selection Process or the LOA or has dealt with matters concerning the Agreement or arising there from, before or after the execution thereof, at any time prior to the expiry of one year from the date such official resigns or retires from or otherwise ceases to be in the service of the Authority, shall be deemed to constitute influencing the actions of a person connected with the Selection Process; or (ii) save as provided herein, engaging in any manner whatsoever, whether during the Selection Process or after the issue of the

LOA or after the execution of the Agreement, as the case may be, any person in respect of any matter relating to the Project or the LOA or the Agreement, who at any time has been or is a legal, financial or technical consultant/ adviser of the Authority in relation to any matter concerning the Project;

- (b) "fraudulent practice" means a misrepresentation or omission of facts or disclosure of incomplete facts, in order to influence the Selection Process;
- (c) "coercive practice" means impairing or harming or threatening to impair or harm, directly or indirectly, any persons or property to influence any person's participation or action in the Selection Process;
- (d) "undesirable practice" means (i) establishing contact with any person connected with or employed or engaged by the Authority with the objective of canvassing, lobbying or in any manner influencing or attempting to influence the Selection Process; or (ii) having a Conflict of Interest; and
- (e) "restrictive practice" means forming a cartel or arriving at any understanding or arrangement among Applicants with the objective of restricting or manipulating a full and fair competition in the Selection Process.

5. PRE-PROPOSAL CONFERENCE

- 5.1 Pre-Proposal Conference of the Applicants shall be convened at the designated date, time and place. Only those Applicants who have downloaded the RFP document from the Official Website of the Authority, shall be allowed to participate in the Pre-Proposal Conference. A maximum of two representatives of each Applicant shall be allowed to participate on production of an authority letter from the Applicant.
- 5.2 The Applicants will be free to seek clarifications and make suggestions for consideration of the Authority. The Authority shall endeavour to provide clarifications and such further information as it may, in its sole discretion, consider appropriate for facilitating a fair, transparent and competitive Selection Process.
- 5.3 The link to the Virtual Meeting (Microsoft Teams) will be shared only to those Bidders who have confirmed their participation and furnished valid email ID for communication.
- 5.4 Prospective Bidders may send queries to Chief Engineer (Civil)i/c through email chiefengineer@nmpt.gov.in or send online through CPP portal on or before 16/05/2022.
- 5.5 Consolidated reply / clarifications to all the queries received till **16/05/2022** will be uploaded on the CPP portal & NMPA website well in advance to the last date of submission of tenders. No queries will be entertained after **16/05/2022**.

6. MISCELLANEOUS

- 6.1 The Selection Process shall be governed by, and construed in accordance with, the laws of India and the Courts at Mangalore shall have exclusive jurisdiction over all disputes arising under, pursuant to and/or in connection with the Selection Process.
- 6.2 The Authority, in its sole discretion and without incurring any obligation or liability, reserves the right, at any time, to:
 - (a) suspend and/or cancel the Selection Process and/or amend and/or supplement the Selection Process or modify the dates or other terms and conditions relating thereto;
 - (b) consult with any Applicant in order to receive clarification or further information;
 - (c) retain any information and/or evidence submitted to the Authority by, on behalf of and/or in relation to any Applicant; and/or
 - (d) independently verify, disqualify, reject and/or accept any and all submissions or other information and/or evidence submitted by or on behalf of any Applicant.
- 6.3 It shall be deemed that by submitting the Proposal, the Applicant agrees and releases the Authority, its employees, agents and advisers, irrevocably, unconditionally, fully and finally from any and all liability for claims, losses, damages, costs, expenses or liabilities in any way related to or arising from the exercise of any rights and/or performance of any obligations hereunder, pursuant hereto and/or in connection herewith and waives any and all rights and/or claims it may have in this respect, whether actual or contingent, whether present or future.
- 6.4 All documents and other information supplied by the Authority or submitted by an Applicant shall remain or become, as the case may be, the property of the Authority. The Authority will not return any submissions made hereunder. Applicants are required to treat all such documents and information as strictly confidential.
- 6.5 The Authority reserves the right to make inquiries with any of the clients listed by the Applicants in their previous experience record.

SCHEDULES

Schedule – 1: Terms of Reference

SCHEDULE – 1 (See Clause 1.1.3)

RFP FOR APPOINTMENT OF TRANSACTION ADVISOR FOR CONCESSIONING OF BERTH NO. 13 FOR HANDLING LIQUID BULK / LPG / POL PRODUCTS / CHEMICALS ON CAPTIVE / PPP MODE AT NEW MANGALORE PORT

TERMS OF REFERENCE (TOR)

1. GENERAL

- 1.1 The Authority seeks the services of a qualified firm for acting as Transaction Adviser (the "Consultant") for appraising the Project, developing a revenue model and preparing bid documents for the Project (collectively the "Consultancy"). The Terms of Reference (the "TOR") and the scope of the Consultancy for this assignment are specified below.
- 1.2 The Consultant shall be guided in its assignment by the Captive Policy Guidelines and Model Concession Agreement (the "MCA") including Construction Standards, Operations & Maintenance Standards, Safety Standards and Performance Standards for Port Projects through Public Private Partnership.
- 1.3The Consultant shall be responsible for preparing the relevant Schedules of the Concession Agreement and for bringing out any special feature or requirement of the Project referred to in the Concession Agreement or applicable standards.
- 1.4 The Consultant shall assist the Authority by furnishing clarifications as required for the structuring, documentation and award of the Project.
- 1.5 The Consultant shall also participate in the pre-bid conferences with the Bidders of the Project and assist the Authority in clarifying the financial and legal aspects arising from the Bid Documents.
- 1.6 The Consultant shall make available the PPP Expert and other Key Personnel to attend and participate in meetings, conferences and discussions with the Authority and shall otherwise advise on and assist the Authority in the diverse commercial issues that may arise from time to time.

2. OBJECTIVE

The objective of this Consultancy (the "Objective") is to procure transaction related advice, develop a revenue model, undertake a financial appraisal and assist in the bidding process and award of the Project in a manner which ensures:

- (a) participation by the best available companies in the bidding process;
- (b) financing of the capital cost by the Concessionaire; and
- (c) optimising the revenue potential of the Project.

3. SCOPE OF SERVICES

- 3.1 The scope of services shall include:
 - 1. Scrutiny of various policies /guidelines issued by the Government of India / statutory authorities from time to time and suggests any action to be taken by Port.
 - 2. Preparation of Techno Economic Feasibility Report.
 - 3. Frame the various formats by incorporating information there be sent to concerned authority.
 - 4. To work out sustainable Royalty/Revenue Share for PPP Project.
 - 5. Preparation of Expression of Interest as MCA guidelines for PPP / Captive projects and its evaluation.
 - 6. As per the Model Documents of Ministry, preparing RFQ on the basis of Feasibility Report and updated figures, data and statistics, etc. scrutiny & evaluation thereof.
 - 7. Preparation for Pre-Bid and other meetings, conducting the same, preparing replies/clarifications in consultation with officials of Ports.
 - 8. As per Model Document of Ministry, preparing RFP, issuance, scrutiny and evaluation thereof.
 - 9. As per Model document of Ministry, preparing Draft Concession Agreement, finalize the same.
 - 10. Follow-up of compliance and scrutiny of conditions precedent and final closure.
 - 11. Preparing material for the Board Meeting for obtaining approvals at various stages.
 - 12. Preparation of letters, drafts, etc. for obtaining Govt. approvals including Security Clearance, etc.
 - 13. Co-ordination and correspondence with concerned statutory State/Central Govt. organizations/persons.
 - 14. Providing Transactional and Legal advice as required.
 - 15. Provide necessary technical, financial and legal assistance from time to time from Head Office.
 - 16. For appointment of IE Consultant /technical advisor for PPP projects, frame bidding documents & subsequent evaluation thereof.
 - 17. Furnish clarification to query from RFQ applicants/bidders, Ministry, TAMP or any other statutory Authority.

- 18. After signing of agreement with concessionaire, provide financial/legal opinion on implementation & operation of terminal.
- 19. To scrutinize security clearance information of bidders/RFQ Applicants.
- 20. To frame proposals like but not limited to SFC, EFC, PPPAC, CCEA etc. to take approval of competent authority.
- 21. To frame presentation to be made by Port before Authorities to finalization evaluation to seek approval of proposal
- 22. To attend the meeting and assist Port officials in respect to finalization, evaluation & approval of project.
- 23. To frame documents for appointing of Independent Engineer and evaluation thereof.
- 24. Any other matter related to procedural issues of PPP.

In making its projections, recommendations and Reports, the Consultant shall identify the underlying assumptions and reach an agreement with the Authority in relation thereto. The services to be rendered by the Transaction Advisor are briefly explained hereunder:

3.2 Transaction Adviser

The Consultant shall be responsible for review of the financial parameters and examination of the viability of the Project. The Consultant will also render advisory services for conducting the bid process and will prepare bidding documents and in conducting the bidding process for selection of the concessionaire for the project. The Consultant shall also maintain, update and disseminate the necessary data and information related to the Project and the bid process. During interaction with the bidders and stakeholders, the Consultant shall assist the Authority in responding to all queries satisfactorily and within the specified time. The Consultant shall render advisory services till the project achieves Financial Close.

3.3 Compilation and analysis of Data

The Consultant shall, based on available information, compile and analyse the financial and commercial data relating to the Project and prepare a revenue and expenditure statement, on commercial accounting principles, for three financial years preceding the consultancy. The expenditure statement shall include expenses on staff, material, contractual payments, etc.

3.4 Review of costs

Feasibility Report, along with applicable standards and the Concession Agreement will indicate the nature and extent of infrastructure, facilities and services to be provided by the Concessionaire. The Consultant shall review and comment on the cost estimates contained in the Feasibility Report. He shall ensure that appropriate provisions have been made for physical and price

contingencies, financing costs, interest during construction, etc. The Consultant shall also make a broad assessment of O&M expenses to be incurred by the Concessionaire during the entire Concession period based on applicable standards

3.5 Estimation of revenues

The Consultant shall evaluate the available data and information with a view to preparing a reasonable estimation of the likely revenues of the concessionaire.

3.6 Comment on draft Concession Agreement

The Consultant shall comment on the Model Concession Agreement (MCA) provided by the Authority and make its recommendations.

3.7 Impact of Project on Government Resources

The Consultant shall also identify and quantify the estimated financial impact of the Project on the resources of the Central / State Governments and the Project Authority.

3.8 Development of Revenue Model

The Consultant shall identify and quantify all costs, expenses and revenues of the Project, and shall prepare cash-flow statements for the concession period. Based on the above, the Consultant shall prepare the Revenue Model which will indicate the possible capital structure, likely sources of financing, the costs of financing, the cash flow, debt service, return on investment etc. (the "Revenue Model"). This would also include sensitivity analysis in relation to the critical parameters of the Revenue Model.

3.9 Project Appraisal

Based on the parameters specified in the Bid documents as well as the Revenue Model, the Consultant shall prepare an Appraisal Report for the Project outlining the salient features of the Project, its financial viability and its social and economic benefits. The Consultant shall work out the financial viability of the Project with a view to estimating the likely IRR over a concession period of 10 (ten) years, 15 (fifteen) years, 20 (twenty) years, 25 (twenty five) years and 30 (thirty) years. The Consultant shall review the Feasibility Report and the Model Concession Agreement to estimate the capital costs, O&M costs, revenues etc. and prepare a financial appraisal report for the Project (the "Appraisal Report).

The Authority shall invite Expression of Interest (EOI) from the Interested Parties for the Project. The Consultant shall prepare Expression of Interest (EOI) document and shall assist the Authority in evaluation of EOI applications.

3.10 Assistance in the RFQ process

The bidding process is specified in the Model Request for Qualification (RFQ) document. The Consultant shall assist in adapting the Model RFQ document for project-specific purposes. The Consultant shall also assist the Authority in the pre-qualification process. The Authority intends to pre-qualify bidders on the basis of the response to the RFQ document.

3.11 Preparation of Bid Documents

The Consultant shall prepare the Bid Documents i.e. the request for Qualification, the Request for Proposal and Draft Concession Agreement based on the Model RFQ, RFP and MCA. The Model Bid Documents shall be provided by the Authority. The Consultant may also suggest improvements in the development plan and project specific changes to be made in the Bid Documents. For this purpose, he shall work closely with the Authority and its associates / technical consultants. Bid documents would also include the draft Concession Agreement and the Feasibility Report. Bids to be submitted by the bidders shall be based on the aforesaid bid documents.

3.12 Assistance in the Bid Process

The Consultant shall assist the Authority in the bid process for selection of the Concessionaire from among the Bidders and till the project achieves Financial Close. This will primarily relate to participation in pre-bid meetings and answering questions or issuing clarifications with the approval of the Authority and evaluation of bids. The Consultant shall also assist the Authority in engaging with the bidders on different aspects of the Project such as its assets, the process of the transaction, the Revenue Model and the structure of the Project. It will also assist the Authority in preparing internal notes and projections for securing governmental approvals, if any.

Beyond the conclusion of bid process, these services may be in the nature of intermittent services required till the Completion of Services. Costs of Services till conclusion of bidding process are to be included in the Financial Proposal whereas payment for Services beyond this stage shall form part of Lump sum payment to be made as specified in Clause 10.2.

3.13 Assistance in selection of the preferred Bidder

The Authority intends to select the preferred bidder on the basis of the Proposals received from pre-qualified bidders. Request for Qualification and Financial Proposals will be invited as part of the Bidding Process. The Consultant shall assist the Authority in evaluating the proposals and in engaging with the selected bidder till the project achieves the Financial Close.

3.14 Rendering advisory services

The Consultant shall provide such other advice and assistance as may be necessary and incidental to the Services and as may be requested by the

Authority in respect of the Project, including but not limited to attending meetings, conferences and discussions with the Authority, and shall otherwise advise on and assist the Authority in the diverse commercial issues that may arise from time to time. The Consultant shall be responsible primarily for providing advice relating to legal, financial and project structuring issues arising from or during the course of the bidding process and the documents relating thereto.

3.15 Tax and insurance-related Matters

During the course of the Consultancy, the Consultant may be called upon to advise on tax and / or insurance related issues affecting the Project.

3.16 Specific Requirements for the Project

As specified in the clause No.3.1 of TOR

3.17 Scope not exhaustive

The Scope of Services specified in this Clause 3 are not exhaustive and the Consultant shall undertake such other tasks as may be necessary to appraise the project financially, prepare the Revenue Model and successfully complete the bid process for the Project.

4. DELIVERABLES

In pursuance of this TOR, the Consultant shall undertake/deliver the following deliverables (the "Deliverables") during the course of this Consultancy. Each deliverable shall include an executive summary, analyses, assumptions, results of computations, tables, charts, recommendations, and such other contents that generally comprise deliverables for similar consultancy work by way of best practices. 5 hard copies (draft reports), 15 hard copies in color (final reports) and 3 soft copies (editable format) of all Reports mentioned herein below shall be submitted to the Authority. The deliverables shall include:

A. Inception Report

On commencement of the Consultancy, the Consultant shall prepare and submit an Inception Report (the "Inception Report"). The Inception Report shall be a further elaboration of the Consultant's submissions towards understanding of the RFP, the methodology to be followed and Work Plan, etc. the Consultant shall submit a Conceptual Business Plan where it should clearly spell out the broad strategy for structuring the project and determining the viability gap funding requirement, if any, assuming an IRR of 12%.

- B. **Revenue Model** (Refer Clause 3.8)
- C. Appraisal Report (Refer Clause 3.9)

In case the viability gap funding projected in the Appraisal Report exceeds the projection in the Inception Report by more than 10% of the capital costs, a deduction of 5% (five per cent) of the Agreement Value shall be made from the payment due to the Consultant. Such deduction shall be deemed to be mutually agreed genuine pre-estimated compensation and damages suffered by the Authority on account of inaccurate projections leading to higher costs.

In the event that a viable project does not seem possible, the Consultant shall not proceed with the Consultancy and the same shall be terminated. Upon such termination, the Consultant shall be entitled to (a) full payment for the Deliverables completed by it; and (b) part payment for the non-completed deliverables on pro-rata basis as assessed by the Authority and the Authority's decision in this regard shall be final.

- D. Preparation of Tariff proposal to TAMP / Adjudicatory Board
- E. Preparation of RFQ, RFP and Concession Agreement (Refer Clauses 3.10 & 3.11).
- F. Submission for security clearance of bidders
- G. Submission of proposal to SFC/PPAC/MoS for approval and assisting the Authority during presentation.
- H. Assistance in Bidding Process till the project achieves financial close (Refer Clause 3.12)

5. TIME AND PAYMENT SCHEDULE

- 5.1 Subject to the provisions of Clause 10.1 of the TOR, the total duration for preparation of the Revenue Model, Appraisal Report and bid documents and the bidding process shall be 52 weeks, excluding the time taken by the Authority in providing the requisite documents or in conveying its comments on the Draft Reports. The Consultant shall deploy the Key Personnel as per the deployment of Personnel proposed. Intermittent services will be required beyond the conclusion of bid process and until the project achieves financial close or six months after the signing of the Concession Agreement, whichever is earlier.
- 5.2 The schedule for completing the Deliverables shall be determined by the maximum number of days/ weeks from the Effective Date of the Agreement (the "Key Dates" or "KD"). Time schedule for important Deliverables of the Consultancy and the payment schedule linked to the specified Deliverables are given below:

Key Dates	Description of Deliverable	Week No.*	Percentage of Payment
KD1	Submission of Inception Report	2	5

KD2	Submission of Feasibility Report and Revenue Model	8	. 5	
KD3	Submission of EOI document for approval and invitation of EOI	8 5		
KD 4	Evaluation of EOI applications and concluding EOI process	20 5		
KD 5	Submission of RFQ for approval and invitation of RFQ			
KD 6	Submission of a sustainable 22 10 Royalty/Revenue Share model for PPP Project		10	
KD 7			10	
KD 8	Submission of documents for Security clearance of bidders	ents for 37 10		
KD 9			10	
KD 10	Submission of appraisal report to PPPAC/SFC/MoS	41	5	
KD 11	Issue of RFP including preparation of Draft Concession Agreement	45	45 5	
KD 12	Conclusion of Bid Process and issue of LoA	49	49 . 5	
KD 13	Signing of Concession Agreement	52	· 10	
KD 14	Completion of Services till financial close	76 5		

^{*} Excluding time taken by Authority in providing comments on Draft Reports. Final Report to be submitted in a week's time after receipt of comments of the Authority.

Note:

The consultant may kindly note in case project being found non-feasible or no response for RFQ / RFP the respective item specified in clause 5.2 will not be operative. The consultant shall not claim the same.

In case for any reasons viz., after finalization and approval of the reports by the Authority, if the Authority requests the Consultant for repetition of any tasks mentioned above, the consultant shall do so and in such case, the Consultant shall be eligible for an additional payment of 50% of the original payment indicated for respective task.

- 5.3 Payment for the Services shall be based on the percentage specified in Clause 5.2.
- 5.4 The professional fees payable to the Consultant in accordance with the Financial Proposal shall cover the costs of telephone/fax, photocopying,

couriers and postage, collections and deliveries, traveling expense, stationery, costs of support staff and counsel fee, overheads, etc., including all taxes and duties except GST. No additional charges in respect thereof shall be due or payable.Payment for the Services shall be based on the percentage specified in Clause 5.2. Reimbursement of expenses relating to travel for participating in meetings and shall be made in accordance with the provisions of the Financial Proposal contained in Form–2 of Appendix-II of the RFP. Bills for such reimbursement may be submitted on a monthly basis.

- 5.5 Deleted
- 5.6 Deleted
- 5.7 Payment shall be made in respect of each Deliverable upon completion thereof.
 - 10% (twenty per cent) of the Agreement Value has been earmarked as Final Payment to be made to the Consultant upon execution of the Concession Agreement. In the event the Concession Agreement does not get executed within 18 (eighteen) months of the Effective Date, the Final Payment shall not become due to the Consultant, save and except the costs incurred for travel costs to attend meeting in the locations other than Mangalore, during the period from the Effective Date.
 - 5% each of the Agreement Value has been earmarked as payment to be made to the Consultant upon execution of the Concession Agreement and on achievement of financial close of the project.
- 5.8 In the event the Concession Agreement does not get executed within 52 weeks of the Effective Date, the Payment shall not become due to the Consultant, save and except the costs incurred for travel costs to attend meeting in the locations other than Mangalore, during the period from the Effective Date.
- In the event that the Consultancy is foreclosed, percentage payment due upto the completed stage will be made as indicated and in case assignment foreclose in the middle of any indicated stage, pro-rata payment shall be made for the completed services as assessed by the Authority and the Authority's decision in this regard shall be final.terminated by the Authority prior to its completion, the Consultant shall be entitled to (a) full payment for the Deliverables completed by it; and (b) part payment for the non-completed deliverables on pro-rata basis as assessed by the Authority and the Authority's decision in this regard shall be final.

6. MEETINGS

- 6.2 The Authority may review with the Consultant, any or all of the documents and advice forming part of the Consultancy, in meetings and conferences which will be held in Mangalore at the Authority's office. Further, the Consultant may be required to attend meetings and conferences with prequalified bidders / the Selected Bidder / Govt. agencies or as directed by the Authority from time to time. The expenses towards attending such meetings in places other than Mangalore shall be undertaken as per request of the Authority. For maximum two key personnel's or one key personnel and an Assistant shall be entitled to economic class air travel, board & lodging in a Three-star hotel. Bills for reimbursement hereunder may be submitted in original, along with supporting materials, Statement of Expenses, duly certified by the Authorised Representative.
- 6.2 The Authority may review with the Consultant, any or all of the documents and advice forming part of the Consultancy, in meetings and conferences which will be held in Mangalore at the Authority's office. Further, the Consultant may be required to attend meetings and conferences with prequalified bidders / the Selected Bidder / Govt. agencies or as directed by the Authority from time to time. The expenses towards attending such meetings in places other than Mangalore, during the period of Consultancy, including travel costs shall be reimbursed in accordance with the Financial Proposal contained in Form-2 of Appendix-II of the RFP.
- 6.3 The Authority may, in its discretion, require the Consultant to participate in extended meetings and/ or work from the offices of the Authority and the Consultant shall, on a best endeavor basis and without unreasonable delay, provide such services at the offices of the Authority.

7. CONSULTANCY TEAM

7.1 The Consultant shall form a team (the "Consultancy Team") for undertaking this assignment. The Consultancy Team shall consist of experts who have the requisite qualifications and experience. PPP Expert shall be the leader of the Consultancy Team. The following Key Personnel whose experience and responsibilities are briefly described herein would be considered for evaluation of the Technical Proposal. Other expertise as required for the services to be rendered by the Consultant shall be included in the Team either through the Key Personnel specified below or through other Professional Personnel, as necessary.

Key Personnel:

PPP Expert-cum-Team Leader (the "PPP Expert")

TIT Expert-cum-ream Lea	
Educational Qualifications	Chartered Accountant / MBA (Finance) /ICWA / CFA or equivalent
	15 years.
Essential Experience	
	He should have led the financial team in
	Preparation of revenue Model, Bid Documents for
	not less than 5 (five) Public Private Partnership
	(PPP) projects, out of which, at least 1 (one)
	should be a Port Sector PPP project.
Job Responsibilities	He will lead, co-ordinate and supervise the
	Consultancy Team for delivering the Consultancy
	in a timely manner as envisaged in this RFP. He
	shall be responsible for preparation of Revenue
	Model and appropriate structure for the Project,
	prepare bid documents for selection of the
	Concessionaire in line with the principles of PPP /
	Captive Policy guidelines. He shall not delegate
	his responsibilities except with the prior written
	· · · · · · · · · · · · · · · · · · ·
	approval of the Authority
Minimum time required on	Fifteen working days in a month till completion of
site	the services.

Port Sector Expert

T OIT Dector Expert	
Educational Qualifications	Master/ Bachelor in Engineering or equivalent
Essential Experience	10 years He should have worked as a sector expert in not less than 3 (three) Port Sector projects preferably Bulk Liquid handling facilities.
Job Responsibilities	
Minimum time required on site	He will provide the sectoral perspective and shall be responsible for preparation of Feasibility Report for the project including cost estimates and traffic projections and the Draft Concession Agreement. He shall assist the Team Leader in the preparation of the Revenue Model and preparation of bid documents for selection of the Concessionaire. Seven working days in a month till completion of
	the services.

Legal Consultant

Educational Qualifications	Master / Bachelor in Law or equivalent
Essential Experience	5 years He should have worked as a Legal Consultant for issues relating to commercial contracts in not less than 3 (three) Port Sector projects.
Job Responsibilities	He will provide the Legal Perspective and will review the Model Bid Documents including Model Concession Agreement to affect project specific changes required from legal perspective.
Minimum time required on Site	Seven working days in a month till completion of the services.

Associate Consultant

	Master / Bachelor degree in Engineering or	
Educational	Finance or equivalent	
Qualifications		
Essential Experience	He should have worked as member of a	
	Transaction Advisory team in preparation of	
	Revenue Model and/or Bid Documents for at least	
	2 (two) Port Sector projects.	
	He will assist the Team Leader / PPP Expert and	
	Port Sector Expert by providing inputs for	
	preparation of Revenue Model, project structure	
Job Responsibilities	and preparation of bid documents.	

7.2 The Consultant shall mobilise and demobilise its Professional Personnel and Support Personnel as required.

8. REPORTING

- 8.1Deleted
- 8.2The Consultant will work closely with the Authority and its technical and legal consultants. The Authority has established a Working Group (the "WG") to enable conduct of this assignment. A designated nodal officer of the Authority will be responsible for the overall coordination and project development. He will play a coordinating role in dissemination of the Consultant's outputs, facilitating discussions, and ensuring required reactions and responses to the Consultant.
- 8.3The Consultant may prepare Issue Papers highlighting issues that could become critical for the timely completion of the Project and that require

- 8.4The Consultant will make a presentation on the inception report for discussion with the WG at a meeting. This will be a working document. The Consultant is required to prepare and submit a monthly report that includes and describes, inter alia, general progress to date; data and reports obtained and reviewed, conclusions to date, if any; concerns about availability of, or access to, data, analyses, reports; questions regarding the TOR or any other matters regarding work scope and related issues; and so on. The Consultants' work on the TOR tasks should continue while the report is under consideration and is being discussed.
- 8.5 Regular communication with the WG and the Nodal officer is required in addition to all key communications. This may take the form of telephone/ teleconferencing, emails, faxes, and occasional meetings.
- 8.6The Deliverables will be submitted as per schedule provided in this RFP.

9. DOCUMENTS TO BE MADE AVAILABLE BY THE AUTHORITY

Available data as may be required by the Consultant will be provided by the Authority on request. The Nodal Officer designated by the Authority shall facilitate handing over of such information to the Consultant.

The Authority shall provide to the Consultant the following:

- (a) Feasibility Report and Detailed Project Report for the project of Construction of Berth No. 13;
- (b) Construction Standards, Operations & Maintenance Standards, Safety Standards and Performance Standards for Port Projects through Public Private Partnership.; and
- (c) A copy of the Model Bid Documents including Concession Agreement. Available data as may be required by the Consultant will be provided by the Authority on request. The Nodal Officer designated by the Authority shall facilitate handing over of such information to the Consultant.

10. COMPLETION OF SERVICES

10.1 All the Deliverables shall be compiled, classified and submitted by the Consultant to the Authority in soft form, to the extent possible. The documents comprising the Deliverables shall remain the property of the Authority and shall not be used by the Consultant for any purpose other than that intended under these Terms of Reference without the permission of the Authority. The Consultancy shall stand completed on acceptance by the Authority of all the Deliverables of the Consultant. Unless completed earlier, the Services shall

be deemed completed and finally accepted by the Authority and the final Deliverable shall be deemed approved by the Authority as satisfactory upon expiry of 60 (sixty) days after receipt of the final Deliverable unless the Authority, within such 60 (sixty) day period, gives written notice to the Consultant specifying in detail, the deficiencies in the Services. The Consultant shall thereupon promptly make any necessary corrections and/or additions, and upon completion of such corrections or additions, the foregoing process shall be repeated. The Consultancy shall in any case be deemed to be completed upon expiry of 18 (eighteen) months or 76 (seventy six) weeks from the Effective Date, unless extended by mutual consent of the Authority and the Consultant.

10.2 Deleted.

SCHEDULE – 2 (See Clause 2.1.3)

AGREEMENT

FOR

APPOINTMENT OF TRANSACTION ADVISOR FOR CONCESSIONING OF BERTH NO. 13 FOR HANDLING LIQUID BULK / LPG / POL PRODUCTS/ CHEMICALS ON CAPTIVE / PPP MODE

CONTENTS Page no.

1. General

- 1.1 Definitions and Interpretation
- 1.2 Relation between the Parties
- 1.3 Rights and obligations
- 1.4 Governing law and jurisdiction
- 1.5 Language
- 1.6 Table of contents and headings
- 1.7 Notices
- 1.8 Location
- 1.9 Authority of Member-in-charge
- 1.10 Authorised representatives
- 1.11 Taxes and duties

2. Commencement, Completion and Termination of Agreement

- 2.1 Effectiveness of Agreement
- 2.2 Commencement of Services
- 2.3 Termination of Agreement for failure to commence Services
- 2.4 Expiration of Agreement
- 2.5 Entire Agreement
- 2.6 Modification of Agreement
- 2.7 Force Majeure
- 2.8 Suspension of Agreement
- 2.9 Termination of Agreement

3. Obligations of the Consultant

- 3.1 General
- 3.2 Conflict of Interest
- 3.3 Confidentiality
- 3.4 Liability of the Consultant
- 3.5 Accounting, inspection and auditing
- 3.6 Consultant's actions requiring the Authority's prior approval
- 3.7 Reporting obligations
- 3.8 Documents prepared by the Consultant to be the property of the Authority
- 3.9 Materials furnished by the Authority
- 3.10 Providing access to the Project Office and Personnel

3.11 Accuracy of Documents

- 4. Consultant's Personnel and Sub-Consultant
 - 4.1 General
 - 4.2 Deployment of Personnel
 - 4.3 Approval of Personnel
 - 4.4 Substitution of Key Personnel
 - 4.5 Working hours, overtime, leave etc.
 - 4.6 Financial Expert and Project Manager
 - 4.7. Sub-Consultants
- 5. Obligations of the Authority
 - 5.1 Assistance in clearances etc.
 - 5.2 Access to land and property
 - 5.3 Change in Applicable Law
 - 5.3 Payment
- 6. Payment to the Consultant
 - 6.1 Cost estimates and Agreement Value
 - 6.2 Currency of payment
 - 6.3 Mode of billing and payment
- 7. Performance Security, Liquidated damages and penalty
 - 7.1 Performance Security
 - 7.2 Liquidated Damages
 - 7.3 Penalty for deficiency in services
- 8. Fairness and Good Faith
 - 8.1 Good Faith
 - 8.2 Operation of the Agreement
- 9. Settlement of Disputes
 - 9.1 Amicable settlement
 - 9.2 Dispute resolution
 - 9.3 Conciliation
 - 9.4 Arbitration

ANNEXES

- Annex 1: Terms of Reference
- Annex 2: Deployment of Key Personnel
- Annex 3: Estimate of Personnel Costs
- Annex 4: Approved Sub-consultant(s)
- Annex 5: Cost of Services
- Annex 6: Payment Schedule
- Annex 7: Bank Guarantee for Performance Security

AGREEMENT

Appointment of Transaction Advisor for Concessioning of Berth No.13 for handling Liquid Bulk / LPG / POL products / Chemicals at New Mangalore Port on Captive / PPP Mode

	EMENT No This AGREEMENT (hereinafter called the ement") is made on the
"Autho	day of the month of 2022, between, on the one hand, PV/ New Mangalore Port Trust (NMPA) (hereinafter called the prity" which expression shall include their respective successors and ted assigns, unless the context otherwise requires) and, on the other
expres assign	(hereinafter called the "Consultant" which ssion shall include their respective successors and permitted is).
WHEF	REAS
Advise No.13	uthority vide its Request for Proposal for appointment of Transaction or (hereinafter called the "Consultancy") for Concessioning of Berth for handling Liquid Bulk / LPG / POL products / Chemicals at Mangalore Port on Captive / PPP Mode (hereinafter called the ct");
(A)	the Consultant submitted its proposals for the aforesaid work, whereby the Consultant represented to the Authority that it had the required professional skills, and in the said proposals the Consultant also agreed to provide the Services to the Authority on the terms and conditions as set forth in the RFP and this Agreement; and
(B)	the Authority, on acceptance of the aforesaid proposals of the Consultant, awarded the Consultancy to the Consultant vide its Letter of Award No dated (the "LOA"); and
(C)	in pursuance of the LOA, the parties have agreed to enter into this Agreement.
NOW	THEREFORE, the parties hereto hereby agree as follows:
,	There is one, the parties herete hereby agree as renewe.

1. GENERAL

1.1 Definitions and Interpretation

- 1.1.1 The words and expressions beginning with capital letters and defined in this Agreement shall, unless the context otherwise requires, have the meaning hereinafter respectively assigned to them:
- (a) "Additional Costs" shall have the meaning set forth in Clause 6.1.2;
- (b)(a) "Agreement" means this Agreement, together with all the Annexes;
- (c)(b) "Agreement Value" shall have the meaning set forth in Clause 6.1.2;
- (d)(c) "Applicable Laws" means the laws and any other instruments having the force of law in India as they may be issued and in force from time to time;
- (e)(d) "Confidential Information" shall have the meaning set forth in Clause 3.3;
- (f)(e) "Conflict of Interest" shall have the meaning set forth in Clause 3.2 read with the provisions of RFP;
- (g)(f) "Dispute" shall have the meaning set forth in Clause 9.2.1;
- (h)(g) "Effective Date" means the date on which this Agreement comes into force and effect pursuant to Clause 2.1;
- (i)(h) "Expatriate Personnel" means such persons who at the time of being so hired had their domicile outside India;
- (j)(i) "Government" means the Government of India;
- (k)(j) "INR, Re. or Rs." means Indian Rupees;
- (I)(k) "Member", in case the Consultant consists of a consortium of more than one entity, means any of these entities, and "Members" means all of these entities:
- (m) "Party" means the Authority or the Consultant, as the case may be, and Parties means both of them;
- (n) "Personnel" means persons hired by the Consultant or by any Sub-Consultant as employees or retainers and assigned to the performance of the Services or any part thereof;
- (o) "Resident Personnel" means such persons who at the time of being

so hired had their domicile inside India;

- (p) "RFP" means the Request for Proposal document in response to which the Consultant's proposal for providing Services was accepted;
- (q) "Services" means the work to be performed by the Consultant pursuant to this Agreement, as described in the Terms of Reference hereto;
- (r) "Sub-Consultant" means any entity to which the Consultant subcontracts any part of the Services in accordance with the provisions of Clause 4.7; and
- (r) "Third Party" means any person or entity other than the Government, the Authority, the Consultant or a Sub-Consultant.
 - All terms and words not defined herein shall, unless the context otherwise requires, have the meaning assigned to them in the RFP.
- 1.1.2 The following documents along with all addenda issued thereto shall be deemed to form and be read and construed as integral part of this Agreement and in case of any contradiction between or among them the priority in which a document would prevail over another would be as laid down below beginning from the highest priority to the lowest priority:
 - (a) Agreement;
 - (b) Annexes of Agreement;
 - (c) RFP and Corrigendum's if any; and
 - (d) Letter of Award

1.2 Relation between the Parties

Nothing contained herein shall be construed as establishing a relation of master and servant or of agent and principal as between the Authority and the Consultant. The Consultant shall, subject to this Agreement, have complete charge of Personnel performing the Services and shall be fully responsible for the Services performed by them or on their behalf hereunder.

1.3 Rights and obligations

The mutual rights and obligations of the Authority and the Consultant shall be as set forth in the Agreement, in particular:

- (a) the Consultant shall carry out the Services in accordance with the provisions of the Agreement; and
- (b) the Authority shall make payments to the Consultant in accordance with the provisions of the Agreement.

1.4 Governing law and jurisdiction

This Agreement shall be construed and interpreted in accordance with and governed by the laws of India, and the courts at Mangalore shall have exclusive jurisdiction over matters arising out of or relating to this Agreement.

1.5 Language

All notices required to be given by one Party to the other Party and all other communications, documentation and proceedings which are in any way relevant to this Agreement shall be in writing and in English language.

1.6 Table of contents and headings

The tables of contents, headings or sub-headings in this Agreement are for convenience of reference only and shall not be used in, and shall not affect, the construction or interpretation of this Agreement.

1.7 Notices

Any notice or other communication to be given by any Party to the other Party under or in connection with the matters contemplated by this Agreement shall be in writing and shall:

- (a) in the case of the Consultant, be given by facsimile or e-mail and by letter delivered by hand to the address given and marked for attention of the Consultant's Representative set out below in Clause 1.10 or to such other person as the Consultant may from time to time designate by notice to the Authority; provided that notices or other communications to be given to an address outside may, if they are subsequently confirmed by sending a copy thereof by registered acknowledgement due, air mail or by courier, be sent by facsimile or e-mail to the number/address as the Consultant may from time to time specify by notice to the Authority;
- (b) in the case of the Authority, be given by facsimile or e-mail and by letter delivered by hand and be addressed to the

Authority with a copy delivered to the Authority Representative set out below in Clause 1.10 or to such other person as the Authority may from time to time designate by notice to the Consultant; provided that if the Consultant does not have an office in Mangalore it may send such notice by facsimile or email and by registered acknowledgement due, air mail or by courier; and

(c) any notice or communication by a Party to the other Party, given in accordance herewith, shall be deemed to have been delivered when in the normal course of post it ought to have been delivered and in all other cases, it shall be deemed to have been delivered on the actual date and time of delivery; provided that in the case of facsimile or email, it shall be deemed to have been delivered on the working days following the date of its delivery.

1.8 Location

- 1.8.1 The Services shall be performed at the offices of the Authority Consultant in accordance with the provisions of RFP and at such locations as are incidental thereto, including the offices of the ConsultantAuthority.
- 1.8.2 The Authority may require the PPP Expert to spend the required man hours at the offices of the Authority and the Consultant agrees and undertakes to provide such services on a best effort basis and without any unreasonable delay.

1.9 Authority of Member-in-charge

In case the Consultant consists of a consortium of more than one entity, the Parties agree that the Lead Member shall act on behalf of the Members in exercising all the Consultant's rights and obligations towards the Authority under this Agreement, including without limitation the receiving of instructions and payments from the Authority.

1.10 Authorised Representatives

1.10.1Any action required or permitted to be taken, and any document required or permitted to be executed, under this Agreement by the Authority or the Consultant, as the case may be, may be taken or executed by the officials specified in this Clause 1.10.

1.10.2 The Authority may, from time to time, designate one of its officials as the Authority Representative. Unless otherwise notified, the Authority Representative shall be:

Chief Engineer (Civil)

Civil Engineering Department

New Mangalore Port Authority

Panambur,

Mangalore - 575 010

Kanrnataka (India)

Phone: 0824 2407493 Fax: 0824 2407493

Email: chiefengineer@nmpt.gov.in

1.10.31.10.2 The Consultant may designate one of its employees as Consultant's Representative. Unless otherwise notified, the Consultant's Representative shall be:

Tel:
Mobile:
Fax:
F-mail [.]

1.11 Taxes and duties

Unless otherwise specified in the Agreement, the Consultant shall pay all such taxes, duties, fees and other impositions as may be levied under the Applicable Laws and the Authority shall perform such duties in regard to the deduction of such taxes as may be lawfully imposed on it.

COMMENCEMENT, COMPLETION AND TERMINATION OF AGREEMENT

2.1 Effectiveness of Agreement

This Agreement shall come into force and effect on the date of this Agreement (the "Effective Date").

2.2 Commencement of Services

The Consultant shall commence the Services within a period of 7 (seven) days from the Effective Date, unless otherwise agreed by the Parties.

2.3 Termination of Agreement for failure to commence Services

If the Consultant does not commence the Services within the period specified in Clause 2.2 above, the Authority may, by not less than 1 (one) weeks' notice to the Consultant, declare this Agreement to be null and void, and in the event of such a declaration, this Agreement shall stand terminated and the Consultant shall be deemed to have accepted such termination.

2.4 Expiration of Agreement

Unless terminated earlier pursuant to Clauses 2.3 or 2.9 hereof, this Agreement shall, unless extended by the Parties by mutual consent, expire upon the earlier of (i) expiry of a period of 60 (sixty) days after the delivery of the final Deliverable to the Authority; and (ii) the expiry of 1 (one) year from the Effective Date. Upon Termination, the Authority shall make payments of all amounts due to the Consultant hereunder.

2.5.1 Entire Agreement

- 2.5.2 This Agreement and the Annexes together constitute a complete and exclusive statement of the terms of the agreement between the Parties on the subject hereof, and no amendment or modification hereto shall be valid and effective unless such modification or amendment is agreed to in writing by the Parties and duly executed by persons especially empowered in this behalf by the respective Parties. All prior written or oral understandings, offers or other communications of every kind pertaining to this Agreement are abrogated and withdrawn; provided, however, that the obligations of the Consultant arising out of the provisions of the RFP shall continue to subsist and shall be deemed to form part of this Agreement.
- 2.5.3 Without prejudice to the generality of the provisions of Clause 2.5.1, on matters not covered by this Agreement, the provisions of RFP shall apply.

2.6 Modification of Agreement

Modification of the terms and conditions of this Agreement, including any modification of the scope of the Services, may only be made by written agreement between the Parties. Pursuant to Clauses 4.2.3 and 6.1.3 hereof, however, each Party shall give due consideration to any proposals for modification made by the other Party.

2.7 Force Majeure

2.7.1 Definition

- (a) For the purposes of this Agreement, "Force Majeure" means an event which is beyond the reasonable control of a Party, and which makes a Party's performance of its obligations hereunder impossible or so impractical as reasonably to be considered impossible in the circumstances, and includes, but is not limited to, war, riots, civil disorder, earthquake, fire, explosion, storm, flood or other adverse weather conditions, strikes, lockouts or other industrial action (except where such strikes, lockouts or other industrial action are within the power of the Party invoking Force Majeure to prevent), confiscation or any other action by government agencies.
- (b) Force Majeure shall not include (i) any event which is caused by the negligence or intentional action of a Party or such Party's Sub-Consultant or agents or employees, nor (ii) any event which a diligent Party could reasonably have been expected to both (A) take into account at the time of the conclusion of this Agreement, and (B) avoid or overcome in the carrying out of its obligations hereunder.
- (c) Force Majeure shall not include insufficiency of funds or failure to make any payment required hereunder.

2.7.2 No breach of Agreement

The failure of a Party to fulfill any of its obligations hereunder shall not be considered to be a breach of, or default under, this Agreement insofar as such inability arises from an event of Force Majeure, provided that the Party affected by such an event has taken all reasonable precautions, due care and reasonable alternative measures, all with the objective of carrying out the terms and conditions of this Agreement.

2.7.3 Measures to be taken

- (a) A Party affected by an event of Force Majeure shall take all reasonable measures to remove such Party's inability to fulfil its obligations hereunder with a minimum of delay.
- (b) A Party affected by an event of Force Majeure shall notify the other Party of such event as soon as possible, and in any

event not later than 14 (fourteen) days following the occurrence of such event, providing evidence of the nature and cause of such event, and shall similarly give notice of the restoration of normal conditions as soon as possible.

(c) The Parties shall take all reasonable measures to minimise the consequences of any event of Force Majeure.

2.7.4 Extension of time

Any period within which a Party shall, pursuant to this Agreement, complete any action or task, shall be extended for a period equal to the time during which such Party was unable to perform such action as a result of Force Majeure.

2.7.5 Payments

The payment shall be made to the Consultant as specified in the Clause 5.0 of Terms of Reference.

During the period of its inability to perform the Services as a result of an event of Force Majeure, the Consultant shall be entitled to be reimbursed for additional costs reasonably and necessarily incurred by it during such period for the purposes of the Services and in reactivating the Services after the end of such period.

2.7.6 Consultation

Not later than 30 (thirty) days after the Consultant has, as the result of an event of Force Majeure, become unable to perform a material portion of the Services, the Parties shall consult with each other with a view to agreeing on appropriate measures to be taken in the circumstances.

2.8 Suspension of Agreement

The Authority may, by written notice of suspension to the Consultant, suspend all payments to the Consultant hereunder if the Consultant shall be in breach of this Agreement or shall fail to perform any of its obligations under this Agreement, including the carrying out of the Services; provided that such notice of suspension (i) shall specify the nature of the breach or failure, and (ii) shall provide an opportunity to the Consultant to remedy such breach or failure within a period not exceeding 15 (fifteen) days after receipt by the Consultant of such notice of suspension.

2.9 Termination of Agreement

2.9.1 By the Authority

The Authority may, by not less than 15 (fifteen) days' written notice of termination to the Consultant, such notice to be given after the occurrence of any of the events specified in this Clause 2.9.1, terminate this Agreement if:

- (a) the Consultant fails to remedy any breach hereof or any failure in the performance of its obligations hereunder, as specified in a notice of suspension pursuant to Clause 2.8 hereinabove, within 15 (fifteen) days of receipt of such notice of suspension or within such further period as the Authority may have subsequently granted in writing;
- (b) the Consultant becomes insolvent or bankrupt or enters into any agreement with its creditors for relief of debt or take advantage of any law for the benefit of debtors or goes into liquidation or receivership whether compulsory or voluntary;
- (c) the Consultant fails to comply with any final decision reached as a result of arbitration proceedings pursuant to Clause 9 hereof;
- (d) the Consultant submits to the Authority a statement which has a material effect on the rights, obligations or interests of the Authority and which the Consultant knows to be false;
- (e) any document, information, data or statement submitted by the Consultant in its Proposals, based on which the Consultant was considered eligible or successful, is found to be false, incorrect or misleading;
- (f) as the result of Force Majeure, the Consultant is unable to perform a material portion of the Services for a period of not less than 60 (sixty) days; or
- (g) the Authority, in its sole discretion and for any reason whatsoever, decides to terminate this Agreement.

2.9.2 By the Consultant

The Consultant may, by not less than 30 (thirty) days' written notice to the Authority, such notice to be given after the occurrence of any of the events specified in this Clause 2.9.2, terminate this Agreement if:

- (a) the Authority fails to pay any money due to the Consultant pursuant to this Agreement and not subject to dispute pursuant to Clause 9 hereof within 45 (forty five) days after receiving written notice from the Consultant that such payment is overdue:
- (b) the Authority is in material breach of its obligations pursuant to this Agreement and has not remedied the same within 45 (fortyfive) days (or such longer period as the Consultant may have subsequently granted in writing) following the receipt by the Authority of the Consultant's notice specifying such breach;
- (c) as the result of Force Majeure, the Consultant is unable to perform a material portion of the Services for a period of not less than 60 (sixty) days; or
- (d) the Authority fails to comply with any final decision reached as a result of arbitration pursuant to Clause 9 hereof.

2.9.3 Cessation of rights and obligations

Upon termination of this Agreement pursuant to Clauses 2.3 or 2.9 hereof, or upon expiration of this Agreement pursuant to Clause 2.4 hereof, all rights and obligations of the Parties hereunder shall cease, except (i) such rights and obligations as may have accrued on the date of termination or expiration, or which expressly survives such Termination; (ii) the obligation of confidentiality set forth in Clause 3.3 hereof; (iii) the Consultant's obligation to permit inspection, copying and auditing of such of its accounts and records set forth in Clause 3.5, as relate to the Consultant's Services provided under this Agreement, and (iv) any right or remedy which a Party may have under this Agreement or the Applicable Laws.

2.9.4 Cessation of Services

Upon termination of this Agreement by notice of either Party to the other pursuant to Clauses 2.9.1 or 2.9.2 hereof, the Consultant shall, immediately upon dispatch or receipt of such notice, take all necessary steps to bring the Services to a close in a prompt and orderly manner and shall make every reasonable effort to keep expenditures for this purpose to a minimum. With respect to documents prepared by the Consultant and materials furnished by the Authority, the Consultant shall proceed as provided respectively

by Clauses 3.8 or 3.9 hereof.

2.9.5 Payment upon Termination

Upon termination of this Agreement pursuant to Clauses 2.9.1 or 2.9.2 hereof, the Authority shall make the following payments to the Consultant (after offsetting against these payments any amount that may be due from the Consultant to the Authority):

- (i) Deleted
- reimbursable expenditures pursuant to Clause 6 hereof for expenditures actually incurred prior to the date of termination; and
- (iii) except in the case of termination pursuant to Sub-clauses (a) through (e) of Clause 2.9.1 hereof, reimbursement of any reasonable cost incidental to the prompt and orderly termination of the Agreement including the cost of the return travel of the Consultant's personnel.

2.9.6 Disputes about Events of Termination

If either Party disputes whether an event specified in Clause 2.9.1 or in Clause 2.9.2 hereof has occurred, such Party may, within 30 (thirty) days after receipt of notice of termination from the other Party, refer the matter to arbitration pursuant to Clause 9 hereof, and this Agreement shall not be terminated on account of such event except in accordance with the terms of any resulting arbitral award.

3. OBLIGATIONS OF THE CONSULTANT

3.1 General

3.1.1 Standards of Performance

The Consultant shall perform the Services and carry out its obligations hereunder with all due diligence, efficiency and economy, in accordance with generally accepted professional techniques and practices, and shall observe sound management practices, and employ appropriate advanced technology and safe and effective equipment, machinery, materials and methods. The Consultant shall always act, in respect of any matter relating to this Agreement or to the Services, as a faithful adviser to the Authority, and shall at all times support and safeguard the Authority's legitimate interests in any dealings with Sub-consultants or Third Parties.

3.1.2 Terms of Reference

The scope of Services to be performed by the Consultant is specified in the Terms of Reference (the "TOR") at Schedule-1 of this Agreement. The Consultant shall provide the Deliverables specified therein in conformity with the time schedule stated therein.

3.1.3 Applicable Laws

The Consultant shall perform the Services in accordance with the Applicable Laws and shall take all practicable steps to ensure that any Sub-Consultant, as well as the Personnel and agents of the Consultant and any Sub-Consultant, comply with the Applicable Laws.

3.2 Conflict of Interest

- 3.2.1 The Consultant shall not have a Conflict of Interest and any breach hereof shall constitute a breach of the Agreement.
- 3.2.2 Consultant and Affiliates not to be otherwise interested in the Project

The Consultant agrees that, during the term of this Agreement and after its termination, the Consultant or any Associate thereof, as well as any Sub-Consultant and any entity affiliated with such Sub-Consultant, shall be disqualified from providing goods, works, services, loans or equity for any project resulting from or closely related to the Services and any breach of this obligation shall amount to a Conflict of Interest; provided that the restriction herein shall not apply after a period of three years from the completion of this assignment or to consulting assignments granted by banks/ lenders at any time; provided further that this restriction shall not apply to consultancy/ advisory services provided to the Authority in continuation of this Consultancy or to any subsequent consultancy/ advisory services provided to the Authority in accordance with the rules of the Authority. For the avoidance of doubt, an entity affiliated with the Consultant shall include a partner in the firm of the Consultant or a person who holds more than 5% (five per cent) of the subscribed and paid up share capital of the Consultant, as the case may be, and any Associate thereof.

3.2.3 Prohibition of conflicting activities

Neither the Consultant nor its Sub-consultant nor the Personnel of

either of them shall engage, either directly or indirectly, in any of the following activities:

- (a) during the term of this Agreement, any business or professional activities which would conflict with the activities assigned to them under this Agreement;
- (b) after the termination of this Agreement, such other activities as may be specified in the Agreement; or
- (c) at any time, such other activities as have been specified in the RFP as Conflict of Interest.

3.2.4 Consultant not to benefit from commissions, discounts, etc.

The remuneration of the Consultant pursuant to Clause 6 hereof shall constitute the Consultant's sole remuneration in connection with this Agreement or the Services and the Consultant shall not accept for its own benefit any trade commission, discount or similar payment in connection with activities pursuant to this Agreement or to the Services or in the discharge of its obligations hereunder, and the Consultant shall use its best efforts to ensure that any Sub-Consultant, as well as the Personnel and agents of either of them, similarly shall not receive any such additional remuneration.

- 3.2.5 The Consultant and its Personnel shall observe the highest standards of ethics and not have engaged in and shall not hereafter engage in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice (collectively the "Prohibited Practices"). Notwithstanding anything to the contrary contained in this Agreement, the Authority shall be entitled to terminate this Agreement forthwith by a communication in writing to the Consultant, without being liable in any manner whatsoever to the Consultant, if it determines that the Consultant has, directly or indirectly or through an agent, engaged in any Prohibited Practices in the Selection Process or before or after entering into of this Agreement. In such an event, the Authority shall forfeit and appropriate the Performance Security, if any, as mutually agreed genuine pre-estimated compensation and damages payable to the Authority towards, inter alia, time, cost and effort of the Authority, without prejudice to the Authority's any other rights or remedy hereunder or in law.
- 3.2.6 Without prejudice to the rights of the Authority under Clause 3.2.5 above and the other rights and remedies which the Authority may

have under this Agreement, if the Consultant is found by the Authority to have directly or indirectly or through an agent, engaged or indulged in any Prohibited Practices, during the Selection Process or before or after the execution of this Agreement, the Consultant shall not be eligible to participate in any tender or RFP issued during a period of 2 (two) years from the date the Consultant is found by the Authority to have directly or indirectly or through an agent, engaged or indulged in any Prohibited Practices.

- 3.2.7 For the purposes of Clauses 3.2.5 and 3.2.6, the following terms shall have the meaning hereinafter respectively assigned to them:
 - "corrupt practice" means the offering, giving, receiving or (a) soliciting, directly or indirectly, of anything of value to influence the actions of any person connected with the Selection Process (for removal of doubt, offering of employment or employing or engaging in any manner whatsoever, directly or indirectly, any official of the Authority who is or has been associated in any manner, directly or indirectly with Selection Process or LOA or dealing with matters concerning the Agreement before or after the execution thereof, at any time prior to the expiry of one year from the date such official resigns or retires from or otherwise ceases to be in the service of the Authority, shall be deemed to constitute influencing the actions of a person connected with the Selection Process); or (ii) engaging in any manner whatsoever, whether during the Selection Process or after the issue of LOA or after the execution of the Agreement, as the case may be, any person in respect of any matter relating to the Project or the LOA or the Agreement, who at any time has been or is a legal, financial or technical adviser the Authority in relation to any matter concerning the Project;
 - (b) "fraudulent practice" means a misrepresentation or omission of facts or suppression of facts or disclosure of incomplete facts, in order to influence the Selection Process;
 - (c) "coercive practice" means impairing or harming, or threatening to impair or harm, directly or indirectly, any person or property to influence any person's participation or action in the Selection Process or the exercise of its rights or performance of its obligations by the Authority under this Agreement;

- (d) "undesirable practice" means (i) establishing contact with any person connected with or employed or engaged by the Authority with the objective of canvassing, lobbying or in any manner influencing or attempting to influence the Selection Process; or (ii) having a Conflict of Interest; and
- (e) "restrictive practice" means forming a cartel or arriving at any understanding or arrangement among Applicants with the objective of restricting or manipulating a full and fair competition in the Selection Process.

3.3 Confidentiality

The Consultant, its Sub-Consultants and the Personnel of either of them shall not, either during the term or within two years after the expiration or termination of this Agreement disclose any proprietary information, including information relating to reports, data, drawings, design software or other material, whether written or oral, in electronic or magnetic format, and the contents thereof; and any reports, digests or summaries created or derived from any of the foregoing that is provided by the Authority to the Consultant, its Sub-Consultants and the Personnel; any information provided by or relating to the Authority, its technology, technical processes, business affairs or finances or any information relating to the Authority's employees, officers or other professionals or suppliers, customers, or contractors of the Authority; and any other information which the Consultant is under an obligation to keep confidential in relation to the Project, the Services or this Agreement ("Confidential Information"), without the prior written consent of the Authority.

Notwithstanding the aforesaid, the Consultant, its Sub-Consultants and the Personnel of either of them may disclose Confidential Information to the extent that such Confidential Information:

- (i) was in the public domain prior to its delivery to the Consultant, its Sub-Consultants and the Personnel of either of them or becomes a part of the public knowledge from a source other than the Consultant, its Sub-Consultants and the Personnel of either of them;
- (ii) was obtained from a third party with no known duty to maintain its confidentiality;
- (iii) is required to be disclosed by Applicable Laws or judicial or administrative or arbitral process or by any governmental

instrumentalities, provided that for any such disclosure, the Consultant, its Sub-Consultants and the Personnel of either of them shall give the Authority, prompt written notice, and use reasonable efforts to ensure that such disclosure is accorded confidential treatment; and

(iv) is provided to the professional advisers, agents, auditors or representatives of the Consultant or its Sub-Consultants or Personnel of either of them, as is reasonable under the circumstances; provided, however, that the Consultant or its Sub-Consultants or Personnel of either of them, as the case may be, shall require their professional advisers, agents, auditors or its representatives, to undertake in writing to keep such Confidential Information, confidential and shall use its best efforts to ensure compliance with such undertaking.

3.4 Liability of the Consultant

- 3.4.1 The Consultant's liability under this Agreement shall be determined by the Applicable Laws and the provisions hereof.
- 3.4.2 The Consultant shall, subject to the limitation specified in Clause 3.4.3, be liable to the Authority for any direct loss or damage accrued or likely to accrue due to deficiency in Services rendered by it.
- 3.4.3 The Parties hereto agree that in case of negligence or wilful misconduct on the part of the Consultant or on the part of any person or firm acting on behalf of the Consultant in carrying out the Services, the Consultant, with respect to damage caused to the Authority's property, shall not be liable to the Authority:
 - (i) for any indirect or consequential loss or damage; and
 - (ii) for any direct loss or damage that exceeds (a) the Agreement Value set forth in Clause 6.1.2 of this Agreement, or (b) the proceeds the Consultant may be entitled to receive from any insurance maintained by the Consultant to cover such a liability, whichever of (a) or (b) is higher.
- 3.4.4 This limitation of liability specified in Clause 3.4.3 shall not affect the Consultant's liability, if any, for damage to Third Parties caused by the Consultant or any person or firm acting on behalf of the Consultant in carrying out the Services subject, however, to a limit equal to 3 (three) times the Agreement Value.

3.5 Accounting, inspection and auditing

The Consultant shall:

- (a) keep accurate and systematic accounts and records in respect of the Services provided under this Agreement, in accordance with internationally accepted accounting principles and standards such as Indian Accounting Standards, GAAP, etc and in such form and detail as will clearly identify all relevant time charges and cost, and the basis thereof (including the basis of the Consultant's costs and charges); and
- (b) permit the Authority or its designated representative periodically, and up to one year from the expiration or termination of this Agreement, to inspect the same and make copies thereof as well as to have them audited by auditors appointed by the Authority.

3.6 Consultant's actions requiring the Authority's prior approval

The Consultant shall obtain the Authority's prior approval in writing before taking any of the following actions:

- (a) Appointing such members of the professional personnel as are not listed in Appendix-1, Form 6;
- (b) entering into a subcontract for the performance of any part of the Services, it being understood (i) that the selection of the Sub-consultant and the terms and conditions of the subcontract shall have been approved in writing by the Authority prior to the execution of the subcontract, and (ii) that the Consultant shall remain fully liable for the performance of the Services by the Sub-Consultant and its Personnel pursuant to this Agreement; or
- (c) any other action that is specified in this Agreement.

3.7 Reporting obligations

3.7.1 The Consultant shall submit to the Authority the reports and documents specified in the Agreement, in the form, in the numbers and within the time periods set forth therein.

3.7.2 Deleted

3.8 Documents prepared by the Consultant to be property of the Authority

- 3.8.1 All reports and other documents (collectively referred to as "Consultancy Documents") prepared by the Consultant (or by the Sub-Consultants or any Third Party) in performing the Services shall become and remain the property of the Authority, and all intellectual property rights in such Consultancy Documents shall vest with the Authority. Any Consultancy Document, of which the ownership or the intellectual property rights do not vest with the Authority under law, shall automatically stand assigned to the Authority as and when such Consultancy Document is created and the Consultant agrees to execute all papers and to perform such other acts as the Authority may deem necessary to secure its rights herein assigned by the Consultant.
- 3.8.2 The Consultant shall, not later than termination or expiration of this Agreement, deliver all Consultancy Documents to the Authority, together with a detailed inventory thereof. The Consultant may retain a copy of such Consultancy Documents. The Consultant, its Sub-Consultants or a Third Party shall not use these Consultancy Documents for purposes unrelated to this Agreement without the prior written approval of the Authority.
- 3.8.3 The Consultant shall hold the Authority harmless and indemnified for any losses, claims, damages, expenses (including all legal expenses), awards, penalties or injuries (collectively referred to as "Claims") which may arise from or due to any unauthorised use of such Consultancy Documents, or due to any breach or failure on part of the Consultant or its Sub-Consultants or a Third Party to perform any of its duties or obligations in relation to securing the aforementioned rights of the Authority.

3.9 Materials furnished by the Authority

Materials made available to the Consultant by the Authority shall be the property of the Authority and shall be marked accordingly. Upon termination or expiration of this Agreement, the Consultant shall furnish forthwith to the Authority, an inventory of such materials and shall dispose of such materials in accordance with the instructions of the Authority.

3.10 Providing access to Project Office and Personnel

The Consultant shall ensure that the Authority, and officials of the Authority having authorization from the Authority, are provided unrestricted access to the office of the Consultant and to all Personnel during office hours. The Authority's official, who has been authorised by the Authority in this behalf, shall have the right to inspect the Services in progress, interact with Personnel of the Consultant and verify the records relating to the Services for his satisfaction.

3.11 Accuracy of Documents

The Consultant shall be responsible for accuracy of the documents drafted and/ or vetted and data collected by it directly or procured from other agencies/authorities, estimates and all other details prepared by it as part of these services. Subject to the provisions of Clause 3.4, it shall indemnify the Authority against any inaccuracy in its work which might surface during implementation of the Project, if such inaccuracy is the result of any negligence or inadequate due diligence on part of the Consultant or arises out of its failure to conform to good industry practice. The Consultant shall also be responsible for promptly correcting, at its own cost and risk, the documents including any re-survey / investigations.

4 CONSULTANT'S PERSONNEL AND SUB-CONSULTANTS

4.1 General

The Consultant shall employ and provide such qualified and experienced Personnel as may be required to carry out the Services.

4.2 Deployment of Personnel

- 4.2.1 The designations, names and other particulars of each of the Consultant's Key Personnel required in carrying out the Services are described in Form-6 of Appendix-I of this Agreement. The estimated deployment schedule of key personnel is provided in Annexure-2 of this Agreement.
- 4.2.2 Adjustments with respect to the estimated periods of engagement of Personnel set forth in the aforementioned Form-3 of Appendix-II may be made by the Consultant by written notice to the Authority, provided that (i) such adjustments shall not alter the originally estimated period of engagement of any individual by more than 30% (thirty per cent), and (ii) the aggregate of such adjustments shall not

cause payments under the Agreement to exceed the Agreement Value set forth in Clause 6.1.2 of this Agreement by more than 25% (twenty five per cent) thereof. Any other adjustments shall only be made with the written approval of the Authority.

4.2.3 If additional work is required beyond the scope of the Services specified in the Terms of Reference, the estimated periods of engagement of Personnel, set forth in the Annexes of the Agreement may be increased by agreement in writing between the Authority and the Consultant, provided that any such increase shall not, except as otherwise agreed, cause payments under this Agreement to exceed the Agreement Value set forth in Clause 6.1.2.

4.3 Approval of Personnel

- 4.3.1 The Key Personnel listed in Form-6 of Appendix-I of the Agreement are hereby approved by the Authority. No other Key Personnel shall be engaged without prior approval of the Authority.
- 4.3.2 If the Consultant hereafter proposes to engage any person as Professional Personnel, it shall submit to the Authority its proposal along with a CV of such person in the form provided at Form No. 6 of Appendix–I of the RFP. The Authority may approve or reject such proposal within 14 (fourteen) days of receipt thereof. In case the proposal is rejected, the Consultant may propose an alternative person for the Authority's consideration. In the event the Authority does not reject a proposal within 14 (fourteen) days of the date of receipt thereof under this Clause 4.3, it shall be deemed to have been approved by the Authority.

4.4 Substitution of Key Personnel

The Authority expects all the Key Personnel specified in the Proposal to be available during implementation of the Agreement. The Authority will not consider any substitution of Key Personnel except under compelling circumstances beyond the control of the Consultant and the concerned Key Personnel. Such substitution shall be limited to not more than two Key Personnel subject to equally or better qualified and experienced personnel being provided to the satisfaction of the Authority. Without prejudice to the foregoing, substitution of Key Personnel shall be permitted only upon reduction of remuneration weightage specified for the original Key Personnel in the scoring criteria as per clause 3.1.3 shall be deducted from the payments due to the Consultant.

The Consultant should specifically note that substitution of the Team Leader cum PPP Expert will not normally be considered during the implementation of the Agreement and may lead to disqualification of the Applicant or termination of the Agreement. The PPP Expert must commit the time required for and be available for delivering the Consultancy in accordance with the terms specified herein.

4.5 Deleted

4.6 PPP Expert and Project Manager

The person designated as the PPP Expert of the Consultant's Personnel shall be responsible for the coordinated, timely and efficient functioning of the Personnel. In addition, the Consultant shall designate a suitable person as Project Manager (the "Project Manager") who shall be responsible for day to day performance of the Services.

4.7 Deleted.

5. OBLIGATIONS OF THE AUTHORITY

5.1 Assistance in clearances etc.

Unless otherwise specified in the Agreement, the Authority shall make best efforts to ensure that the Government shall:

- (a) provide the Consultant, its Sub-Consultants and Personnel with work permits and such other documents as may be necessary to enable the Consultant, its Sub-Consultants or Personnel to perform the Services;
- (b) facilitate prompt clearance through customs of any property required for the Services; and
- (c) issue to officials, agents and representatives of the Government all such instructions as may be necessary or appropriate for the prompt and effective implementation of the Services.

5.2 Access to land and property

The Authority warrants that the Consultant shall have, free of charge, unimpeded access to the site of the project in respect of which access is required for the performance of Services; provided that if such access shall not be made available to the Consultant as and

when so required, the Parties shall agree on (i) the time extension, as may be appropriate, for the performance of Services, and (ii) the additional payments, if any, to be made to the Consultant as a result thereof pursuant to Clause 6.1.3.

5.3 Change in Applicable Law

All service tax and other taxes other than income tax, as may be applicable from time to time, on the payment of the professional fees to the Consultant, shall be borne by the Authority.

5.4 Payment

In consideration of the Services performed by the Consultant under this Agreement, the Authority shall make to the Consultant such payments and in such manner as is provided in Clause 6 of this Agreement.

6. PAYMENT TO THE CONSULTANT

6.1 Cost estimates and Agreement Value

- 6.1.1 An abstract of the cost of the Services payable to the Consultant is set forth in 'Schedule of Prices', Form-2 of Appendix II of the Agreement.
- 6.1.3 Notwithstanding anything to the contrary contained in Clause 6.1.2, if pursuant to the provisions of Clause 2.6, the Parties agree that additional payments shall be made to the Consultant in order to cover any additional expenditures not envisaged in the cost estimates referred to in Clause 6.1.1 above, the Agreement Value set forth in Clause 6.1.2 above shall be increased by the amount or amounts, as the case may be, of any such additional payments. Deleted.

6.2 Currency of payment

All payments shall be made in Indian Rupees. The Consultant shall be free to convert Rupees into any foreign currency as per Applicable Laws.

6.3 Mode of billing and payment

Billing and payments in respect of the Services shall be made as follows:-

- (a) The Consultant shall be paid for its services as per the Payment Schedule at clause 4 & 5, sub-clauses 5.2, 5.3, 5.4, 5.5, 5.6 and 5.7 of the TOR, and the rates specified in the Price Schedule of this Agreement / RFP, subject to the Consultant fulfilling the following conditions:
 - (i) No payment shall be due for the next stage till the Consultant completes to the satisfaction of the Authority the work pertaining to the preceding stage; and
 - (ii) The Authority shall pay to the Consultant, only the undisputed amount.
- (b) The Authority shall cause the payment due to the Consultant to be made within 30 (thirty) days after the receipt by the Authority of duly completed bills with necessary particulars (the "Due Date").
- (c) The final payment under this Clause 6.3 shall be made only after the final Deliverable shall have been submitted by the Consultant and approved as satisfactory by the Authority. The Services shall be deemed completed and finally accepted by the Authority and the final Deliverable shall be deemed approved by the Authority as satisfactory upon expiry of 60 (sixty) days after receipt of the final Deliverable unless the Authority, within such 60 (sixty) day period, gives written notice to the Consultant specifying in detail, the deficiencies in the Services. The Consultant shall thereupon promptly make any necessary corrections and/or additions, and upon completion of such corrections or additions, the foregoing process shall be repeated. The Authority shall make the final payment upon acceptance or deemed acceptance of the final Deliverable by the Authority.
- (d) Any amount which the Authority has paid or caused to be paid in

excess of the amounts actually payable in accordance with the provisions of this Agreement shall be reimbursed by the Consultant to the Authority within 30 (thirty) days after receipt by the Consultant of notice thereof. Any such claim by the Authority for reimbursement must be made within 1 (one) year after receipt by the Authority of a final report in accordance with Clause 6.3 (c). Any delay by the Consultant in reimbursement by the due date shall attract simple interest @ PLR SBI +2% (two per cent) per annum.

(e) All payments under this Agreement shall be made to the account of the Consultant as may be notified to the Authority by the Consultant.

7. Performance Security, Liquidated Damages and Penalties7.1 Performance Security

Within ten days of the receipt of the award letter from Employer, the successful bidder/s shall furnish to the Employer a performance security in the form of a Bank Guarantee of a Nationalized/ Scheduled Bank, enforceable and encashable at Mangalore for an amount equivalent to 10% of the awarded cost/ fee including GST rounded off to nearest highest 1000 as per the draft proforma specified at Annexure-1 of the RFP document. Failure of the successful bidder to lodge and subsequently renew the required Bank Guarantee shall be constitute sufficient grounds for the termination of the contract and forfeiting of "Bid security". The performance security shall remain in force until the satisfactory completion of the contract and will be discharged thereafter. The obtaining of such guarantee (and the cost of guarantee) shall be at the expense of the Bidder/s

7.2 Liquidated Damages

7.2.1 Liquidated Damages for error/variation

In case any error or variation is detected in the reports submitted by the Consultant and such error or variation is the result of negligence or lack of due diligence on the part of the Consultant, the consequential damages thereof shall be quantified by the Authority in a reasonable manner and recovered from the Consultant by way of deemed liquidated damages, subject to a maximum of the Agreement Value.

7.2.2 Liquidated Damages for delay

In case of delay in completion of Services, liquidated damages not exceeding an amount equal to 0.5% (zero point five per cent) per week of the activity specified in Time & Payment Schedule - Clause 5 of TOR, subject to a maximum of 10% (ten per cent) of Agreement Value will be imposed and shall be recovered by appropriation from the Performance Security or otherwise. However, in case of delay due to reasons beyond the control of the Consultant, suitable extension of time shall be granted.

7.2.3 Encashment and appropriation of Performance Security

The Authority shall have the right to invoke and appropriate the proceeds of the Performance Security, in whole or in part, without notice to the Consultant in the event of breach of this Agreement or for recovery of liquidated damages specified in this Clause 7.2.

7.3 Penalty for deficiency in Services

In addition to the liquidated damages not amounting to penalty, as specified in Clause 7.2, warning may be issued to the Consultant for minor deficiencies on its part. In the case of significant deficiencies in Services causing adverse effect on the Project or on the reputation of the Authority, other penal action including debarring for a specified period may also be initiated as per policy of the Authority.

In the event of forfeiting the EMD/SD/Performance Security and imposition of liquidated damages, GST is applicable and while imposing penalty GST shall be collected. The GST shall be shown claimed as a separate line item in Tax Invoice. The Consultant shall file the applicable returns with Tax Dept. in time and submit the same as a documentary proof. In case, input Tax credit lost due to non filing of returns will be recovered from Consultant.

8. FAIRNESS AND GOOD FAITH

8.1 Good Faith

The Parties undertake to act in good faith with respect to each other's rights under this Agreement and to adopt all reasonable measures to ensure the realisation of the objectives of this Agreement.

8.2 Operation of the Agreement

The Parties recognise that it is impractical in this Agreement to provide for every contingency which may arise during the life of the Agreement, and the Parties hereby agree that it is their intention that this Agreement shall operate fairly as between them, and without detriment to the interest of either of them, and that, if during the term of this Agreement either Party believes that this Agreement is

operating unfairly, the Parties will use their best efforts to agree on such action as may be necessary to remove the cause or causes of such unfairness, but failure to agree on any action pursuant to this Clause 8.2 shall not give rise to a dispute subject to arbitration in accordance with Clause 9 hereof.

9. SETTLEMENT OF DISPUTES

9.1 Amicable settlement

The Parties shall use their best efforts to settle amicably all disputes arising out of or in connection with this Agreement or the interpretation thereof.

9.2 Dispute resolution

- 9.2.1 Any dispute, difference or controversy of whatever nature howsoever arising under or out of or in relation to this Agreement (including its interpretation) between the Parties, and so notified in writing by either Party to the other Party (the "Dispute") shall, in the first instance, be attempted to be resolved amicably in accordance with the conciliation procedure set forth in Clause 9.3.
- 9.2.2 The Parties agree to use their best efforts for resolving all Disputes arising under or in respect of this Agreement promptly, equitably and in good faith, and further agree to provide each other with reasonable access during normal business hours to all non-privileged records, information and data pertaining to any Dispute.

9.3 Conciliation

In the event of any Dispute between the Parties, either Party may call upon Secretary, Ministry of Shipping and the Managing Partner/ Chairman of the Board of Directors of the Consultant or a substitute thereof for amicable settlement, and upon such reference, the said persons shall meet no later than 10 (ten) days from the date of reference to discuss and attempt to amicably resolve the Dispute. If such meeting does not take place within the 10 (ten) day period or the Dispute is not amicably settled within 15 (fifteen) days of the meeting or the Dispute is not resolved as evidenced by the signing of written terms of settlement within 30 (thirty) days of the notice in writing referred to in Clause 9.2.1 or such longer period as may be mutually agreed by the Parties, either Party may refer the Dispute to arbitration in accordance with the provisions of Clause 9.4.

9.4 Arbitration

9.4.1 Any Dispute which is not resolved amicably by conciliation, as provided in Clause 9.3, shall be finally decided by reference to

arbitration by an Arbitral Tribunal appointed in accordance with Clause 9.4.2. Such arbitration shall be held in accordance with the Rules of Arbitration of the International Centre for Alternative Dispute Resolution, New Delhi (the "Rules"), or such other rules as may be mutually agreed by the Parties, and shall be subject to the provisions of the Arbitration and Conciliation (Amendment) Act,2015. Arbitration and Conciliation Act,1996. The venue of such arbitration shall be Mangalore and the language of arbitration proceedings shall be English.

- 9.4.2 There shall be a sole arbitrator whose appointment shall be made in accordance with the Rules.
- 9.4.3 The arbitrators shall make a reasoned award (the "Award"). Any Award made in any arbitration held pursuant to this Clause 9 shall be final and binding on the Parties as from the date it is made, and the Consultant and the Authority agree and undertake to carry out such Award without delay.
- 9.4.4 The Consultant and the Authority agree that an Award may be enforced against the Consultant and/or the Authority, as the case may be, and their respective assets wherever situated.
- 9.4.5 This Agreement and the rights and obligations of the Parties shall remain in full force and effect, pending the Award in any arbitration proceedings hereunder.

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be signed in their respective names as of the day and year first above written

SCHEDULE-3

(See Clause 2.3.3)

Guidance Note on Conflict of Interest

- 1. This Note further explains and illustrates the provisions of Clause 2.3 of the RFP and shall be read together therewith in dealing with specific cases.
- Consultants should be deemed to be in a conflict of interest situation
 if it can be reasonably concluded that their position in a business or
 their personal interest could improperly influence their judgment in
 the exercise of their duties. The process for selection of consultants
 should avoid both actual and perceived conflict of interest.
- 3. Conflict of interest may arise between the Authority and a consultant or between consultants and present or future concessionaries/ contractors. Some of the situations that would involve conflict of interest are identified below:
 - (a) Authority and consultants:
 - (i) Potential consultant should not be privy to information from the Authority which is not available to others.
 - (ii) Potential consultant should not have defined the project when earlier working for the Authority.
 - (iii) Potential consultant should not have recently worked for the Authority overseeing the project.
 - (b) Consultants and concessionaires/contractors:
 - (i) No consultant should have an ownership interest or a continuing business interest or an on-going relationship with a potential concessionaire/ contractor save and except relationships restricted to project-specific and short-term assignments.
 - (ii) No consultant should be involved in owning or operating entities resulting from the project.
 - (iii) No consultant should bid for works arising from the project.

The participation of companies that may be involved as investors or consumers and officials of the Authority who have current or recent connections to the companies involved, therefore, needs to be avoided.

- 4. The normal way to identify conflicts of interest is through self-declaration by consultants. Where a conflict exists, which has not been declared, competing companies are likely to bring this to the notice of the Authority. All conflicts must be declared as and when the consultants become aware of them.
- 5. Another approach towards avoiding a conflict of interest is through the use of "Chinese walls" to avoid the flow of commercially sensitive information from one part of the consultant's company to another. This could help overcome the problem of availability of limited numbers of experts for the project. However, in reality effective operation of "Chinese walls" may be a difficult proposition. As a general rule, larger companies will be more capable of adopting Chinese walls approach than smaller companies. Although, "Chinese walls" have been relatively common for many years, they are an increasingly discredited means of avoiding conflicts of interest and should be considered with caution. As a rule, "Chinese walls" should be considered as unacceptable and may be accepted only in exceptional cases upon full disclosure by a consultant coupled with provision of safeguards to the satisfaction of the Authority.
- 6. Another way to avoid conflicts of interest is through the appropriate grouping of tasks. For example, conflicts may arise if consultants drawing up the terms of reference or the proposed documentation are also eligible for the consequent assignment or project.
- 7. Another form of conflict of interest called "scope-creep" arises when consultants advocate either an unnecessary broadening of the terms of reference or make recommendations which are not in the best interests of the Authority but which will generate further work for the consultants. Some forms of contractual arrangements are more likely to lead to scope-creep. For example, lump-sum contracts provide fewer incentives for this, while time and material contracts provide built in incentives for consultants to extend the length of their assignment.
- 8. Every project contains potential conflicts of interest. Consultants should not only avoid any conflict of interest, they should report any present/ potential conflict of interest to the Authority at the earliest. Officials of the Authority involved in development of a project shall be responsible for identifying and resolving any conflicts of interest. It should be ensured that safeguards are in place to preserve fair and open competition and measures should be taken to eliminate any conflict of interest arising at any stage in the process.

Annexure -1

Bank Guarantee for Performance Security (Refer Clause 7.1)

In consideration of Chairman acting on behalf of the New Mangalore Port

To
Board Members
New Mangalore Port Authority
acting through
Chairman New
Mangalore Port
Authority

Authority (hereinafter referred as the "Authority", which expression shall, unless repugnant to the context or meaning thereof, include its successors, administrators and assigns) having awarded to M/s, having its office at (hereinafter referred as the "Consultant" which expression shall, unless repugnant to the context or meaning thereof, include its successors, administrators, executors and assigns), vide the Authority's Agreement no dated valued at Rs
(Rupees), (hereinafter referred to as the "Agreement") the work of Transaction advisory services for Concessioning of Berth No.13 for handling Liquid Bulk / LPG / POL products / Chemicals at New Mangalore Port on Captive / PPP Mode and the Consultant having agreed to furnish a Bank Guarantee amounting to Rs
We, (hereinafter referred to as the "Bank") at the request of the Consultant do hereby undertake to pay to the Authority an amount not exceeding Rs (Rupees) against any loss or damage caused to or suffered or would be caused to or suffered by the Authority by reason of any breach by the said Consultant of any of the terms or conditions contained in the said Agreement.
2. We, (indicate the name of the Bank) do hereby undertake to pay the amounts due and payable under this Guarantee without any demur, merely on a demand from the Authority stating that the amount/claimed is due by way of loss or damage caused to or would be caused to or suffered by the Authority by reason of breach by the said Consultant of any of the terms or conditions contained in the said Agreement or by reason of the Consultant's failure to perform the said Agreement. Any such demand made on the bank shall be conclusive as regards the amount due and payable by the Bank under this Guarantee. However, our liability under this Guarantee shall be restricted to an amount not exceeding Rs

- 3. We, (indicate the name of Bank) undertake to pay to the Authority any money so demanded notwithstanding any dispute or disputes raised by the Consultant in any suit or proceeding pending before any court or tribunal relating thereto, our liability under this present being absolute and unequivocal. The payment so made by us under this bond shall be a valid discharge of our liability for payment thereunder and the Consultant shall have no claim against us for making such payment.
- 4. We, (indicate the name of Bank) further agree that the Guarantee herein contained shall remain in full force and effect during the period that would be taken for the performance of the said Agreement and that it shall continue to be enforceable till all the dues of the Authority under or by virtue of the said Agreement have been fully paid and its claims satisfied or discharged or till the Authority certifies that the terms and conditions of the said Agreement have been fully and properly carried out by the said Consultant and accordingly discharges this Guarantee. Unless a demand or claim under this Guarantee is made on us in writing on or before a period of one year from the date of this Guarantee, we shall be discharged from all liability under this Guarantee thereafter.
- 6. This Guarantee will not be discharged due to the change in the constitution of the Bank or the Consultant(s).
- 7. We, (indicate the name of Bank) lastly undertake not to revoke this Guarantee during its currency except with the previous consent of the Authority in writing.
- 8. For the avoidance of doubt, the Bank's liability under this Guarantee shall be restricted to Rs. *** crore (Rupees ***** crore) only. The Bank shall be liable to pay the said amount or any part thereof only if the Authority

serves a written claim on the Bank in accordance with paragraph 2 hereof, on or before [*** (indicate date falling 180 days after the Bid Due Date specified in the RFQ/RFP for the Project)].

Dated, the	day of 20	
For		
(Name of Bank)		
	•	the authorised gnatory) Seal of the ank:

NOTES:

- (i) The Bank Guarantee should contain the name, designation and code number of the officer(s) signing the Guarantee.
- (ii) The address, telephone no. and other details of the Head Office of the Bank as well as of issuing Branch should be mentioned on the covering letter of issuing Branch.

Annexure -2

(Refer Clause 7.1 of ToR)

Proposed Deployment Schedule

Key Personnel	Man weeks	1	2	3	5	6	7	8	9	•	•	52
PPP Expert cum Team Leader												
Port Sector Expert												
Legal Consultant												
Associate Consultant												

APPENDICES

(See Clause 2.1.3)

TECHNICAL PROPOSAL

Form-1

Letter of Proposal (On Applicant's letter head)

(Date and Reference)

To

Sub: Appointment of Transaction Advisor for Concessioning of Berth No.13 for handling Liquid Bulk / LPG / POL products / Chemicals at New Mangalore Port on Captive / PPP Mode

Dear Sir,

With reference to your RFP Document dated, I/We, having examined all relevant documents and understood their contents, hereby submit our Proposal for Appointment of Transaction Advisor for Concessioning of Berth No.13 for handling Liquid Bulk / LPG / POL products / Chemicals at New Mangalore Port on Captive / PPP Mode.

- . The proposal is unconditional and unqualified.
- 2. I/We acknowledge that the Authority will be relying on the information provided in the Proposal and the documents accompanying the Proposal for selection of the Consultant, and we certify that all information provided in the Proposal and in the Appendices is true and correct, nothing has been omitted which renders such information misleading; and all documents accompanying such Proposal are true copies of their respective originals.
- 3. This statement is made for the express purpose of appointment as the Consultant for the aforesaid Project.
- 4. I/We shall make available to the Authority any additional information it may deem necessary or require for supplementing or authenticating the Proposal.
- 5. I/We acknowledge the right of the Authority to reject our application without assigning any reason or otherwise and hereby waive our right to challenge the same on any account whatsoever.

6. I/We certify that in the last three years, we or any of our Associates have neither failed to perform on any contract, as evidenced by imposition of a penalty by an arbitral or judicial authority or a judicial pronouncement or arbitration award against the Applicant, nor been expelled from any project or contract by any public authority nor have had any contract terminated by any public authority for breach on our part.

7. I/We declare that:

- (a) I/We have examined and have no reservations to the RFP Documents, including any Addendum issued by the Authority;
- (b) I/We do not have any conflict of interest in accordance with Clause 2.3 of the RFP Document;
- (c) I/We have not directly or indirectly or through an agent engaged or indulged in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice, as defined in Clause 4.3 of the RFP document, in respect of any tender or request for proposal issued by or any agreement entered into with the Authority or any other public sector enterprise or any government, Central or State; and
- (d) I/We hereby certify that we have taken steps to ensure that in conformity with the provisions of Section 4 of the RFP, no person acting for us or on our behalf will engage in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice.
- 8. I/We understand that you may cancel the Selection Process at any time and that you are neither bound to accept any Proposal that you may receive nor to select the Consultant, without incurring any liability to the Applicants in accordance with Clause 2.8 of the RFP document.
- 9. I/We declare that we are not a member of any other Consortium applying for Selection as a Consultant.
- 10. I/We certify that in regard to matters other than security and integrity of the country, we or any of our Associates have not been convicted by a Court of Law or indicted or adverse orders passed by a regulatory authority which would cast a doubt on our ability to undertake the Consultancy for the Project or which relates to a grave offence that outrages the moral sense of the community.

- 11. I/We further certify that in regard to matters relating to security and integrity of the country, we have not been charge-sheeted by any agency of the Government or convicted by a Court of Law for any offence committed by us or by any of our Associates.
- 12. I/We further certify that no investigation by a regulatory authority is pending either against us or against our Associates or against our CEO or any of our Directors/Managers/employees.
- 13. I/We hereby irrevocably waive any right or remedy which we may have at any stage at law or howsoever otherwise arising to challenge or question any decision taken by the Authority [and/ or the Government of India] in connection with the selection of Consultant or in connection with the Selection Process itself in respect of the above mentioned Project.
- 14. I/We agree and understand that the proposal is subject to the provisions of the RFP document. In no case, shall I/we have any claim or right of whatsoever nature if the Consultancy for the Project is not awarded to me/us or our proposal is not opened or rejected.
- 15. I/We agree to keep this offer valid for 90 (ninety) days from the PDD specified in the RFP.
- 16. A Power of Attorney in favour of the authorised signatory to sign and submit this Proposal and documents is attached herewith in Form 4.
- 17. In the event of my/our firm being selected as the Consultant, I/we agree to enter into an Agreement in accordance with the form at Schedule–2 of the RFP. We agree not to seek any changes in the aforesaid form and agree to abide by the same.
- 18. In the event of my/our firm being selected as the Consultant, I/we agree and undertake to provide the services of the PPP Expert in accordance with the provisions of the RFP and that the PPP Expert shall be responsible for providing the agreed services himself and not through any other person or Associate.
- 19. I/We have studied RFP and all other documents carefully. We understand that except to the extent as expressly set forth in the Agreement, we shall have no claim, right or title arising out of any documents or information provided to us by the Authority or in respect of any matter arising out of or concerning or relating to the Selection Process including the award of Consultancy.
- 20. The Financial Proposal is being submitted in a separate cover. This Technical Proposal read with the Financial Proposal shall constitute

the Application which shall be binding on us.

21. I/We agree and undertake to abide by all the terms and conditions of the RFP Document.

In witness thereof, I/we submit this Proposal under and in accordance with the terms of the RFP Document.

Yours faithfully,

(Signature, name and designation of the authorised signatory)

(Name and seal of the Applicant/ Lead Member)

Form-2

Particulars of the Applicant

1.1	Title of Consultanting services: TRANSACTION ADVISOR
1.2	Title of Project: Appointment of Transaction Advisor for Concessioning of Berth No.13 for handling Liquid bulk / LPG / POL products / Chemicals at New Mangalore Port on Captive / PPP Mode
1.3	State whether applying as Sole Firm or Lead Member of a consortium: Sole Firm or Lead Member of a consortium
1.4	State the following:
	Name of Company or Firm:
	Legal status (e.g. incorporated private company, unincorporated business, partnership etc.):
	Country of incorporation: Registered address: \Year of Incorporation: Year of commencement of business: Principal place of business:
	Brief description of the Company including details of its main lines of business
	Name, designation, address and phone numbers of Authorised Signatory of the Applicant:
	Name: Designation: Company: Address
	Phone No.: Fax No. : E-mail address::
1.5	If the Applicant is Lead Member of a consortium, state the following for each of the other Member Firms:

Form-3

Statement of Legal Capacity

(To be forwarded on the letter head of the Applicant)

Ref. Date:
То
Chief Engineer Administrative Office Building ^{2rd} Floor, New Mangalore Port Trust Panambur Mangalore 575010
Dear Sir,
Sub: APPOINTMENT OF TRANSACTION ADVISOR FOR CONCESSIONING OF BERTH NO.13 FOR HANDLING LIQUID BULK LPG / POL PRODUCTS / CHEMICALS ON CAPTIVE / PPP MODE AT NEW MANGALORE PORT
I/We hereby confirm that we, the Applicant (along with other members in case of consortium, constitution of which has been described in the Proposal*), satisfy the terms and conditions laid down in the RFP document.
I/We have agreed that (insert Applicant's name) will act as the Lead Member of our consortium.
I/We have agreed that (insert individual's name) will act as our Authorised Representative/ will act as the Authorised Representative of the consortium on our behalf and has been duly authorized to submit our Proposal. Further, the authorised signatory is vested with requisite powers to furnish such proposal and all other documents, information or communication and authenticate the same.
Yours faithfully,
(Signature, name and designation of the authorised signatory
*Please strike out whichever is not applicable

Form-4

Financial Capacity of the Applicant

(Ref. Clause 2.3.3)

SI. No.	Financial Year	Annual Revenue (Rs.)				
1.	2019-20	,				
2.	2020-21					
3.	2021-22					
	Certificate from the Statutory Audit	or ^{\$}				
This is to certify that (name of the Applicant) has received the payments shown above against the respective years on account of professional fees.						
(Signature, name and designation of the authorised signatory)						
Date:	Name and seal of the audit fi	rm:				

^{\$} In case the Applicant does not have a statutory auditor, it shall provide the certificate from its chartered accountant that ordinarily audits the annual accounts of the Applicant.

Form-5

Power of Attorney

Know all men by these presen	ts, We,	(name of
Firm and address of the registe		•
appoint and authorise Mr / M		
and presently residing at		
employed with/ retained by us		
true and lawful attorney (he		
Representative") to do in our na		
and things as are necessary or		
submission of our Proposal		
Providing Transaction Advisory		
for Handling POL / Chemicals		
Port, proposed to be developed including but not limited to si		
proposals and other document		
other conferences and providing		
representing us in all matters be		
all contracts and undertakings	, · · · · ·	
and generally dealing with the	•	
relating to or arising out of our	Proposal for the said Proje	ct and/or upon
award thereof to us till the	entering into of the Agree	ment with the
Authority.		
AND we do hereby caree to r	atificand confirm all pata do	ada and things
AND, we do hereby agree to re	-	_
lawfully done or caused to be of pursuant to and in exercise of		=
Attorney and that all acts, dee		
Representative in exercise of t		
always be deemed to have bee		Silali alia Silali
always be decined to have bee	in done by do.	
IN WITNESS WHEREOF W	VE, THE AB	OVE NAMED
PRINCIPAL HAVE EXECUTED		
DAY OF		
	, 20**	
	For	
	(Signature, name, designation	on and
	address)	
	,	
\\ /ita = = = = = :		
Witnesses:		

1.

2.

Notarised

Accepted

(Signature, name, designation and address of the Attorney)

Notes:

The mode of execution of the Power of Attorney should be in accordance with the procedure, if any, laid down by the applicable law and the charter documents of the executant(s) and when it is so required the same should be under common seal affixed in accordance with the required procedure.

Wherever required, the Applicant should submit for verification the extract of the charter documents and other documents such as a resolution/power of attorney in favour of the person executing this Power of Attorney for the delegation of power hereunder on behalf of the Applicant.

For a Power of Attorney executed and issued overseas, the document will also have to be legalised by the Indian Embassy and notarised in the jurisdiction where the Power of Attorney is being issued. However, the Power of Attorney provided by Applicants from countries that have signed the Hague Legislation Convention, 1961 are not required to be legalised by the Indian Embassy if it carries a conforming Appostille certificate.

APPENDIX-I Form-6

Particulars of Key Personnel

(Ref. Clause 2.3.4)

SI. No			ational Length of Professional		Employment	No. of Eligible Assignment	Remuneration of Key	
140	Other reformer		Qualification	Experience	Name of Firm	Employed Since	Assignment	personnel
1	2	3	4	5	6	7	8	9
1	Port Expert cum Team Leader							
2	Port Sector Expert							
3	Legal Consultant							
4	Associate Consultant(s)							

^{*}Refer Form 8 of Appendix I Eligible Assignments of Key Personnel

Form-7

Abstract of Eligible Assignments of the Applicant[#] (Refer Clause 3.1)

Professional fee## S.No Name of Name of **Estimated Project** Client capital cost of received by the Project (in Rs Applicant (in Rs crore/ US\$ crore) million) $(1)^{*}$ (2) (3)(4) (5) 1 2 3 4

The Applicant should provide details of only those assignments that have been undertaken by it under its own name.

Exchange rate should be taken as Rs. 75 per US \$ for conversion to Rupees.

Copy of Contract/work orders along with completion certificate indicating the details of previous assignment completed, client, value of assignment/proportionate value in use of projects/ assignment in process date and year of award.

* The names and chronology of Eligible Assignments included here should conform to the project-wise details submitted in Form-9 of Appendix-I.

Certificate from the Statutory Auditor\$

This is to certify that the information contained in Column 5 above is correct as per the accounts of the Applicant and/ or the clients.

(Signature, name and designation of the authorised signatory)

Date: Name and seal of the audit firm:

^{\$} In case the Applicant does not have a statutory auditor, it shall provide the certificate from its chartered accountant that ordinarily audits the annual accounts of the Applicant.

Form-7A

Abstract of other relevant experience of the Applicant[#] (Refer Clause 3.1.5)

S.No	Name of Project	Name of Client	Estimated capital cost of Project (in Rs crore/ US\$ million)	Professional fee ^{##} received by the Applicant (in Rs crore) [£]
(1)*	(2)	(3)	(4)	(5)
1				
2				
3				
4				

[#] The Applicant should provide details of only those assignments that have been undertaken by it under its own name.

Exchange rate should be taken as Rs. 75 per US \$ for conversion to Rupees.

Copy of Contract/work orders along with completion certificate indicating the details of previous assignment completed, client, value of assignment/proportionate value in use of projects/ assignment in process date and year of award.

* The names and chronology of the projects included here should conform to the project-wise details submitted in Form-9A of Appendix-I.

Certificate from the Statutory Auditor\$

This is to certify that the information contained in Column 5 above is correct as per the accounts of the Applicant and/ or the clients.

(Signature, name and designation of the authorised signatory)

Date:

Name and seal of the audit firm:

^{\$} In case the Applicant does not have a statutory auditor, it shall provide the certificate from its chartered accountant that ordinarily audits the annual accounts of the Applicant.

Form - 8[@]

Abstract of Eligible Assignments of Key Personnel^{\$} (Refer Clause 3.1)

Name of Key/ Other Personnel:

Designation:

S.No	Name of Project*	Name of Client	Estimated capital cost of project (in Rs cr./ US\$ million)	Name of firm for which the Key Personnel worked	Designation of the Key Personnel on the assignment	Date of completion of the assignment	Man hours spent on the assignment
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
1							
2							
3							
4							
5							

[@] Use separate Form for each Key Personnel.

^{\$} In the case of PPP Expert, only those assignments shall be included where the PPP Expert worked as Team Leader or Leader of the Financial/ PPP Team in the relevant assignment.

^{*} The names and chronology of projects included here should conform to the project-wise details submitted in Form-10 7 of Appendix-I.

Form-8A[@]

Abstract of other relevant experience of Key Personnel^{\$} (Refer Clause 3.1.5)

Name of Key Personnel:

Designation:

S.No	Name of Project*	Name of Client	Estimated capital cost of project (in Rs cr./ US\$ million)	Name of firm for which the Key Personnel worked	Designation of the Key Personnel on the assignment	Date of completion of the assignment	Man hours spent on the assignment
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
1							
2							
3							
4							
5							

[@] Use separate Form for each Key Personnel.

^{\$} In the case of PPP Expert, only those assignments shall be included where the PPP Expert worked as the Team Leader or the leader of the Financial/ PPP Team in the relevant assignment.

^{*} The names and chronology of assignments included here should conform to the project-wise details submitted in Form 10A of Appendix-I.

Form-9

Eligible Assignments of Applicant (Refer Clause 3.1.4)

Name of Applicant:	
Name of the Project:	
Project size particulars:	
Description of services performed by	
the Applicant firm:	
Name of client and Address:	
Name, telephone no. and fax no. of	
client's representative:	
Estimated capital cost of Project (in Rs	
crore or US\$ million):	
Payment received by the Applicant as	
professional fees (in Rs. crore):	
Start date and finish date of the	
services (month/ year):	
Brief description of the Project:	
Notes:	
1. Use separate sheet for each assignr	nent.
2. Exchange rate should be taken	as Rs. 75 per US \$ for
converting to Rupees.	

Form-9A

Other relevant assignments of Applicant (Refer Clause 3.1.5)

Name of Applicant:	
Name of the Project:	
Project size particulars:	
Description of services performed by	
the Applicant firm:	
Name of client and Address:	
· Name, telephone no. and fax no. of	
client's representative:	
Estimated capital cost of Project (in Rs	
crore or US\$ million):	
Payment received by the Applicant as	
professional fees (in Rs. crore):	
Start date and finish date of the	
services (month/ year):	
Brief description of the Project:	
Notes:	
3. Use separate sheet for each assignr	nent.
4. Exchange rate should be taken	as Rs. 75 per US \$ for
converting to Rupees.	

Form-10

Eligible Assignments of Key Personnel

(Refer Clause 3.1.4)

Notes:

- 1. Use separate sheet for each assignment.
- 2. In the case of PPP Expert, only those assignments shall be included where the PPP Expert worked as the Team Leader or the leader of the Financial Team in the relevant assignment.
- 3. Exchange rate should be taken as Rs. 75 per US \$ for converting in Indian Rupees.

Form-10A

Other relevant assignments of Key Personnel

(Refer Clause 3.1.5)

Name of Key Personnel:	
Designation of Key Personnel:	
Name of the Project:	
Project size particulars:	
Name of Consulting Firm where employed:	
Description of services performed by the Key Personnel (including designation):	
Name of client and Address:	
Name, telephone no. and fax no. of client's representative	
Estimated capital cost of the Project (in Rs crore or US\$ million):	
Start date and finish date of the services (month/ year):	
Brief description of the Project:	

Notes:

- 1. Use separate sheet for each assignment.
- 2. In the case of PPP Expert, only those assignments shall be included where the PPP Expert worked as the Team Leader or the leader of the Financial Team in the relevant assignment.
- 3. Exchange rate should be taken as Rs. 75 per US \$ for converting in Indian Rupees.

Form - 11

Curriculum Vitae (CV) of Professional Personnel

- 1. Proposed Position:
- 2. Name of Personnel:
- Date of Birth:
- 4. Nationality:
- 5. Educational Qualifications:
- 6. Employment Record:

(Starting with present position, list in reverse order every employment held.)

7. List of projects on which the Personnel has worked

Name of Project

Description of responsibilities

Certification:

- I am willing to work on the Project and I will be available for entire duration of the Project assignment as required.
- I, the undersigned, certify that to the best of my knowledge and belief, this CV correctly describes my qualifications, my experience and me.

(Signature and name of the Professional Personnel)

P	lace				
	lacc	 	 	 	

(Signature and name of the authorised signatory of the Applicant)

Notes:

- 1. Use separate form for each Key Personnel and Professional Personnel.
- 2. In the case of PPP Expert, only those assignments shall be included where the PPP Expert worked as the Team Leader or the leader of the Financial/PPP Team in the relevant assignment.
- 3. The names and chronology of assignments included here should conform to the project-wise details submitted in Form-7 or Form-7A, as the case may be, of Appendix-I.
- 4. Each page of the CV shall be signed in ink by both the Personnel concerned and by the Authorised Representative of the Applicant firm along with the seal of the firm. Photocopies will not be considered for evaluation.

APPENDIX-I Form – 12 Integrity Pact

General

General
This pre-bid pre-contract Agreement (hereinafter called the Integrity Pact) is made on day of the month of 20, between, on one hand, the Board of Trustees of New Mangalore Port Authority acting through Shri. , Chief Engineer (Civil), New Mangalore Port Authority
(hereinafter called the 'AUTHORITY', which expression shall mean and include, unless the context otherwise requires, his successors in office and assigns) of the First Part and M/s represented by Shri, Chief Executive Officer (hereinafter called the "APPLICANT" which expression shall mean and include, unless the context otherwise requires, his successors and permitted assigns) of the Second Part.
WHEREAS the AUTHORITY has invited bids for the project of "Request for Proposal (RFP) for "Appointment of Transaction Advisor for Concessioning of Berth No. 13 for handling Liquid Bulk / LPG / POL products / Chemicals on Captive / PPP Mode at New Mangalore Port" (hereinafter referred to as the "Work") and the "APPLICANT" is submitting its Bid for the Project, and
WHEREAS the APPLICANT is a Private Limited Company/ Public Limited Company/ Government undertaking/ registered partnership firm/ Joint Venture/ Consortium, constituted in accordance with the relevant law in the matter.
AUTHORITY NOW, THEREFORE,
To avoid all forms of corruption by following a system that is fair, transparent and free from any influence/prejudiced dealings prior to, during and subsequent to the currency of the contract to be entered into with a view to:-
Enabling APPLICANTS to abstain from bribing or indulging in any corrupt practice in order to secure the contract by providing assurance to them that their competitors will also abstain from bribing and other corrupt practices and the AUTHORITY will commit to prevent corruption, in any form, by its officials by

New Mangalore Port 123

The parties hereto hereby agree to enter into this Integrity Pact and agree as

following transparent procedures.

follows:

Commitments of the AUTHORITY

- 1.1 The AUTHORITY undertakes that no official of the AUTHORITY, connected directly or indirectly with the contract, will demand, take a promise for or accept, directly or through intermediaries, any bribe, consideration, gift, reward, favour or any material or immaterial benefit or any other advantage from the APPLICANT, either for themselves or for any person, organisation or third party related to the contract in exchange for an advantage in the bidding process, bid evaluation, contracting or implementation process related to the contract.
- 1.2 The AUTHORITY will, during the pre-contract stage, treat all APPLICANTS alike and will provide to all APPLICANTS the same information and will not provide any such information to any particular APPLICANT which could afford an advantage to that particular APPLICANT in comparison to other APPLICANTS.
- 1.3 All the officials of the CONCESSIONING AUTHORITY will report to the appropriate Government office any attempted or completed breaches of the above commitments as well as any substantial suspicion of such a breach.
- In case any such preceding misconduct on the part of such official(s) is reported by the APPLICANT to the AUTHORITY with full and verifiable facts and the same is prima facie found to be correct by the AUTHORITY necessary disciplinary proceedings, or any other action as deemed fit, including criminal proceedings may be initiated by the AUTHORITY and such a person shall be debarred from further dealings related to the contract process. In such a case, while an enquiry is being conducted by the AUTHORITY the proceedings under the contract would not be stalled.

Commitments of APPLICANTS

- 3. The APPLICANT commits itself to take all measures necessary to prevent corrupt practices, unfair means and illegal activities during any stage of its bid or during any pre-contract or post-contract stage in order to secure the contract or in furtherance to secure it and in particular commit itself to the following:-
- 3.1 The APPLICANT will not offer, directly or through intermediaries, any bribe, gift, consideration, reward, favour, any material or immaterial benefit or other advantage, commission, fees, brokerage or inducement to any official of the AUTHORITY connected directly or indirectly with the bidding process, or to any person, organisation or third party related to the contract in exchange for

any advantage in the bidding, evaluation, contracting and implementation of the contract.

- 3.2 The APPLICANT further undertakes that it has not given, offered or promised to give, directly or indirectly any bribe, gift, consideration, reward, favour, any material or immaterial benefit or other advantage, commission, fees, brokerage or inducement to any official of the AUTHORITY or otherwise in procuring the Contract or forbearing to do or having done any act in relation to the obtaining or execution of the contract or any other contract with the Government for showing or forbearing to show favour or disfavour to any person in relation to the contract or any other contract with the Government.
- 3.3* APPLICANTS shall disclose the name and address of agents and representatives and Indian APPLICANTS shall disclose their foreign principals or associates.
- 3.4* APPLICANTS shall disclose the payments to be made by them to agents/brokers or any other intermediary, in connection with this bid/contract.
- 3.5* The APPLICANT further confirms and declares to the AUTHORITY that the APPLICANT has not engaged any individual or firm or company whether Indian or foreign to intercede, facilitate or in any way to recommend to the AUTHORITY or any of its functionaries, whether officially or unofficially to the award of the contract to the APPLICANT, nor has any amount been paid, promised or intended to be paid to any such individual, firm or company in respect of any such intercession, facilitation or recommendation.
- 3.6 The APPLICANT, either while presenting the bid or during pre-contract negotiations or before signing the contract/ Concession Agreement, shall disclose any payments he has made, is committed to or intends to make to officials of the AUTHORITY or their family members, agents, brokers or any other intermediaries in connection with the contract and the details of services agreed upon for such payments.
- 3.7 The APPLICANT will not collude with other parties interested in the contract to impair the transparency, fairness and progress of the bidding process, bid evaluation, contracting and implementation of the contract.
- 3.8 The APPLICANT will not accept any advantage in exchange for any corrupt practice, unfair means and illegal activities.
- 3.9 The APPLICANT shall not use impropriety for purposes of competition or personal gain, or pass on to others, any information provided by the AUTHORITY as part of the business relationship, regarding plans, technical

- proposals and business details, including information contained in any electronic data carrier. The APPLICANT also undertakes to exercise due and adequate care lest any such information is divulged.
- 3.10 The APPLICANT commits to refrain from giving any complaint directly or through any other manner without supporting it with full and verifiable facts.
- 3.11 The APPLICANT shall not instigate or cause to instigate any third person to commit any of the actions mentioned above.
- 3.12 If the APPLICANT or any employee of the APPLICANT or any person acting on behalf of the APPLICANT, either directly or indirectly, is a relative of any of the officers of the AUTHORITY, or alternatively, if any relative of an officer of the AUTHORITY has financial interest/stake in the APPLICANT's firm, the same shall be disclosed by the APPLICANT at the time of filing of tender.
 - The term 'relative' for this purpose would be as defined in Section 6 of the Companies Act 1956.
- 3.13 The APPLICANT shall not lend to or borrow any money from or enter into any monetary dealings or transactions, directly or indirectly, with any employee of the AUTHORITY.

4. <u>Previous Transgression</u>

- 4.1 The APPLICANT declares that no previous transgression occurred in the last three years immediately before signing of this Integrity Pact, with any other company in any country in respect of any corrupt practices envisaged hereunder or with any Public Sector Enterprise in India or any Government Department in India that could justify Applicant's exclusion from the tender process.
- 4.2 The APPLICANT agrees that if it makes incorrect statement on this subject, APPLICANT can be disqualified from the tender process or the contract, if already awarded, can be terminated for such reason.

5. <u>Sanctions for Violations</u>

- Any breach of the aforesaid provisions by the APPLICANT or any one employed by it or acting on its behalf (whether with or without the knowledge of the APPLICANT) shall entitle the AUTHORITY to take all or any one of the following actions, wherever required:
 - a. To immediately call off the pre-contract negotiations without assigning any reason or giving any compensation to the APPLICANT. However, the proceedings with the other APPLICANT(s) would continue.

- b. The Bid Security (in pre-contract stage) and/or /Performance Security (after the contract is signed) shall stand forfeited either fully or partially, as decided by the AUTHORITY and the AUTHORITY shall not be required to assign any reason therefore.
- c. To immediately cancel the contract, if already signed, without giving any compensation to the APPLICANT.
- d. To cancel all or any other Contracts with the APPLICANT. The APPLICANT shall, be liable to pay compensation for any loss or damage to the AUTHORITY resulting from such cancellation/rescission and the AUTHORITY shall be entitled to deduct the amount so payable from the money(s) due to the APPLICANT.
- e. To debar the APPLICANT from participating in future bidding processes of the Government of India for a minimum period of five years, which may be further extended at the discretion of the AUTHORITY.
- f. To recover all sums paid in violation of this Pact by APPLICANT(s) to any middleman or agent or broker with a view to securing the contract.
- g. Forfeiture of Performance Security in case of a decision by the AUTHORITY to forfeit the same without assigning any reason for imposing sanction for violation of this Pact.
- 6.2 The AUTHORITY will be entitled to take all or any of the actions mentioned at para 6.1(a) to (g) of this Pact also on the Commission by the APPLICANT or any one employed by it or acting on its behalf (whether with or without the knowledge of the APPLICANT), of an offence as defined in Chapter IX of the Indian Penal code, 1860 or Prevention of Corruption Act, 1988 or any other statute enacted for prevention of corruption.
- 6.3 The decision of the AUTHORITY to the effect that a breach of the provisions of this Pact has been committed by the APPLICANT shall be final and conclusive on the APPLICANT. However, the APPLICANT can approach the Independent Monitor(s) appointed for the purposes of this Pact.

7. Fall Clause

7.1 The APPLICANT undertakes that it has not performed /is not performing similar project at a price lower than that offered in the present bid in respect of any other Ministry/Department of the Government of India or PSU and if it is found at any stage that similar project was performed by the APPLICANT to any other Ministry/Department of the Government of India or a PSU at a lower price, then that very price, with due allowance for elapsed time, will be applicable to the present case and the difference in the cost would be

refunded by the APPLICANT to the AUTHORITY, if the contract has already been concluded.

8. Independent Monitors

8.1 The AUTHORITY has appointed the following Independent Monitors (hereinafter referred to as Monitor) for this Pact in consultation with the Central Vigilance Commission.

Sri. Prem Chand Pankaj,

Ex-CMD, NEEPCO, M 402, Pioneer Park, Sector-61, Golf Course Extn. Road, Gurgaon, Haryana -

Mob: 9717433886

Email: prempankaj@gmail.com

- 8.2 The task of the Monitors shall be to review independently and objectively, whether and to what extent the parties comply with the obligations under this Pact.
- 8.3 The Monitors shall not be subject to instructions by the representatives of the parties and perform their functions neutrally and independently.
- 8.4 Both the parties accept that the Monitors have the right to access all the documents relating to the project/bidding, including minutes of meetings.
- 8.5 As soon as the Monitor notices, or has reason to believe, a violation of this Pact, he will so inform the Authority designated by the AUTHORITY,
- 8.6 The APPLICANT(s) accepts that the Monitor has the right to access without restriction to all Project documentation of the AUTHORITY including that provided by the APPLICANT. The APPLICANT will also grant the Monitor, upon his request and demonstration of a valid interest, unrestricted and unconditional access to his project documentation. The same is applicable to Subcontractors. The Monitor shall be under contractual obligation to treat the information and documents of the APPLICANT/Subcontractor(s) with confidentiality.
- 8.7 The AUTHORITY will provide to the Monitor sufficient information about all meetings among the parties related to the Project provided such meetings could have an impact on the contractual relations between the parties. The parties will offer to the Monitor the option to participate in such meetings.
- 8.8 The Monitor will submit a written report to the designated Authority of AUTHORITY within 8 to 10 weeks from the date of reference or intimation to him by the AUTHORITY / APPLICANT and, should the occasion arise, submit proposals for correcting problematic situations.

9. Facilitation of Investigation

In case of any allegation of violation of any provisions of this pact or payment of commission, the AUTHORITY or its agencies shall be entitled to examine all the documents including the Books of Accounts of the APPLICANT and the APPLICANT shall provide necessary information and documents in English and shall extend all possible help for the purpose of such examination.

10. <u>Law and Place of Jurisdiction</u>

This Pact is subject to Indian Law.' The place of performance and jurisdiction is the seat of the AUTHORITY.

11. Other Legal Actions

The actions stipulated in this Integrity Pact are without prejudice to any other legal action that may follow in accordance with the provisions of the extant law in force relating to any civil or criminal proceedings.

12. Validity

applicants.

- 12.1 The validity of this Integrity Pact shall be from date of its signing and extend upto 5 years or the complete execution of the contract to the satisfaction of both the AUTHORITY and the APPLICANT, including warranty period, whichever is later. In case APPLICANT is unsuccessful, this Integrity Pact shall expire after six months from the date of the signing of the contract.
- 12.2 Should one or several provisions of this Pact turn out to be invalid; the remainder of this Pact shall remain valid. In this case, the parties will strive to come to an agreement to their original intentions.

13. The parties hereby sign this Inte	grity Pact aton
APPLICANT	CONCESSIONING AUTHORITY
Name of the Officer.	Chief Engineer (Civil)
Designation	Dept./MINISTRY/PSU
<u>Witness</u> 1	<u>Witness</u> 1
2	2
	eed to be amended/ deleted in line with the to involvement of Indian agents of foreign

FINANCIAL PROPOSAL

Form - 1

Covering Letter

(On Applicant's letter head)
(Date and Reference)
To,
Chief Engineer Administrative Office Building ^{2rd} Floor, New Mangalore Port Authority Panambur Mangalore 575010
Dear Sir,
Subject: "Appointment of Transaction Advisor for Concessioning of Berth No. 13 for handling Liquid Bulk / LPG / POL products Chemicals on Captive / PPP Mode at New Mangalore Port"
I/We, (Applicant's name) herewith enclose the Financial Proposal for selection of my/our firm as Consultant for Transaction Advisory Services for above.
I/We agree that this offer shall remain valid for a period of 120 (one hundred twenty) days from the Proposal Due Date or such further period as may be mutually agreed upon.
Yours faithfully,
(Signature, name and designation of the authorised signatory)
Note: The Financial Proposal is to be submitted strictly as per forms given in the RFP.

(See Clause 2.1.3)

Form - 2

Reference Format of Financial Proposal

[Shall be submitted in E-Tender only]

Name of Work Appointment of Transaction Advisor for Concessioning of Berth No. 13 for handling Liquid Bulk / LPG / POL products / Chemicals on Captive / PPP Mode at New Mangalore Port.

SCHEDULE OF PRICES

SI. No.	Description	Amount (Rs.)
1	Transaction Advisory Services for Concessioning of Berth No. 13 for handling Liquid Bulk / LPG / POL products / Chemicals on Captive / PPP Mode at New Mangalore Port.	Amount in

Total cost of consultancy in words -----

Note.

- 1. The aforesaid professional fees, payable to the Consultant in accordance with the Financial Proposal, shall cover the costs of telephone/fax, photocopying, couriers and postage, collections and deliveries, traveling expense, stationery, costs of support staff and counsel fee, cost of stamping Agreement, cost towards making 10 sets of agreement, overheads, etc., including all taxes and duties except GST. No additional charges in respect thereof shall be due or payable. The fees shall be limited to the amounts indicated above and no escalation on any account will be payable on the above amounts.
- 2. Outstation travel (for locations other than Mangalore) shall be undertaken as per request of the Authority. The Financial Expert One Key Personnel and one AssistantAssociate Consultant shall be entitled to economic class air travel, and board and lodging in a Three-star hotel. Bills for reimbursement hereunder may be submitted along in original supporting materials, along with a Statement of Expenses, duly certified by the Authorised Representative.
- 3. All payments shall be made in Indian Rupees and shall be subject to applicable Indian withholding taxes if any.

3.4. Applicable GST shall be paid separately.

Signature Name Designation Date For and on behalf

NEW MANGALORE PORT TRUST

RFP for appointment of Transaction Advisor for Concessioning of Berth No. 13 for handling POL / Chemicals on Captive / PPP Mode

PROJECT IMPLEMENTATION SCHEDULE

Key Dates	Activities / Deliverables	Period in Weeks*	%age of Percentage	1	2	3 4	5	6 7	7 8	9	10	11 12	13	14 1	5 16	17	18 19	20	21	22 23	3 24	25 2	26 27	28	29	30 31	32	33	34 3	5 36	37	38 3	9 40	41	42 43	3 44	45	46 4	.7 48	3 49	50	51	52
KD1	Submission of Inception Report	2	5			T																							T	T		T		П				T	Т	T	T		
KD2	Submission of Feasibility Report & Revenue Model	8	5																						П					T									T		T		_
KD3	Submission of EOI document for approval and invitation of EOI.	4	5		П	Т			T																									П							T	T	_
KD4	Evaluation of EOI applications & Concluding EOI process	16	5																																						T	T	
KD 5	Submission of RFQ for approval and invitation of RFQ	4	10			T	Г									П													T	T		\top		П							T		_
KD 6	Submission of Tariff proposal to TAMP / Adjudicatory Board	4	10			Ť	t									Ħ		П				\top		t	П					t									+		t	T	_
KD 7	Evaluation of RFQ applications and Concluding of RFQ process	16	10			T												П	T																				+		t		_
KD 8	Submission of documents for Security clearance of bidders	4	10														\top	П				П	Т	Т	П		П	П											+		t	t	_
KD 9	Submission of RFP and Draft Concession Agreement for approval	4	10															П																П					+		t	t	_
KD 10	Submission of appraisal report to PPPAC/SFC/MoS	4	5																										T			İ							\dagger		T	T	_
KD 11	Issue of RFP including preparation of Draft Concession Agreement	4	5			T	T						П					П					T						T	T		T		П							T		_
KD 12	Conclusion of Bid Process and issue of LoA	4	5			t	T										†	П					Ť	t		1	H		t	T										h		T	_
KD 13	Signing of Concession Agreement	4	10			Ť	t										†	П					\dagger			\dagger			T	\dagger		+											
KD 14	Completion of Services till financial close	24	5	53	54	55 5	6 57	58	59 6	60 61	62	63 64	65	66 6	67 68	69	70 7	1 72	73	74 7	5 76					\dagger				\dagger									+		Т		

Note:
* Excluding time taken by Authority in providing comments on Draft Reports. Final Report to be submitted in a week's time after receipt of comments of the Authority.

